# Measures and applications of quantum correlations

## Gerardo Adesso, Thomas R. Bromley, Marco Cianciaruso

Centre for the Mathematics and Theoretical Physics of Quantum Non-Equilibrium Systems, School of Mathematical Sciences, The University of Nottingham, University Park, Nottingham NG7 2RD, United Kingdom

Abstract. Quantum information theory is built upon the realisation that quantum resources like coherence and entanglement can be exploited for novel or enhanced ways of transmitting and manipulating information, such as quantum cryptography, teleportation, and quantum computing. We now know that there is potentially much more than entanglement behind the power of quantum information processing. There exist more general forms of non-classical correlations, stemming from fundamental principles such as the necessary disturbance induced by a local measurement, or the persistence of quantum coherence in all possible local bases. These signatures can be identified and are resilient in almost all quantum states, and have been linked to the enhanced performance of certain quantum protocols over classical ones in noisy conditions. Their presence represents, among other things, one of the most essential manifestations of quantumness in cooperative systems, from the subatomic to the macroscopic domain. In this work we give an overview of the current quest for a proper understanding and characterisation of the frontier between classical and quantum correlations in composite states. We focus on various approaches to define and quantify general quantum correlations, based on different yet interlinked physical perspectives, and comment on the operational significance of the ensuing measures for quantum technology tasks such as information encoding, distribution, discrimination and metrology. We then provide a broader outlook of a few applications in which quantumness beyond entanglement looks fit to play a key role.

PACS numbers: 03.67.Mn, 03.65.Ud, 03.65.Yz

## CONTENTS

Contents						
1	Intro	Introduction 4				
2	Qua	ntum co	orrelations	5		
	2.1	The ma	any shades of quantumness of correlations	. 5		
	2.2	Classic	ally correlated states	. 7		
	2.3	Charac	terising classically correlated states	. 10		
		2.3.1	Invariance under a complete local projective measurement	. 11		
		2.3.2	Non-creation of entaglement with an apparatus	. 15		
		2.3.3	Invariance under a local non-degenerate unitary	. 16		
		2.3.4	Incoherence in a local basis	. 16		
		2.3.5	Invariance under an entanglement-breaking channel	. 17		
	2.4	Definin	ag and characterising quantumly correlated states	. 18		
3	Mea	sures of	quantum correlations	20		
	3.1	Require	ements for a <i>bona fide</i> measure of quantum correlations	. 20		
	3.2	Types of	of measures of quantum correlations	. 23		
		3.2.1	Geometric measures	. 24		
		3.2.2	Measurement induced geometric measures	. 25		
		3.2.3	Measurement induced informational measures	. 26		
		3.2.4	Entanglement activation measures	. 27		
		3.2.5	Unitary response measures	. 28		
		3.2.6	Coherence based measures	. 29		
		3.2.7	Recoverability measures	. 31		
	3.3	Geome	tric measures	. 31		
		3.3.1	Relative entropy	. 31		
		3.3.2	Hilbert-Schmidt distance	. 32		
		3.3.3	Trace distance	. 33		
		3.3.4	Bures distance	. 34		
		3.3.5	Hellinger distance	. 34		
		3.3.6	Hierarchy of geometric measures	. 34		
		3.3.7	Other distances	. 35		
	3.4	Measu	rement induced geometric measures	. 36		
		3.4.1	Hilbert-Schmidt and trace distance	. 36		
		3.4.2	Bures and Hellinger distance	. 36		
	3.5	Measu	rement induced informational measures	. 36		
		3.5.1	von Neumann entropy based measures: Quantum deficit	. 36		
		3.5.2	Mutual information based measures: Quantum discord	. 37		
		3.5.3	Hierarchy of informational measures and their regularisation	. 41		
		3.5.4	Other informational measures	. 42		
	3.6	Entang	lement activation measures	. 43		
		3.6.1	Distillable entanglement and relative entropy	. 43		
		3.6.2	Negativity of quantumness	. 44		
	3.7	Unitary	response measures	. 45		
		3.7.1	Hilbert-Schmidt distance	. 45		
		3.7.2	Trace distance	. 45		
		3.7.3	Bures and Hellinger distance: Discord of response	. 45		
		3.7.4	Chernoff distance: Discriminating strength	. 46		

2

	3.8	Coherence based measures			
		3.8.1	Relative entropy	47	
		3.8.2	Infidelity	47	
		3.8.3	$\ell_1$ norm	48	
		3.8.4	Asymmetry: Local quantum uncertainty and interferometric power .	48	
		3.8.5	Other coherence quantifiers	49	
3.9 Recoverability measures		erability measures	50		
		3.9.1	Surprisal of measurement recoverability	50	
	3.10	Summa	ary of measures	51	
4	App	lication	s and operational interpretations of quantum correlations	53	
-	4.1	Ouanti	im information and communication	53	
		411	Local broadcasting	53	
		412	Entanglement distribution	55	
		413	Quantum state merging	57	
		414	Quantum state redistribution	59	
		415	Remote state preparation	60	
		416	Quantum cryptography	62	
		4.1.0	Quantum locking of classical correlations	62	
	12	-1.1.7	um metrology and discrimination	64	
	4.2		Plack how quantum metrology	64	
		4.2.1	State discrimination and succession illumination	04	
	12	4.2.2 Example of		00 60	
	4.3 Further physics applications			09 (0	
		4.3.1	Quantum thermodynamics	69	
		4.3.2	Many-body systems and quantum phase transitions	71	
5	Con	cluding	remarks and outlook	73	

## 1. Introduction

Quantum theory has been astonishingly successful for roughly a century now. Beyond its explanatory power, it has enabled us to break new grounds in technology. Lasers, semiconductor devices, solar panels, and magnetic resonance imaging, are just examples of everyday technologies based on quantum theory, nowadays classified as "Quantum Technologies 1.0". The ultimate exploitation of the quantum laws applied to information processing is now promising to revolutionise the information and communication technology sector as well, unfolding the era of "Quantum Technologies 2.0" [1]. Secure quantum communication, enhanced quantum sensing, and the prospects of quantum computation, have been ignited by the realisation that quintessential quantum features, like superposition and entanglement, can be exploited as resources for data encoding and transmission in ways which are substantially more efficient or radically novel compared to those allowed by classical resources alone [2]. It is then clear that the deep understanding of the most genuine traits of quantum mechanics bears a strong promise for application into disruptive technologies.

Remarkably, the question of defining when a system behaves in a *quantum* way, rather than as an effectively *classical* one, still lacks a universal answer. This statement holds in particular when analysing composite systems and the nature of the correlations between their subsystems. Until recently, theoretical and experimental attention has been mainly devoted to entanglement among different subsystems [3]: entangled states clearly display a non-classical nature, and some of them can exhibit even stronger deviations from classicality such as steering [4] and nonlocality [5]. However, even unentangled states are not, in most cases, amenable to a fully classical description. More general forms of quantum correlations, exemplified by the so-called quantum discord [6, 7], capture basic aspects of quantum theory, such as the fact that local measurements on parts of a composite system necessarily induce an overall disturbance in the state. If one recognises genuine "quantumness" according to this paradigm, then almost all generally mixed quantum states of two or more subsystems display quantum correlations [8], even in the absence of entanglement.

The present topical review focuses precisely on this type of quantum correlations (QCs). In the last one-and-a-half decades, an increasing interest has been devoted by the international community to the study and the characterisation of QCs beyond entanglement, and notable progress has been achieved. Pursuing such an investigation further is important for two main reasons. On the fundamental side, it shines light on the ultimate frontiers of the quantum world, that is, on the most elemental (and consequently elusive) traits that distinguish the behaviour of a quantumly correlated system from one fully ascribed to a joint classical probability distribution. On the practical side, it may reveal operational tasks where QCs even in absence of entanglement can still translate into a quantum enhancement, thus yielding more resilient and more accessible resources for future quantum technologies.

Given the large extent of research devoted to various aspects of QCs in recent years, there have been already a few introductory and review articles discussing the main concepts and presenting some of the most important findings in this research area. In particular, the reader may consult [9] for a pedagogical introduction, [10] for a discussion of QCs in the context of resource theories, [11] for a self-contained exposition of QCs and their role in quantum information theory, and [12] for a longer and more encyclopaedic coverage of relevant results until 2012. Nevertheless, it is fair to say that the topic of general QCs has still not reached the level of comprehension and appreciation by the broader mathematical and physical community that is instead claimed by entanglement theory. The reasons for this are varied. On the one hand, the topic is certainly not mature yet, and many open questions remain at the foundational level which are still in need of deeper insights and original solutions.

On the other hand, this is also partly due to the excessive dispersion of certain research on QCs towards studies with little physical content, such as the mere calculation of some QCs quantifier in a particular system or model. Such a fragmentation has made it more difficult for some landmark advances stemming from the study of QCs to stand out and be clearly recognised by the international pool of non-specialists.

This review aims to give the "*ABC*" of QCs,‡ with the intention of highlighting the physical understanding of the concepts involved without compromising their mathematical rigour, and with a specific focus on the following three basic questions:

- (A) What are the signature traits of QCs from a fundamental point of view?
- (B) How can we meaningfully quantify QCs in arbitrary quantum states?
- (C) What are QCs good for in practical applications?

The review is particularly targeting beginners who are willing to embark in fascinating research at the quantum-classical border — and who may hopefully find here the right motivation and a set of problems to start tackling — as well as expert colleagues who may have already addressed a particular aspect of QCs research — and who may be looking for further inspiration from the bigger picture to move the next step forward. To keep it close to its focus while maintaining an acceptable size, however, the review is also necessarily incomplete (meaning that several topics which even witnessed intense research are excluded, such as the one dealing with the dynamics of QCs in open quantum systems, or their potential role in quantum computing schemes with mixed states, among all), and many relevant references may have been omitted from our already comprehensive bibliography. In those cases, while issuing apologies in advance, we invite the interested reader to consult e.g. [12] as well as the original literature for further information.

The review is organised as follows. In Section 2 we introduce the hierarchy of correlations in composite quantum systems, briefly highlighting the classification of QCs with respect to entanglement, steering and nonlocality. We then give the formal definition of QCs (or lack thereof) and present in an original fashion various defining traits that pinpoint QCs as opposed to classical correlations. In Section 3 we first introduce the progress achieved so far in the formalisation of a resource theory of QCs, collecting some necessary requirements and desiderata that any valid measure of QCs should satisfy. We then review a plethora of recently introduced QCs measures, which are shown to capture quantitatively the different defining traits introduced in the previous section. Particular effort is devoted to highlighting interlinks and dependency relations between the various types of measures, including the most recent insights not covered in other existing reviews. Section 4 contains an overview of various important applications of OCs in the contexts of quantum information theory, thermodynamics, and many-body physics. Emphasis is placed on those settings which provide direct operational interpretations for some of the measures defined in the previous section. We conclude in Section 5 with a summary of covered and uncovered topics accompanied by an outlook of a few currently open questions in QCs research.

## 2. Quantum correlations

#### 2.1. The many shades of quantumness of correlations

The simplest testbed for the study of correlations is that of a composite quantum system made of two subsystems A and B, each associated with a (finite dimensional) Hilbert space  $\mathcal{H}_A$  and

# "ABC" also stands for the acronym of the authors' initials.



Figure 1. Hierarchy of correlations in states of composite quantum systems. Pure states can be either uncorrelated or just entangled. For mixed states, several layers of non-classical correlations have been identified, going significantly beyond the seminal paradigm of [13]. In order of decreasing strength, these can be classified as: nonlocality  $\Rightarrow$  steering  $\Rightarrow$  entanglement  $\Rightarrow$  general quantum correlations. All of these forms of non-classical correlations can enable classically impossible tasks. For instance, device-independent quantum cryptography requires nonlocality [5], entanglement-assisted subchannel discrimination depends on steering [14], while quantum teleportation and dense coding exploit plain entanglement [3]. In this review we shall focus on the lower end of the spectrum, i.e., quantum correlations (QCs) beyond entanglement. They incarnate the most general yet arguably the least understood manifestation of non-classical correlations in composite quantum systems. Their fundamental and operational value will be illustrated in various physical settings throughout the review.

 $\mathcal{H}_B$ , respectively. If the system is prepared in a pure quantum state  $|\psi\rangle_{AB} \in \mathcal{H}_{AB}$ , where the Hilbert space of the composite system is defined as the tensor product  $\mathcal{H}_{AB} := \mathcal{H}_A \otimes \mathcal{H}_B$  of the marginal Hilbert spaces, then essentially two possibilities can occur. The first is that the two subsystems are completely independent, in which case the state takes the form of a tensor product state  $|\psi\rangle_{AB} = |\alpha\rangle_A \otimes |\beta\rangle_B$ , with  $|\alpha\rangle_A \in \mathcal{H}_A$  and  $|\beta\rangle_B \in \mathcal{H}_B$ . In this case there are no correlations of any form (classical or quantum) between the two parts of the composite system. The second possibility is that, instead, there exists no local state for *A* and *B* such that the global state can be written in tensor product form,

$$|\psi\rangle_{AB} \neq |\alpha\rangle_A \otimes |\beta\rangle_B . \tag{1}$$

In this case,  $|\psi\rangle_{AB}$  describes an *entangled* state of the two subsystems A and B. Entanglement encompasses any possible form of correlations in pure bipartite states, and can manifest in different yet equivalent ways. For instance, every pure entangled state is nonlocal, meaning that it can violate a Bell inequality [15]. Similarly, every pure entangled state is necessarily disturbed by the action of any possible local measurement [6, 7]. Therefore, entanglement, nonlocality, and QCs are generally synonymous for pure bipartite states.

As illustrated in Fig. 1, the situation is subtler and richer in case A and B are globally prepared in a mixed state, described by a density matrix  $\rho_{AB} \in \mathcal{D}_{AB}$ , where  $\mathcal{D}_{AB}$  denotes the convex set of all density operators acting on  $\mathcal{H}_{AB}$ . The state  $\rho_{AB}$  is *separable*, or unentangled, if it can be prepared by means of local operations and classical communication (LOCC), i.e., if it takes the form

$$\rho_{AB} = \sum_{i} p_i \varsigma_A^{(i)} \otimes \tau_B^{(i)} , \qquad (2)$$

with  $\{p_i\}$  a probability distribution, and quantum states  $\{\varsigma_A^{(i)}\}$  of *A* and  $\{\tau_B^{(i)}\}$  of *B*. The set  $S_{AB} \subset \mathcal{D}_{AB}$  of separable states is constituted therefore by all states  $\rho_{AB}$  of the form given by Eq. (2),

$$S_{AB} := \left\{ \rho_{AB} \mid \rho_{AB} = \sum_{i} p_i \varsigma_A^{(i)} \otimes \tau_B^{(i)} \right\}.$$
(3)

Any other state  $\rho_{AB} \notin S_{AB}$  is *entangled*. Mixed entangled states are hence defined as those which cannot be decomposed as a convex mixture of product states. Notice that, unlike the special case of pure states, the set of separable states is in general strictly larger than the set of product states,  $S_{AB} \supset \mathcal{P}_{AB}$ , where

$$\mathcal{P}_{AB} := \left\{ \rho_{AB} \mid \rho_{AB} = \rho_A \otimes \rho_B \right\}.$$
(4)

Entanglement, one of the most fundamental resources of quantum information theory, can be then recognised as a direct consequence of two key ingredients of quantum mechanics: the superposition principle and the tensorial structure of the Hilbert space. We defer the reader to [3] for a comprehensive review on entanglement, and to [16] for a compendium of the most widely adopted entanglement measures. Within the set of entangled states, one can further distinguish some layers of more stringent forms of non-classicality. In particular, some but not all entangled states are *steerable*, and some but not all steerable states are *nonlocal*.

Steering, i.e. the possibility of manipulating the state of one subsystem by making measurements on the other, captures the original essence of inseparability adversed by Einstein, Podolsky and Rosen (EPR) [17] and appreciated by Schrödinger [18], and has been recently formalised in the modern language of quantum information theory [19]. It is an asymmetric form of correlations, which means that some states can be steered from *A* to *B* but not the other way around. The reader may refer to [4] for a recent review on EPR steering.

On the other hand, nonlocality, intended as a violation of EPR local realism [17], represents the most radical departure from a classical description of the world, and has received considerable attention in the last half century since Bell's 1964 theorem [20]. Recently, a triplet of experiments demonstrating Bell nonlocality free of traditional loopholes have been accomplished [21, 22, 23], confirming the predictions of quantum theory. Nonlocality, like entanglement, is a symmetric type of correlations, invariant under the swap of parties *A* and *B*. The reader is referred to [5] for a review on nonlocality and its applications.

As remarked, here we are mainly interested in signatures of quantumness beyond entanglement. Therefore, an important question we should consider is: Are the correlations in separable states completely classical? In the following we argue that, in general, this is not the case. The only states which may be regarded as classically correlated form a negligible corner of the subset of separable states, and will be formally defined in the next subsection.

## 2.2. Classically correlated states

Consider a composite system consisting of a classical bit and a quantum bit (qubit). For convenience, we shall adopt the same formalism for both. The classical bit can either be in the state  $|0\rangle$  or in the state  $|1\rangle$ , representing e.g. "off" or "on" in modern binary electronics and communications. If the classical bit is in  $|0\rangle$ , one can write the composite state as  $|0\rangle \langle 0|_A \otimes \rho_B$ , where we have identified the classical bit as subsystem *A* and the qubit as subsystem *B*, with  $\rho_B$  a quantum state. However, if the state of the classical bit is unknown, then the composite state state  $\rho_{AB}$  is a statistical mixture, i.e.

$$\rho_{AB} = p_0 |0\rangle \langle 0|_A \otimes \rho_B^{(0)} + p_1 |1\rangle \langle 1|_A \otimes \rho_B^{(1)}, \tag{5}$$

where  $p_0$  and  $p_1$  are probabilities adding up to one, while  $\rho_B^{(0)}$  and  $\rho_B^{(1)}$  denote the state of the quantum bit *B* when the classical bit *A* is in  $|0\rangle$  and  $|1\rangle$ , respectively. This is an example of what we call a *classical-quantum* state, or classical on *A*, since subsystem *A* is classical. Equivalently, we can say that  $\rho_{AB}$  is *classically correlated* with respect to subsystem *A*, the motivation for this terminology becoming clear soon.

Going beyond just bits, one can think about any classical system as consisting of a collection of distinct states, which can be represented using an orthonormal basis  $\{|i\rangle\}$ . Any classical-quantum state can then be written as a statistical mixture in the following way [24],

$$\rho_{AB} = \sum_{i} p_{i} |i\rangle \langle i|_{A} \otimes \rho_{B}^{(i)}, \qquad (6)$$

where  $\rho_B^{(i)}$  is the quantum state of subsystem *B* (now in general a *d*-dimensional quantum system, or qudit) when *A* is in the state  $|i\rangle_A$ , and  $\{p_i\}$  is a probability distribution. The set  $\mathcal{C}_A$  of classical-quantum states is then formed by any state that can be written as in Eq. (6),

$$\mathcal{C}_A := \left\{ \rho_{AB} \mid \rho_{AB} = \sum_i p_i \left| i \right\rangle \left\langle i \right|_A \otimes \rho_B^{(i)} \right\},\tag{7}$$

where  $\{|i\rangle_A\}$  is any orthonormal basis of subsystem A and  $\{\rho_B^{(i)}\}\$  are any quantum states of subsystem B. We stress that the orthonormal basis  $\{|i\rangle_A\}\$  appearing in Eq. (7) is not fixed but rather can be chosen from all the orthonormal bases of subsystem A. The set  $\mathcal{C}_A$  may look similar to the set  $\mathcal{S}_{AB}$  of separable states, defined in Eq. (3), but there is an important difference: in Eq. (3), any state of subsystem A is allowed in the ensemble, while in Eq. (7) only projectors corresponding to an orthonormal basis can be considered. This reveals that classical-quantum states form a significantly smaller subset of the set of separable states,  $\mathcal{C}_A \subset \mathcal{S}_{AB}$ .

Swapping the roles of A and B, one can define the *quantum-classical* states along analogous lines,

$$\rho_{AB} = \sum_{j} p_{j} \rho_{A}^{(j)} \otimes |j\rangle \langle j|_{B}, \qquad (8)$$

and the corresponding set  $\mathcal{C}_B$ ,

$$\mathcal{C}_B := \left\{ \rho_{AB} \mid \rho_{AB} = \sum_j p_j \rho_A^{(j)} \otimes |j\rangle \langle j|_B \right\},\tag{9}$$

where  $\{|j\rangle_B\}$  is any orthonormal basis of subsystem *B* and  $\{\rho_A^{(j)}\}\$  are any quantum states of subsystem *A*. These states can be equivalently described as *classically correlated* with respect to subsystem *B*.

Finally, if we consider the composition of two classical objects, we can introduce the set of *classical-classical* states, or classical on *A* and *B*, which are *classically correlated* with respect to both subsystems. A state is classical-classical if it can be written as

$$\rho_{AB} = \sum_{ij} p_{ij} |i\rangle \langle i|_A \otimes |j\rangle \langle j|_B, \qquad (10)$$

where we now have a joint probability distribution  $\{p_{ij}\}$  and orthonormal bases for both subsystems A and B. The set  $\mathcal{C}_{AB}$  of classical-classical states is then formed by any state that can be written as in Eq. (10),

$$\mathcal{C}_{AB} := \left\{ \rho_{AB} \mid \rho_{AB} = \sum_{ij} p_{ij} \left| i \right\rangle \left\langle i \right|_A \otimes \left| j \right\rangle \left\langle j \right|_B \right\},\tag{11}$$



Figure 2. Visualisation of different subsets of correlated states in bipartite quantum systems. In both panels, we depict in blue the (uncorrelated) product states belonging to the subset  $\mathcal{P}_{AB}$ , in magenta the classical-classical states belonging to the subset  $\mathcal{C}_{AB}$ , in dashed red the classical-quantum states belonging to the subset  $C_A$ , in dashed green the quantum-classical states belonging to the subset  $\mathcal{C}_B$ , in light grey (with dotted purple boundary) the separable states belonging to the subset  $S_{AB}$ , and in dark grey (with solid black boundary) all the remaining (entangled) states, completing the whole set  $\mathcal{D}_{AB}$ . Panel (a) is an artistic impression which does not faithfully reflect the topology of the various subsets (in particular, the sets  $C_{AB}$ ,  $\mathcal{C}_A$ ,  $\mathcal{C}_B$ , and  $\mathcal{P}_{AB}$  are all of null measure and nowhere dense). Panel (b) depicts the actual subsets for the two-qubit states  $\rho_{AB}(x, y)$  of Eq. (13), with  $0 \le x + y \le 1$ . For y = 0 (horizontal axis, dashed red), we have  $\rho_{AB}(x, 0) = \frac{1-x}{2} |0\rangle \langle 0|_A \otimes |0\rangle \langle 0|_B + \frac{1+x}{2} |1\rangle \langle 1|_A \otimes \rho_B^{(1)} \in \mathcal{C}_A$ , with  $\rho_{R}^{(1)} = (1+x)^{-1} (x|0\rangle \langle 0|_{B} + x|0\rangle \langle 1|_{B} + x|1\rangle \langle 0|_{B} + |1\rangle \langle 1|_{B}).$  Analogously, for x = 0 (vertical axis, dashed green), we have  $\rho_{AB}(0, y) \in \mathcal{C}_B$ , obtained from the previous case upon swapping  $A \leftrightarrow B$  and  $x \leftrightarrow y$ . In particular, for x = y = 0 (magenta dot), we have  $\rho_{AB}(0,0) = \frac{1}{2}(|00\rangle \langle 00| +$  $|11\rangle\langle 11|\rangle \in \mathcal{C}_{AB}$ . Furthermore,  $\rho_{AB}(1,0) = |1\rangle\langle 1|_A \otimes |+\rangle\langle +|_B \in \mathcal{P}_{AB}$  (blue dot), with  $|+\rangle = (|0\rangle + |1\rangle)/\sqrt{2}$ , and similarly  $\rho_{AB}(0, 1) = |+\rangle \langle +|_A \otimes |1\rangle \langle 1|_B \in \mathcal{P}_{AB}$  (blue dot). Finally, the state  $\rho_{AB}(x, y)$  has QCs for all the other admissible values of x and y, and is further entangled if  $1 + 10xy - 16x^2y^2(4x+1)(4y+1) - 7(x^2+y^2) + 2(x+y)(1+2(x^2-4xy+y^2)) < 0$  (the threshold is indicated by the dotted purple curve), as it can be verified by applying the partial transposition criterion [3].

where  $\{|i\rangle_A\}$  and  $\{|j\rangle_B\}$  are any orthonormal bases of subsystem A and B, respectively. These states can be thought of as the embedding of a joint classical probability distribution  $\{p_{ij}\}$  into a density matrix formalism labelled by orthonormal index vectors on each subsystem.

It holds by definition that classical-classical states amount to those which are both classical-quantum and quantum-classical, that is,  $\mathcal{C}_{AB} = \mathcal{C}_A \cap \mathcal{C}_B$ . More generally, we have

$$\mathcal{P}_{AB} \subset \mathcal{C}_{AB} \subset \{\mathcal{C}_A, \mathcal{C}_B\} \subset \mathcal{S}_{AB} \subset \mathcal{D}_{AB}.$$
(12)

In this hierarchy, only the two rightmost sets (containing separable states, and all states, respectively) are convex, while all the remaining ones are not; that is, mixing two classically correlated states one may obtain a state which is not classically correlated anymore. Another interesting fact is that, while separable states span a finite volume in the space of all quantum states, the sets  $C_A$ ,  $C_B$ , and consequently  $C_{AB}$  are of null measure and nowhere dense within  $S_{AB}$  [8]. Their topology is therefore difficult to visualise, and we can only present an artistic impression in Fig. 2(a). Nevertheless, to achieve a more specific rendition, in Fig. 2(b) we

discuss a particular example of a family of two-qubit states defined as

$$\rho_{AB}(x,y) = \frac{1}{2+8xy} \begin{pmatrix} 1-x-y & 0 & 0 & 0\\ 0 & 4xy+y & 4xy & y\\ 0 & 4xy & 4xy+x & x\\ 0 & y & x & 1 \end{pmatrix},$$
(13)

dependent on two real parameters x and y with  $0 \le x + y \le 1$ , and spanning the various types of correlations discussed in this section (from no correlations up to entanglement). Another frequently studied instance where the geometry of correlations is particularly appealing is that of so-called Bell diagonal states of two qubits, defined as arbitrary mixtures of the four maximally entangled Bell states, which (up to local unitaries) represent all two-qubit states with maximally mixed marginals, see e.g. [25, 26, 27].

Throughout the review, we will consider without loss of generality the settings in which classicality is either attributed to subsystem *A* alone, or to both subsystems, and we will sometimes use the simple terminology "classical states" to refer either to classical-quantum or classical-classical states, depending on the context.

#### 2.3. Characterising classically correlated states

We now aim to illustrate several alternative characterisations of classical states in (finite dimensional) composite quantum systems, in order to aid with their physical understanding. This will accordingly be useful to appreciate the various defining traits of QCs in Section 2.4, and guide the possible approaches to their quantification in Section 3.

Here we first succinctly introduce the reader to such physical properties, which are illustrated schematically in the left column of Fig. 3. We will then go into further details for each property (including the definition and explanation of the specialised concepts mentioned below) in the following subsections.

Specifically, in a bipartite system, a state  $\rho_{AB}$  is classical if and only if it complies with either one of the following defining properties, which are all equivalent to each other.

- **Property 1:**  $\rho_{AB}$  is of the form given in Eq. (6) or Eq. (10) (referring to elements of  $\mathcal{C}_A$  or  $\mathcal{C}_{AB}$ , respectively).
- **Property 2:**  $\rho_{AB}$  is invariant under at least one local complete projective measurement on one or both subsystems (referring to elements of  $C_A$  or  $C_{AB}$ , respectively).
- **Property 3:**  $\rho_{AB}$  does not become entangled with an apparatus during the pre-measurement stage of at least one local complete projective measurement of one or both subsystems (referring to elements of  $C_A$  or  $C_{AB}$ , respectively).
- **Property 4:**  $\rho_{AB}$  is invariant under at least one local unitary operation with non-degenerate spectrum on subsystem A (referring to elements of  $C_A$ ).
- **Property 5:**  $\rho_{AB}$  is incoherent with respect to at least one local orthonormal basis for one or both subsystems (referring to elements of  $\mathcal{C}_A$  or  $\mathcal{C}_{AB}$ , respectively).
- **Property 6:**  $\rho_{AB}$  is invariant under at least one entanglement-breaking channel acting on one or both subsystems (referring to elements of  $\mathcal{C}_A$  or  $\mathcal{C}_{AB}$ , respectively).

Most of these notions can be extended to introduce a hierarchy of partly classical states in multipartite systems [28], however we shall focus on the bipartite case in the present review. Notice further that additional necessary and sufficient characterisations of classical states have been provided e.g. in [24, 25, 29, 30, 31, 12].



Figure 3. Schematics of the defining properties of classical-quantum states (left) versus states with one-sided QCs with respect to subsystem A (right) in a bipartite system AB, as detailed in Sections 2.3 and 2.4. Analogous schemes can be drawn to depict classical-classical states versus states with two-sided QCs, with the exception of row (c). In the drawings, single, double, and triple ribbon connectors denote respectively classical correlations, QCs, and entanglement. Row (a): Classical states may remain invariant under a complete local projective measurement (Property 2), while states with nonzero QCs are always altered by any such measurement (Definition 2). Row (b): Classical states may not get entangled with an apparatus A' during a local measurement (Property 3), while states with nonzero QCs always lead to creation of entanglement with an apparatus at the pre-measurement stage (Definition 3); the pre-measurement interaction is represented as a local unitary followed by a generalised control-NOT gate. Row (c): Classical states may remain invariant under a local non-degenerate unitary (Property 4), while states with nonzero QCs are always altered by any such operation (Definition 4). Row (d): Classical states may be incoherent with respect to a local basis (Property 5), while states with nonzero QCs exhibit coherence (rendered as ability to display interference in a double slit experiment) in all local bases (Definition 5). Row (e): Classical states may remain invariant under an entanglement-breaking channel (Property 6), while states with nonzero QCs are always altered by any such channel (Definition 6); the entanglementbreaking channel is depicted as a local measurement followed by a preparation map. These characterisations form the basis to quantify QCs in an arbitrary bipartite quantum state as well, as discussed further in Section 3.2 (see Table 1).

2.3.1. Invariance under a complete local projective measurement We begin this section with a brief refresh of measurements in quantum mechanics [32, 2], in particular focusing on how

local measurements can be used to define and motivate the notion of classically correlated states. A detailed introduction to measurements within the context of QCs can be found in [33].

A generalised quantum measurement can be mathematically described by resorting to the notion of a *quantum instrument*, which is a set  $\mathcal{M} = {\mu_i}$  of subnormalised (i.e., trace nonincreasing) completely positive maps  $\mu_i$ , that sum up to a completely positive trace preserving (CPTP) map, i.e., to a quantum channel  $\mu := \sum_i \mu_i$ . We can think of each subnormalised map  $\mu_i$  as corresponding to a measurement outcome *i*. Given a state  $\rho$ , the probability of obtaining the outcome *i* when applying the quantum instrument  $\mathcal{M}$  to  $\rho$  is

$$p_i = \operatorname{Tr}\left(\mu_i[\rho]\right),\,$$

and the corresponding state after the measurement, also called 'subselected' postmeasurement state, is

$$\rho_i = \mu_i[\rho]/p_i.$$

If the outcome of the measurement is not known, then the resultant state, simply called postmeasurement state, is the statistical mixture

$$\sum_{i} p_{i}\rho_{i} = \sum_{i} \mu_{i}[\rho] =: \mu[\rho].$$
(14)

It is also useful to think about how the result of this quantum measurement is stored in a classical register. If we think of an orthonormal basis  $\{|i\rangle\}$  corresponding to the measurement outcomes  $\{i\}$ , the classical register will read  $|i\rangle\langle i|$  when the outcome *i* occurs. Furthermore, if the outcome is not known, then the system composed of the output quantum system and the output classical register will be overall in the state

$$\sum_{i} p_{i} \rho_{i} \otimes |i\rangle \langle i| = \sum_{i} \mu_{i}[\rho] \otimes |i\rangle \langle i|.$$

Consequently, by ignoring the measured quantum system, i.e., by tracing it out, we get the following classical output

$$\sum_{i} p_{i} |i\rangle \langle i| = \sum_{i} \operatorname{Tr} \left( \mu_{i}[\rho] \right) |i\rangle \langle i| =: M[\rho].$$
(15)

The map *M* transforming an arbitrary quantum state  $\rho$  into the classical state  $M[\rho]$  is referred to as the *quantum-to-classical map* (or *measurement map*) corresponding to the quantum instrument  $\mathcal{M}$  with subnormalised maps  $\{\mu_i\}$ .

We note that in order to know the probabilities  $p_i$  of each outcome *i* we do not need to know all the details of  $\mathcal{M}$ . Conversely, we just need to know how the dual  $\mu_i^*$  of every subnormalised map acts on the identity. Indeed,

$$p_i = \operatorname{Tr}(\mu_i[\rho]) = \operatorname{Tr}(\mu_i[\rho]\mathbb{I}) = \operatorname{Tr}(\rho\mu_i^*[\mathbb{I}]).$$

The operators  $M_i := \mu_i^*[\mathbb{I}]$  form a so-called positive operator valued measure (POVM), i.e., a set of positive semidefinite operators that sum up to the identity,  $\sum_i M_i = \mathbb{I}$ .

On the other hand, we must stress that, in general, given only a POVM  $\{M_i\}$  we can just determine the probabilities of the outcomes but we do not know how the quantum system transforms. For example, one can easily see that a given POVM  $\{M_i\}$  is compatible with infinitely many quantum instruments, such as all the ones that can be defined as follows:  $\mu_i[\rho] = m_i \rho m_i^{\dagger}$ , where the  $m_i$ 's are such that  $M_i = m_i^{\dagger} m_i$ . Indeed, it is clear that such  $m_i$ 's are not uniquely defined, but rather also any  $m_i' = U_i m_i$ , with  $U_i$  being arbitrary unitaries, are such that  $M_i = m_i'^{\dagger} m_i'$ .

A special type of measurement is a projective measurement, for which the maps  $\mu_i$  are Hermitian orthogonal projectors, i.e., they satisfy the constraint  $\mu_i\mu_j = \delta_{ij}\mu_i$ . Furthermore, rank one measurements are those that consist only of rank one subnormalised maps  $\mu_i$ .

Let us now consider a quantum system composed of two subsystems *A* and *B*. A *local* generalised measurement (herein called an LGM) is a generalised measurement acting either only on *A* or only on *B* or separately on both *A* and *B*, and whose subnormalised maps are, respectively, either  $\{(\tilde{\pi}_A)_a \otimes \mathbb{I}_B\}$  or  $\{\mathbb{I}_A \otimes (\tilde{\pi}_B)_b\}$  or  $\{(\tilde{\pi}_A)_a \otimes (\tilde{\pi}_B)_b\}$ . We shall use a tilde from now on to denote an LGM, whereas the lack of a tilde will be reserved for the particular case of a *local projective measurement* (LPM).

The probability of obtaining the outcome *a* when performing an LGM only on *A* is

$$p_a = \operatorname{Tr} \left\{ \left( (\tilde{\pi}_A)_a \otimes \mathbb{I}_B \right) [\rho_{AB}] \right\},\$$

with corresponding subselected post-measurement state

$$\rho_{AB|a} = \left\lfloor \left( (\tilde{\pi}_A)_a \otimes \mathbb{I}_B \right) \left[ \rho_{AB} \right] \right] / p_a$$

For each outcome *a*, one can associate a vector in an orthonormal basis  $\{|a\rangle_{A'}\}$  representing a classical register *A'* (note that this basis may live in a different Hilbert space to that of subsystem *A*). By ignoring the state of the measured subsystem *A*, the resulting state of the classical register *A'* and of the unmeasured subsystem *B* is then  $|a\rangle\langle a|_{A'} \otimes \rho_{B|a}$ , where  $\rho_{B|a} = \text{Tr}_A(\rho_{AB|a})$ . Hence, if the result is not known, this composite system will be in the classical-quantum state

$$\tilde{\Pi}_{A}[\rho_{AB}] := \sum_{a} p_{a} |a\rangle \langle a|_{A'} \otimes \rho_{B|a}.$$
(16)

Likewise, for an LGM on both subsystems A and B, the probability of getting respectively the outcomes a and b is

$$p_{ab} = \operatorname{Tr} \left\{ \left( (\tilde{\pi}_A)_a \otimes (\tilde{\pi}_B)_b \right) [\rho_{AB}] \right\}$$

with corresponding subselected post-measurement state

$$\rho_{AB|ab} = \left[ \left( (\tilde{\pi}_A)_a \otimes (\tilde{\pi}_B)_b \right) [\rho_{AB}] \right] / p_{ab}$$

Again, associating to each outcome *a* one vector of the orthonormal basis  $\{|a\rangle_{A'}\}$  corresponding to a classical register *A'* and to each outcome *b* one vector of the orthonormal basis  $\{|b\rangle_{B'}\}$  corresponding to a classical register *B'*, the composite classical register reads  $|a\rangle\langle a|_{A'} \otimes |b\rangle\langle b|_{B'}$  when both outcomes *a* and *b* are recorded. By ignoring the state of the measured subsystems *A* and *B*, the resulting state of the classical registers *A'* and *B'* is then  $|a\rangle\langle a|_{A'} \otimes |b\rangle\langle b|_{B'}$ . Hence, if the result of the LGM is unknown, the system composed of the two classical registers will be in the classical-classical state

$$\tilde{\Pi}_{AB}[\rho_{AB}] := \sum_{ab} p_{ab} |a\rangle \langle a|_{A'} \otimes |b\rangle \langle b|_{B'} .$$
(17)

Analogous definitions can be obtained for LPMs by just removing the tilde. Notice that, according to Naimark's theorem, every generalised measurement can be represented as a projective measurement on a larger system [34]. In particular, this means that we can think of the action of an LGM applied only to subsystem A as equivalent to the action of an LPM applied only to a subsystem A' whose Hilbert space is obtained by extending the Hilbert space of A into a larger one. Similarly, we can think of the action of an LGM applied separately to both subsystems A and B as equivalent to the action of an LPM applied separately to both subsystem A' and B' whose Hilbert spaces are obtained by extending, respectively, the Hilbert spaces of A and B into larger ones. See e.g. [33] for a more detailed account.

Let us then consider in detail the relevant case of measuring  $\rho_{AB}$  via a so-called complete rank one LPM, also known as a local von Neumann measurement, whose subnormalised maps form a complete set of orthogonal rank one projectors, i.e.,  $(\pi_A)_a[\rho_A] = |a\rangle \langle a|_A \rho_A |a\rangle \langle a|_A$ and  $(\pi_B)_b[\rho_B] = |b\rangle \langle b|_B \rho_B |b\rangle \langle b|_B$ , with  $\{|a\rangle_A\}$  and  $\{|b\rangle_B\}$  orthonormal bases and  $\rho_A$  and  $\rho_B$ arbitrary states for subsystem A and B respectively. Then, it is immediate to see that the post-measurement states can be written as

$$\pi_{A}[\rho_{AB}] := \sum_{a} (|a\rangle \langle a|_{A} \otimes \mathbb{I}_{B}) \rho_{AB} (|a\rangle \langle a|_{A} \otimes \mathbb{I}_{B})$$
$$= \sum_{a} p_{a} |a\rangle \langle a|_{A} \otimes \rho_{B|a}, \qquad (18)$$

$$\pi_{AB}[\rho_{AB}] := \sum_{ab}^{a} (|a\rangle \langle a|_{A} \otimes |b\rangle \langle b|_{B}) \rho_{AB} (|a\rangle \langle a|_{A} \otimes |b\rangle \langle b|_{B})$$
$$= \sum_{ab}^{ab} p_{ab} |a\rangle \langle a|_{A} \otimes |b\rangle \langle b|_{B} , \qquad (19)$$

i.e., any state  $\rho_{AB}$  (even if initially entangled or generally non-classical) is *mapped* into a classical-quantum or a classical-classical state, after such a complete rank one LPM acting on subsystem A or on both subsystems A and B, respectively. This simple observation captures the fundamental perturbing role of local (von Neumann) quantum measurements on general states of composite systems.

Indeed, since the action of any complete LPM  $\pi_A$  on any input state  $\rho_{AB}$  results in a classical-quantum post-measurement state of the form given in Eq. (18), if a state  $\rho_{AB}$  is invariant under such an operation it must be necessarily classical-quantum. This means that the only states left invariant by a complete rank one LPM on subsystem A are classical-quantum states. Conversely, for every classical-quantum state  $\rho_{AB} \in C_A$  as defined in Eq. (6), there exists a complete rank one LPM on subsystem A which leaves such a state invariant, defined precisely as the one with a complete set of orthogonal rank one projectors given by  $(\pi_A)_i[\rho_A] = |i\rangle \langle i|_A \rho_A |i\rangle \langle i|_A$ . It thus holds that

$$\rho_{AB} \in \mathcal{C}_A \quad \Leftrightarrow \quad \exists \text{ a complete rank one LPM on } A \text{ such that } \pi_A[\rho_{AB}] = \rho_{AB}.$$
 (20)

A similar argument for classical-classical states and LPMs on both subsystems implies that

 $\rho_{AB} \in \mathcal{C}_{AB} \quad \Leftrightarrow \quad \exists \text{ a complete rank one LPM on } A \text{ and } B \text{ such that } \pi_{AB}[\rho_{AB}] = \rho_{AB}.$  (21)

These two statements formalise the defining Property 2 given earlier for classical states, and illustrated in Fig. 3(a). Moreover, they also motivate why they are called classically correlated states: because it is possible to perform a local von Neumann measurement on them that leaves such states unperturbed.

Even more, since the state of a measured subsystem becomes itself diagonal in the basis in which the complete rank one LPM is performed, we have that the corresponding post-measurement state  $\pi_A[\rho_{AB}]$  (resp.,  $\pi_{AB}[\rho_{AB}]$ ) and output of the quantum-to-classical map  $\Pi_A[\rho_{AB}]$  (resp.,  $\Pi_{AB}[\rho_{AB}]$ ) can be considered equivalent.

Overall, we can thus locally access the information about a system in a classical state by probing the subsystem via a von Neumann measurement without inducing a global disturbance (in contrast to the general quantum mechanical scenario). This latter feature reveals that the correlations that are left after a complete rank one LPM are akin to those described by classical probability theory, hence justifying the terminology for the corresponding output states. 2.3.2. Non-creation of entaglement with an apparatus Let us proceed by analysing in more detail the workings of an LPM according to von Neumann's model [32]. Given a bipartite system in the initial state  $\rho_{AB}$ , any LPM on subsystem A with a complete set of orthogonal rank one projectors  $(\pi_A)_a[\rho_A] = |a\rangle \langle a|_A \rho_A |a\rangle \langle a|_A$  can be realised by letting subsystem A interact via a unitary operation  $V_{AA'}^{[[a]]}$  with an ancillary system A' initialised in a reference pure state, say  $|0\rangle_{A'}$  in its computational basis; in this case the ancillary system A' has the same dimension as A and plays the role of a measurement apparatus [35]. The state of the three parties A, B, A' after the interaction is known as the pre-measurement state

$$\rho_{ABA'}^{\langle [|a\rangle_A]} := (V_{AA'}^{\{|a\rangle_A\}} \otimes \mathbb{I}_B)(\rho_{AB} \otimes |0\rangle \langle 0|_{A'})(V_{AA'}^{\{|a\rangle_A\}} \otimes \mathbb{I}_B)^{\dagger} .$$

$$(22)$$

The LPM is completed by a readout of the apparatus A' in its eigenbasis, in such a way that

$$\operatorname{Tr}_{A'}\rho_{ABA'}^{\prime||a\rangle_{A}} = \sum_{a} \left( |a\rangle \langle a|_{A} \otimes \mathbb{I}_{B} \right) \rho_{AB} \left( |a\rangle \langle a|_{A} \otimes \mathbb{I}_{B} \right) = \pi_{A}[\rho_{AB}] \,.$$
(23)

Imposing Eq. (23), one finds that the pre-measurement interaction between A and A' has to be of a very specific form (up to a local unitary on A'), namely that of an isometry

$$V_{AA'}^{\{|a\rangle_A\}} |a\rangle_A |0\rangle_{A'} = |a\rangle_A |a\rangle_{A'} .$$
(24)

We can always think of  $V_{AA'}^{\{|a\rangle_A\}}$  as the combination  $V_{AA'}^{\{|a\rangle_A\}} = C_{AA'}(U_A^{\{|a\rangle_A\}} \otimes \mathbb{I}_{A'})$  of a local unitary  $U_A^{\{|a\rangle_A\}}$  on A, which serves the purpose of determining in which basis  $\{|a\rangle_A\}$  the subsystem is going to be measured (i.e., which observable is being probed), followed by a generalised controlled-not gate  $C_{AA'}$ , whose action on the computational basis  $|i\rangle_A |i'\rangle_{A'}$  of  $\mathbb{C}^d \otimes \mathbb{C}^d$  is  $C_{AA'}|i\rangle_A |i'\rangle_{A'} = |i\rangle_A |i \oplus i'\rangle_{A'}$ , with  $\oplus$  denoting addition modulo d [36, 28].

In case of a joint LPM on both subsystems A and B, the pre-measurement state reads

$$\rho_{ABA'B'}^{\prime[|a\rangle_{A},|b\rangle_{B}]} := (V_{AA'}^{\{|a\rangle_{A}\}} \otimes V_{BB'}^{\{|b\rangle_{B}\}})(\rho_{AB} \otimes |0\rangle \langle 0|_{A'} \otimes |0\rangle \langle 0|_{B'})(V_{AA'}^{\{|a\rangle_{A}\}} \otimes V_{BB'}^{\{|b\rangle_{B}\}})^{\dagger},$$
(25)

where B' is an additional ancilla of the same dimension as B, which plays the role of an apparatus measuring B in the basis  $\{|b\rangle_B\}$ .

In general, the pre-measurement state  $\rho_{ABA'}^{\prime[|a|\rangle}$  ( $\rho_{ABA'B'}^{\prime[|a|\rangle_A,|b|\rangle_B}$ ) is entangled across the split AB : A'(AB : A'B'), meaning that the ancilla(e) become entangled with the whole system AB due to the pre-measurement interaction(s); it is indeed because of the presence of such entanglement that the system AB is mapped to a classical state upon tracing over the ancillary system(s), an act which generally amounts to an irreversible information loss similar to a decoherence phenomenon, perceived as a disturbance on the state of the system due to the local measurement(s) [37]. However, we know that classical states can be locally measured without perturbation. In the present context, this means that for classical states there is at least one local measurement basis such that the corresponding pre-measurement state remains separable across the system versus ancilla(e) split. Formally, it has been proven in [35, 36, 28] that

$$\rho_{AB} \in \mathcal{C}_A \quad \Leftrightarrow \quad \exists \text{ a local basis } \{|a\rangle_A\} \text{ such that } \rho_{ABA'}^{\langle \{|a\rangle_A\}} \in \mathcal{S}_{AB:A'},$$
(26)

$$\rho_{AB} \in \mathcal{C}_{AB} \quad \Leftrightarrow \quad \exists \text{ local bases } \{|a\rangle_A, |b\rangle_B\} \text{ such that } \rho_{ABA'B'}^{\langle [a\rangle_A, |b\rangle_B\}} \in S_{AB:A'B'}, \quad (27)$$

where we denote by  $S_{X:Y}$  the set of separable states, Eq. (3), with the explicit indication of the considered bipartition X : Y. These two statements formalise the defining Property 3 of classical states, as illustrated in Fig. 3(b). Equivalently, one may say that classical correlations are not always *activated* into entanglement by the act of local measurements [36, 38].

2.3.3. Invariance under a local non-degenerate unitary The possibility of invariance under a non-trivial local unitary evolution is another distinctive characteristic of classical states. Consider a local unitary  $U_A^{\Gamma}$  with a spectrum  $\Gamma$  that acts on subsystem A of a bipartite system in the state  $\rho_{AB}$ . With a slight abuse of notation, we can define the transformed state after the action of such local unitary as  $U_A^{\Gamma}[\rho_{AB}]$ , with

$$U_{A}^{\Gamma}[\rho_{AB}] := \left(U_{A}^{\Gamma} \otimes \mathbb{I}_{B}\right) \rho_{AB} \left(U_{A}^{\Gamma} \otimes \mathbb{I}_{B}\right)^{\mathsf{T}}.$$
(28)

In general, if we fix a non-degenerate spectrum  $\Gamma$ , the transformed state  $U_A^{\Gamma}[\rho_{AB}]$  will be different from the initial  $\rho_{AB}$ . However, if, and only if,  $\rho_{AB}$  is classical-quantum, then one can find at least one local non-degenerate unitary which leaves this state invariant. Clearly, such a unitary  $U_A^{\Gamma}$  is characterised by having its eigenbasis given precisely by the orthonormal basis  $\{|i\rangle_A\}$  entering the definition of the classical-quantum state  $\rho_{AB}$  as in Eq. (6). Formally, it thus holds that [39, 40, 41]

 $\rho_{AB} \in \mathcal{C}_A \quad \Leftrightarrow \quad \exists \text{ a local unitary } U_A^{\Gamma} \text{ with non-degenerate } \Gamma \text{ such that } U_A^{\Gamma}[\rho_{AB}] = \rho_{AB},$ (29)

which formalises the defining Property 4 of classical states, as illustrated in Fig. 3(c).

The spectrum  $\Gamma$  must be non-degenerate to exclude such trivialities as the choice of  $\mathbb{I}_A$  as a local unitary (which would leave any state, not only those classical-quantum, invariant). Notice that, in contrast with the case of a local measurement discussed in Section 2.3.1, a local unitary does not alter the correlations between *A* and *B* at all, yet can assess their nature based on the global change (or lack thereof) of the state  $\rho_{AB}$  in *response* to the action of any such local unitary [41].

Curiously, unlike all the other definitions discussed in the current section, a necessary and sufficient condition analogous to Eq. (29) does not hold for classical-classical states in this case. Namely, considering also a local unitary  $U_B^{\Gamma}$  acting on *B* with the same spectrum  $\Gamma$ , and defining the joint action of  $U_A^{\Gamma}$  and  $U_B^{\Gamma}$  on  $\rho_{AB}$  as

$$U_{AB}^{\Gamma}[\rho_{AB}] := \left(U_{A}^{\Gamma} \otimes U_{B}^{\Gamma}\right) \rho_{AB} \left(U_{A}^{\Gamma} \otimes U_{B}^{\Gamma}\right)^{\dagger}, \qquad (30)$$

then if  $\Gamma$  is non-degenerate it still holds that

 $\rho_{AB} \in \mathcal{C}_{AB} \implies \exists \text{ local unitaries } U_A^{\Gamma}, U_B^{\Gamma} \text{ with non-degenerate } \Gamma \text{ such that } U_{AB}^{\Gamma}[\rho_{AB}] = \rho_{AB},$ (31)

but the reverse implication is no longer true. In fact, there are even entangled states  $\rho_{AB}$  which may be left unchanged by a joint non-trivial local unitary evolution; an example is the maximally entangled two-qubit Bell state  $|\Phi\rangle_{AB} = \frac{1}{\sqrt{2}}(|0\rangle_A |0\rangle_B + |1\rangle_A |1\rangle_B)$ , which is invariant under the action of any tensor product of two matching Pauli matrices,

$$(\sigma_{iA} \otimes \sigma_{iB}) |\Phi\rangle \langle \Phi|_{AB} (\sigma_{iA} \otimes \sigma_{iB}) = |\Phi\rangle \langle \Phi|_{AB}, \qquad (32)$$

for i = 1, 2, 3 (notice that the Pauli matrices are unitary and Hermitian and with nondegenerate spectrum  $\Gamma = \{\pm 1\}$ ). It is an open question how to give a suitable local unitary based condition characterising the set of classical-classical states, with one possibility being to fix a different spectrum between A and B, or to involve multiple local unitaries.

2.3.4. Incoherence in a local basis The previous analysis reveals that classical-quantum states commute with at least a local unitary  $U_A^{\Gamma}$  on subsystem A. An alternative way of

describing this is by saying that classical-quantum states are locally *incoherent* with respect to the eigenbasis of  $U_A^{\Gamma}$ . To be more precise, let us recall some basic terminology used in the characterisation of quantum coherence, see e.g. [42, 43, 44, 45].

Quantum coherence represents superposition with respect to a fixed reference orthonormal basis. In the case of a single system, fixing a reference basis  $\{|i\rangle\}$ , the corresponding set  $\mathcal{I}^{\{|i\rangle\}}$  of incoherent states (i.e., states with vanishing coherence) is defined as the set of those states diagonal in the reference basis,

$$\mathcal{I}^{\{|i\rangle\}} := \left\{ \rho \mid \rho = \sum_{i} p_{i} |i\rangle \langle i| \right\},$$
(33)

with  $\{p_i\}$  a probability distribution. Any other state  $\rho \notin \mathcal{I}^{\{|i\rangle\}}$  exhibits quantum coherence with respect to the fixed reference basis  $\{|i\rangle\}$ .

Consider now a bipartite system *AB*. We can fix local reference bases  $\{|a\rangle_A\}$  for subsystem *A* and  $\{|b\rangle_B\}$  for subsystem *B*, and characterise coherence with respect to these. In particular, we can define two different sets of locally incoherent states. Namely, the set  $\mathcal{J}_A^{\{|a\rangle_A\}}$  of incoherent-quantum (or incoherent on *A*) states is defined as [46, 47]

$$\mathcal{J}_{A}^{\{|a\rangle_{A}\}} := \left\{ \rho_{AB} \mid \rho_{AB} = \sum_{a} p_{a} \mid a \rangle \langle a \mid_{A} \otimes \rho_{B}^{(a)} \right\}, \tag{34}$$

for any quantum states  $\{\rho_B^{(a)}\}$  of subsystem *B*, with  $\{p_a\}$  a probability distribution. Similarly, the set  $\mathcal{I}_{AB}^{\{[a]_A,[b]_B\}}$  of incoherent-incoherent (or incoherent on *A* and *B*) states is defined as [48, 49, 44]

$$\mathcal{I}_{AB}^{\{|a\rangle_{A},|b\rangle_{B}\}} := \left\{ \rho_{AB} \mid \rho_{AB} = \sum_{ab} p_{ab} \left|a\right\rangle \left\langle a\right|_{A} \otimes \left|b\right\rangle \left\langle b\right|_{B} \right\},\tag{35}$$

with  $\{p_{ab}\}$  a joint probability distribution. Notice that in the above definitions the reference bases are fixed, which makes Eqs. (34) and (35) different from the definitions of the sets of classical-quantum and classical-classical states, Eqs. (7) and (11), respectively. However, it is straightforward to realise that a state  $\rho_{AB}$  is classical-quantum (classical-classical) if, and only if, there exist a local basis for A (and for B) with respect to which  $\rho_{AB}$  is incoherent on A (and B). In formulae,

$$\rho_{AB} \in \mathcal{C}_A \quad \Leftrightarrow \quad \exists \text{ a local basis } \{|a\rangle_A\} \text{ such that } \rho_{AB} \in \mathcal{J}_A^{\{|a\rangle_A\}}, \tag{36}$$

$$\rho_{AB} \in \mathcal{C}_{AB} \quad \Leftrightarrow \quad \exists \text{ local bases } \{|a\rangle_A, |b\rangle_B\} \text{ such that } \rho_{AB} \in \mathcal{J}_{AB}^{\{|a\rangle_A, |b\rangle_B\}}.$$
 (37)

These characterisations amount to the defining Property 5 of classical states, as illustrated in Fig. 3(d). In the present paradigm, classicality is thus pinned down as a very intuitive notion, intimately related to the lack of quantum superposition in a specific local reference frame.

2.3.5. Invariance under an entanglement-breaking channel Finally, a classical state can be identified as a fixed point of an entanglement-breaking channel [50]. Entanglement-breaking channels [51] acting on subsystem A of a bipartite system are defined as those CPTP operations  $\Lambda_A^{\text{EB}}$  which map any state  $\rho_{AB}$  to a separable state, i.e.,

$$\Lambda_A^{\text{EB}}[\rho_{AB}] := (\Lambda_A^{\text{EB}} \otimes \mathbb{I}_B)[\rho_{AB}] \in \mathcal{S}_{AB} \quad \forall \rho_{AB} \in \mathcal{D}_{AB}.$$
(38)

Equivalently, the entanglement-breaking channels are all and only the maps that can be written as the concatenation of a local measurement followed by a preparation [51], i.e.,

$$\Lambda_A^{\text{EB}}[\rho_A] := \sum_i \operatorname{Tr}(M_i \rho_A) \rho_A^{(i)},\tag{39}$$

where the  $\{M_i\}$  constitute a POVM while the  $\{\rho_A^{(i)}\}$  are arbitrary quantum states of subsystem A. In other words, an entanglement-breaking channel applied to the subsystem state  $\rho_A$  can be simulated classically as follows: a sender performs an LGM with POVM elements  $\{M_i\}$  on subsystem A and sends the outcome i, which occurs with probability  $p_i = \text{Tr}(M_i\rho_A)$ , via a classical channel to a receiver, who then prepares the subsystem A in a corresponding prearranged state  $\rho_A^{(i)}$  [51].

It then turns out that a state  $\rho_{AB}$  is classical-quantum if, and only if, it is left invariant by at least one entanglement-breaking channel acting on subsystem A [50], i.e.,

 $\rho_{AB} \in \mathcal{C}_A \quad \Leftrightarrow \quad \exists \text{ an entanglement-breaking channel } \Lambda_A^{\text{EB}} \text{ such that } \Lambda_A^{\text{EB}}[\rho_{AB}] = \rho_{AB}.$ (40)

To see this, let us first assume that  $\rho_{AB} \in C_A$ , i.e.,  $\rho_{AB}$  is defined as in Eq. (6); we have then from Section 2.3.1 that such a state is invariant under the complete LPM with rank one projectors  $\{|i\rangle \langle i|_A\}$ , which is a particular case of an entanglement-breaking channel. On the other hand, if there exists at least one entanglement-breaking channel  $\Lambda_A^{EB}$  such that  $\Lambda_A^{EB}[\rho_{AB}] = \rho_{AB}$ , then also the channel  $\overline{\Lambda}_A^{EB}$ , defined as  $\overline{\Lambda}_A^{EB} = \lim_{N\to\infty} \frac{1}{N} \sum_{n=1}^N \left( \Lambda_A^{EB} \right)^n$ , leaves  $\rho_{AB}$  invariant. In [52] it has been shown that this map is an entanglement-breaking channel as well, whose action can be written as follows,  $\overline{\Lambda}_A^{EB}[\rho_A] = \sum_i \operatorname{Tr}(M_i \rho_A) \sigma_A^{(i)}$ , where the  $\{M_i\}$ form a POVM while the  $\{\sigma_A^{(i)}\}$  are states of subsystem A with orthogonal support. Therefore the channel  $\overline{\Lambda}_A^{EB}$  transforms  $\rho_{AB}$  into a classical-quantum state, and since  $\rho_{AB}$  is invariant with respect to the action of this channel, then  $\rho_{AB}$  must be classical-quantum itself.

Analogously, one can see that a state  $\rho_{AB}$  is classical-classical if, and only if, it is left invariant by at least one entanglement-breaking channel  $\Lambda_{AB}^{EB} := \Lambda_{A}^{EB} \otimes \Lambda_{B}^{EB}$  acting on both subsystems *A* and *B* [50], that is,

$$\rho_{AB} \in \mathcal{C}_{AB} \quad \Leftrightarrow \quad \exists \text{ an entanglement-breaking channel } \Lambda_{AB}^{EB} \text{ such that } \Lambda_{AB}^{EB}[\rho_{AB}] = \rho_{AB}.$$
(41)

Notice that, apart from the special case of local measurements, entanglement-breaking channels do not generally destroy the QCs content in a state  $\rho_{AB}$ , as they only vanquish its entanglement. Nevertheless, similarly to the case of local unitaries in Section 2.3.3, it is the global change (or lack thereof) of the state  $\rho_{AB}$  under the action of any entanglement-breaking channel that discerns the nature of the correlations in  $\rho_{AB}$ . Unlike the case of local unitaries, though, the invariance under suitable entanglement-breaking channels allows us to characterise both sets of classical-quantum and classical-classical states. The two statements in Eqs. (40) and (41) thus formalise the defining Property 6 of classical states, as illustrated in Fig. 3(e).

## 2.4. Defining and characterising quantumly correlated states

Having familiarised ourselves with classically correlated states and their physical characterisations, we can now formally define *quantumly correlated* states in a composite system — the true object of our interest — as simply those which are *not* classically correlated. Any state which is not a classical state will be said to possess some form of QCs. Specifically, we shall adopt the following terminology, in line with the majority of the current literature:

- A state ρ<sub>AB</sub> ∉ C<sub>A</sub> is said to have *one-sided* QCs, or equivalently is *quantumly correlated* with respect to subsystem A, or in short quantum on A;
- A state  $\rho_{AB} \notin C_{AB}$  is said to have *two-sided* QCs, or equivalently is *quantumly correlated* with respect to either subsystem A or B, or in short is quantum on A or B.

A brief remark on the semantics is in order. Here the attributes 'one-sided' and 'two-sided' are motivated by the set of states on which the corresponding QCs vanish, namely one-sided QCs are intended as those which vanish on one-sided classical states (alias classical-quantum states) belonging to the set  $\mathcal{C}_A$ , while two-sided QCs are intended as those which vanish on two-sided classical states (alias classical-classical-classical states) belonging to the set  $\mathcal{C}_{AB}$ . In this sense, note that the set of states with two-sided QCs as defined above is strictly larger than the set of states with one-sided QCs, which in turn is strictly larger than the set of states  $\rho_{AB} \notin \mathcal{C}_A \cup \mathcal{C}_B$ , which are quantumly correlated with respect to *both* subsystems A and B, or in short quantum on A and B.

In general, to be precise, QCs capture the genuinely quantum character of the marginal subsystems of a composite quantum system, a feature which can yet manifest only in case the subsystems are correlated. In this paradigm, in fact, any state of a single system is necessarily regarded as classical, since it can always be diagonalised in its orthonormal eigenbasis (hence being invariant under a complete projective measurement in such eigenbasis).

Referring to the analysis of Section 2.3, by negation of the various defining Properties 1–6 of classical states, we can hence provide the following equivalent definitions of bipartite states possessing QCs, as illustrated schematically in the right column of Fig. 3.

- **Definition 1:** A state  $\rho_{AB}$  has QCs if it is not a classical state, i.e., if  $\rho_{AB} \notin C_A$  then  $\rho_{AB}$  has one-sided QCs [negation of Eq. (6)], whilst if  $\rho_{AB} \notin C_{AB}$  then  $\rho_{AB}$  has two-sided QCs [negation of Eq. (10)].
- **Definition 2:** A state  $\rho_{AB}$  has QCs if it is always perturbed by any local complete rank one projective measurement, i.e., if for every complete rank one LPM on *A* it holds  $\pi_A[\rho_{AB}] \neq \rho_{AB}$  then  $\rho_{AB}$  has one-sided QCs [negation of Eq. (20)], whilst if for every complete rank one LPM on both *A* and *B* it holds  $\pi_{AB}[\rho_{AB}] \neq \rho_{AB}$  then  $\rho_{AB}$  has two-sided QCs [negation of Eq. (21)]; see Fig. 3(a).
- **Definition 3:** A state  $\rho_{AB}$  has QCs if it always becomes entangled with an apparatus during any local complete rank one projective measurement, i.e., if for every local basis  $\{|a\rangle_A\}$  the pre-measurement state  $\rho_{ABA'}^{\prime \{|a\rangle_A\}} \notin S_{AB:A'}$  then  $\rho_{AB}$  has one-sided QCs [negation of Eq. (26)], whilst if for every local bases  $\{|a\rangle_A, |b\rangle_B\}$  the pre-measurement state  $\rho_{ABA'B'}^{\prime \{|a\rangle_A, |b\rangle_B} \notin S_{AB:A'B'}$  then  $\rho_{AB}$  has two-sided QCs [negation of Eq. (27)]; see Fig. 3(b).
- **Definition 4:** A state  $\rho_{AB}$  has QCs if it is always altered by any local non-degenerate unitary, i.e. if for every local unitary  $U_A^{\Gamma}$  with non-degenerate spectrum  $\Gamma$  it holds  $U_A^{\Gamma}[\rho_{AB}] \neq \rho_{AB}$ then  $\rho_{AB}$  has one-sided QCs [negation of Eq. (29)]; see Fig. 3(c).
- **Definition 5:** A state  $\rho_{AB}$  has QCs if it exhibits quantum coherence in all local bases, i.e., if for every local basis  $\{|a\rangle_A\}$  it holds  $\rho_{AB} \notin \mathcal{J}_A^{[|a\rangle_A]}$  then  $\rho_{AB}$  has one-sided QCs [negation of Eq. (36).], whilst if for every local bases  $\{|a\rangle_A, |b\rangle_B\}$  it holds  $\rho_{AB} \notin \mathcal{J}_{AB}^{[|a\rangle_A, |b\rangle_B\}}$  then  $\rho_{AB}$  has two-sided QCs [negation of Eq. (37)]; see Fig. 3(d).
- **Definition 6:** A state  $\rho_{AB}$  has QCs if it is always altered by any local entanglement-breaking channel, i.e. if for every entanglement-breaking channel  $\Lambda_A^{\text{EB}}$  on *A* it holds  $\Lambda_A^{\text{EB}}[\rho_{AB}] \neq \rho_{AB}$  then  $\rho_{AB}$  has one-sided QCs [negation of Eq. (40)], whilst if for every entanglement-breaking channel  $\Lambda_{AB}^{\text{EB}} = \Lambda_A^{\text{EB}} \otimes \Lambda_B^{\text{EB}}$  on both *A* and *B* it holds  $\Lambda_{AB}^{\text{EB}}[\rho_{AB}] \neq \rho_{AB}$  then  $\rho_{AB}$  has two-sided QCs [negation of Eq. (41)]; see Fig. 3(e).

In the next Section, we promote each of the above definitions from the qualitative to the quantitative domain, and report on some of the most up-to-date progress available on the quest for the quantification of QCs in bipartite quantum systems.

## 3. Measures of quantum correlations

In the previous Section we have discussed the defining characteristics of quantumly correlated states. It is a natural question now to ask whether one can meaningfully compare the QCs present in a pair or more of such states. This is achieved by imposing a quantitative ordering on the set of quantum states according to a suitable *measure* of QCs. However, as it is often the case in quantum information theory [16], it turns out that there is not a unique, universal measure, but rather different measures resulting from different physical motivations and imposing different orderings on the set of quantum states.

In this Section we will review a variety of such measures. Our approach will be thematic more than historic, and guided by the general intuition that a measure of QCs in a state  $\rho_{AB}$  should capture essentially how much  $\rho_{AB}$  deviates from being a classical state. Having introduced different ways to characterise classical states in the previous Section, in the following we will accordingly classify existing (and potentially future) measures of QCs into suitably different categories, each reflecting a particular signature of QCs, in accordance with the guiding scheme of Fig. 3. We will accompany our discussion with several Tables, useful to summarise our notation as well as the main properties of the measures and comparisons between them, and we will also link in with the applications of some representatives of these measures to quantum information processing and beyond, given later in Section 4.

Before delving into the zoo of QCs measures, however, we must consider what the requirements are for a given quantifier to be a true *bona fide* measure of QCs. Namely, while a measure must naturally incarnate the spirit of QCs, there are as well some important supplementary conditions, or potentially welcome properties, that any given quantifier should satisfy for mathematical and physical consistency. Henceforth we proceed by first discussing such a set of conditions.

## 3.1. Requirements for a bona fide measure of quantum correlations

Despite the intense research drive into characterising QCs over the past few years, the completion of the rule book of criteria for a given quantifier to be a valid measure of QCs is still an open question, with no clear consensus in the literature. Various proposals for mandatory, optional and desirable requirements for a measure of QCs have been discussed e.g. in [35, 53, 12, 54, 27, 33, 55, 56], to which the reader is referred for additional details.

Here we focus on an essential set of five Requirements that reflect (in our opinion) some very natural *desiderata* that any measure of QCs should obey. While the first four such Requirements can be regarded as reasonably well established, the fifth Requirement is only a proposed one, whose validation or reconsideration will necessitate further research.

A one-sided (asymmetric) measure of QCs  $Q_A(\rho_{AB})$  on bipartite quantum states  $\rho_{AB}$  must be a real, non-negative function satisfying the following Requirements:

**Requirement (i):**  $Q_A(\rho_{AB}) = 0$  if  $\rho_{AB} \in \mathcal{C}_A$  is a classical-quantum state as defined in Eq. (6); **Requirement (ii):**  $Q_A(\rho_{AB})$  is invariant under local unitaries, i.e.  $Q_A((U_A \otimes U_B)\rho_{AB}(U_A^{\dagger} \otimes U_B^{\dagger})) = Q_A(\rho_{AB})$ , for any state  $\rho_{AB}$  and any local unitary operation  $U_A$  and  $U_B$  acting on subsystem *A* and *B*, respectively;

**Requirement (iii):**  $Q_A(\rho_{AB})$  reduces to a measure of entanglement  $E(\rho_{AB})$  on pure states, i.e.  $Q_A(|\psi\rangle_{AB}) = E(|\psi\rangle_{AB})$  for any pure bipartite state  $|\psi\rangle_{AB}$ .

Similar constraints hold for a two-sided (symmetric) measure of QCs  $Q_{AB}(\rho_{AB})$ , where one simply substitutes the classical-quantum set  $\mathcal{C}_A$  in Requirement (i) with the classical-classical set  $\mathcal{C}_{AB}$  defined in Eq. (11).

#### Measures and applications of quantum correlations

Requirement (i) formalizes the basic requirement that a valid measure of QCs must vanish on classical (i.e., free) states. We anticipate that all the measures reviewed in this article are in fact *faithful*, meaning that they vanish on and *only* on classical states. Requirement (ii) ensures that any measure of QCs is locally basis independent, as expected for any quantifier of correlations [7]. Moreover, as already discussed, all forms of non-classical correlations reduce to entanglement for pure states, hence one expects any valid measure of QCs to reduce to a corresponding valid measure of entanglement in such a special case. This is imposed by Requirement (iii), which has appeared frequently in recent literature on QCs [27, 54, 57, 58, 59, 41, 55, 56]

Another important consideration is how a measure of QCs should behave under the action of local quantum channels. For example, any valid entanglement measure must be non-increasing on average under the action of LOCC [3], so what are the analogous constraints for a measure of general QCs? A strict Requirement is the following, first formalised in [35],

**Requirement (iv):**  $Q_A(\rho_{AB})$  is monotonically non-increasing under any local operation on the party whose quantumness is not being measured (in our convention subsystem *B*), that is,  $Q_A((\mathbb{I}_A \otimes \Lambda_B)[\rho_{AB}]) \leq Q_A(\rho_{AB})$  for any state  $\rho_{AB}$  and any CPTP map  $\Lambda_B$  on *B*.

This Requirement guarantees that it must be physically impossible to increase the quantumness degree of party A by operating on party B alone. Obviously, an analogous constraint cannot be stated for two-sided measures  $Q_{AB}(\rho_{AB})$  of QCs. However, it remains to analyse which operations on A (or also on B in the case of two-sided measures) may or should not lead to a creation or an increase of QCs in a bipartite state  $\rho_{AB}$ . To help with this task, one can view QCs as a resource within the framework of *resource theories* [10, 60, 61].

In a resource theory, the *free states* and the *free operations* are identified as, respectively, the states without any resource content and the operations that are freely implementable, i.e., that map free states into free states. The resource theory framework then imposes that any valid measure of a resource should vanish on the free states and should be monotonically non-increasing under the action of the corresponding free operations, or in other words that one can only produce resource by implementing operations that themselves cost resource. For example, turning again to entanglement, the free states are the separable states of Eq. (2), the free operations are the LOCC, and any entanglement measure must be monotonically non-increasing under LOCC [62, 16, 3]. It is thus clear that the freely implementable operations associated to QCs must be identified in order to impose a similar monotonicity constraint.

Unfortunately, there is no consensus yet on the free operations for an eventual resource theory of QCs [10]. What is fully characterised, though, is the maximal set of local operations that cannot create a state with QCs from a classical (free) state. This set is constituted by so-called local commutativity preserving operations (LCPO), defined as follows. A CPTP map  $\Lambda$  is commutativity preserving if

$$[\Lambda[\rho], \Lambda[\sigma]] = 0 \quad \forall \ \rho, \sigma \text{ such that } [\rho, \sigma] = 0, \tag{42}$$

with  $[\cdot, \cdot]$  denoting the commutator. In a bipartite system, a CPTP map  $\Lambda_A^{\text{LCPO}} := \Lambda_A \otimes \mathbb{I}_B$  is LCPO on subsystem *A* if  $\Lambda_A$  is commutativity preserving. Similarly, a CPTP map  $\Lambda_{AB}^{\text{LCPO}} := \Lambda_A \otimes \Lambda_B$  is LCPO on subsystems *A* and *B* if both  $\Lambda_A$  and  $\Lambda_B$  are commutativity preserving. It then holds that [63]

$$(\Lambda_A \otimes \mathbb{I}_B)[\rho_{AB}] \in \mathcal{C}_A \quad \forall \rho_{AB} \in \mathcal{C}_A \quad \Leftrightarrow \quad \Lambda_A \text{ is commutativity preserving,}$$
(43)  
$$(\Lambda_A \otimes \Lambda_B)[\rho_{AB}] \in \mathcal{C}_{AB} \quad \forall \rho_{AB} \in \mathcal{C}_{AB} \quad \Leftrightarrow \quad \Lambda_A \text{ and } \Lambda_B \text{ are commutativity preserving.}$$
(44)

In other words, LCPO are all and only the admissible local channels, acting on the subsystem(s) whose quantumness is being measured, that preserve the set of classical states.

Any other operation, even if locally performed on party A only, can create QCs out of a classically correlated state. This may seem counterintuitive, as local operations alone cannot create entanglement or even classical correlations instead [7]. However, if we recall the meaning of QCs as discussed in Section 2.4, we can convince ourselves that implementing local operations which involve, say, the creation of coherence on A, may consequently map a classical state out of its set, breaking the conditions of Section 2.3; the possibility of local creation of QCs was first studied in [64, 65]. It becomes important, then, to provide a more detailed characterisation of the LCPO set, for which such a possibility is denied. It was shown in [65] that, if subsystem A is a qubit, LCPO on A amount either to unital channels,  $\Lambda_A[\mathbb{I}_A] = \mathbb{I}_A$ , i.e. preserving the maximally mixed state, or to completely decohering maps,  $\Lambda_A[\rho_A] = \sum_i p_i^{(\rho_A)} |i\rangle \langle i|_A$ , i.e. mapping any  $\rho_A$  to a state diagonal in some basis  $\{|i\rangle_A\}$ with probabilities  $\{p_i^{(\rho_A)}\}$ . It was further shown in [63, 66] that, if instead subsystem A is a qudit with dimension  $d_A > 2$ , LCPO on A can be either isotropic channels,  $\Lambda_A[\rho_A] =$  $t\Upsilon[\rho_A] + (1-t)\mathrm{Tr}(\rho_A)\frac{\mathbb{I}}{d_A}$ , with  $\Upsilon$  being either a unitary operation or unitarily equivalent to transposition (see [67] for further details on these channels, including their Kraus operator sum representation), or completely decohering maps.

Any physically consistent set of free local operations for QCs should therefore be contained into the set of LCPO, or coincide with it. In entanglement theory, for example, LOCC are singled out by the physical paradigm of distant laboratories, and form a strict subset of the maximal set of operations preserving separable states [68] (as the latter also contain certain nonlocal operations such as the swap gate), while in the resource theories of quantum coherence there are currently several alternative proposals for free operations, with different characterisations, and a still ongoing debate on their physical consistency [69, 43, 70, 44, 71, 45, 72, 73].

Here, following [55], we postulate a *strong monotonicity* desideratum for measures of QCs under the maximal set of local operations preserving classical states, given by

## **Requirement** (v): $Q_A(\rho_{AB})$ is monotonically non-increasing under any LCPO, that is, $Q_A(\Lambda_A^{\text{LCPO}}[\rho_{AB}]) \leq Q_A(\rho_{AB})$ , for any state $\rho_{AB}$ and any LCPO $\Lambda_A^{\text{LCPO}}$ on subsystem A.

A similar desideratum is postulated for a two-sided measure of QCs  $Q_{AB}(\rho_{AB})$ , where one then considers LCPO  $\Lambda_{AB}^{\text{LCPO}}$  acting on both subsystems *A* and *B*. Notice that Requirement (v) is only a proposed one, and should not be regarded yet as mandatory, unlike the previous four Requirements. Nonetheless, it makes sense to consider it for the following reasons.

If a measure of QCs is proven to obey Requirement (v), then it will be a valid monotone in any resource theory of QCs with free operations restricted to local channels. On the other hand, in case an otherwise valid quantifier is found to fail Requirement (v), it may still happen to be monotonic under a smaller (as yet uncharacterised) set of local operations, which might form the basis for a physically consistent resource theory of QCs, with an accordingly modified Requirement (v). In particular, since local completely decohering operations on one or both subsystems always map any state  $\rho_{AB}$  to a classical state belonging respectively to  $C_A$ or  $C_{AB}$ , every measure of QCs which obeys Requirement (i) is automatically a monotone under local completely decohering operations. For any valid measure, therefore, testing Requirement (v) amounts to verifying monotonicity under local unital channels when the dimension of the measured party (or parties) is 2, and under local isotropic channels when the dimension of the measured party (or parties) is higher.

Let us now briefly mention a couple of additional potential desiderata for measures of QCs. First, *continuity* has been analysed in [53]: this captures the reasonable requirement that QCs should not change too much if a state is slightly perturbed. Moreover, we recall that our Requirements (ii) and (iv) together imply that measures of one-sided QCs  $Q_A$  are maximal

Туре	One-sided	Two-sided	Section
Geometric	$Q^{G_\delta}_A$	$Q^{G_\delta}_{AB}$	3.2.1, 3.3
Measurement induced geometric	$Q^{M_\delta}_A$	$Q^{M_\delta}_{AB}$	3.2.2, 3.4
Measurement induced informational	$Q^{I_\epsilon}_A$	$Q^{I_\epsilon}_{AB}$	3.2.3, 3.5
Entanglement activation	$Q_A^{E^\zeta}$	$Q^{E^{\zeta}}_{AB}$	3.2.4, 3.6
Unitary response	$Q_A^{U_\delta}$	_	3.2.5, 3.7
Coherence based	$Q^{C^\eta}_A$	$Q^{C^\eta}_{AB}$	3.2.6, 3.8
Recoverability	$Q^{R_\delta}_A$	$Q^{R_\delta}_{AB}$	3.2.7, 3.9

Table 1. The various types of measures of QCs, their notation and location within this review.

on pure maximally entangled states if  $\dim(\mathcal{H}_A) \leq \dim(\mathcal{H}_B)$  [74, 33]. The authors of [56] additionally postulate that measures of one-sided QCs should be maximal *only* on such states. Notice however that these properties are no longer valid for measures of two-sided QCs  $Q_{AB}$ , which (quite surprisingly) can reach higher values on mixed partially entangled states than on pure maximally entangled states [36].

Let us further remark that, while *convexity* is an optional property that one may wish to have for entanglement measures [16] (even though there exist valid entanglement monotones that are not convex [75]), general measures of QCs are (and should be) neither convex nor concave. In fact, while mixing e.g. a quantum state  $\rho_{AB}$  with the maximally mixed one  $\propto \mathbb{I}_{AB}$  typically reduces the QCs content of  $\rho_{AB}$ , one can also create nonzero QCs by mixing two classical states, provided they are locally diagonal in two different bases.

In what follows, we shall adopt the minimal set of (four plus a potential one) Requirements (i)-(v) introduced above, and aim to validate QCs quantifier against them.

## 3.2. Types of measures of quantum correlations

We begin our exploration of the zoology of established measures of QCs by introducing a classification which groups them into different types, based upon how the QCs in an arbitrary bipartite state  $\rho_{AB}$  are captured conceptually by the characterisations of Section 2.4 (see Fig. 3), as summarised in Table 1.

- Geometric measures (Section 3.2.1) quantify how distant  $\rho_{AB}$  is from the set of classical states, cf. Fig. 2(a).
- Measurement induced geometric measures (Section 3.2.2) quantify how distinguishable  $\rho_{AB}$  is from the output of a least perturbing local measurement, as in Fig. 3(a).
- Measurement induced informational measures (Section 3.2.3) quantify how much information about  $\rho_{AB}$  is lost due to a least perturbing local measurement, another take on Fig. 3(a).
- Entanglement activation measures (Section 3.2.4) quantify the minimum entanglement created with an apparatus in a local pre-measurement, as in Fig. 3(b).
- Unitary response measures (Section 3.2.5) quantify how distinguishable  $\rho_{AB}$  is from its image after a least perturbing local non-degenerate unitary, as in Fig. 3(c).
- Coherence based measures (Section 3.2.6) quantify the minimum quantum coherence of  $\rho_{AB}$  in any local basis, as in Fig. 3(d).
- **Recoverability measures** (Section 3.2.7) quantify how distant  $\rho_{AB}$  is from being a fixed point of an entanglement-breaking channel, as in Fig. 3(e).

In the following subsections, we first discuss general properties of each type of QCs measures, such as their compliance with the aforementioned Requirements, and later specialise to particular instances of measures within each category and how they interlink across the categories. For each reviewed measure, we will discuss some of the history, experimental investigations, and progress on their analytical or numerical evaluation, where relevant. We will also highlight any operational interpretations linking in with Section 4. Finally, a schematic summary of most of the reviewed measures will be presented in Table 6.

Note that we will focus mainly on finite dimensional bipartite systems in this review. However, most of the presented measures have also been extended to continuous variable systems of two or more quantum harmonic oscillators (or modes of the radiation field), and studied in particular in the case of two-mode Gaussian states. Deferring the reader to [76] (and references therein) for a recent review on continuous variable Gaussian states including some investigations of their QCs, we will briefly mention a few relevant results in the following, where deemed necessary.

3.2.1. Geometric measures Definition 1 gives a negative characterisation of quantumly correlated states, i.e. quantumly correlated states are all the states that are not classical. The most natural quantitative extension of this is to consider the distance to the set of classical states as a measure of QCs. Given a distance  $D_{\delta}$ , we can then define *geometric* measures of one-sided and two-sided QCs of a bipartite state  $\rho_{AB}$  as follows,

$$Q_{A}^{G_{\delta}}(\rho_{AB}) := \inf_{\chi_{AB} \in \mathcal{C}_{A}} D_{\delta}(\rho_{AB}, \chi_{AB}),$$

$$Q_{AB}^{G_{\delta}}(\rho_{AB}) := \inf_{\chi_{AB} \in \mathcal{C}_{AB}} D_{\delta}(\rho_{AB}, \chi_{AB}),$$
(45)

where  $\delta$  labels the particular distance functional  $D_{\delta}$  on the set of states  $\mathcal{D}_{AB}$ , and the superscript  $G_{\delta}$  indicates that the measures are of the geometric type with respect to this chosen distance.

By construction, the geometric measures are zero only for classical states, hence Requirement (i) is satisfied. Requirements (ii), (iv), and (v) are also satisfied for any choice of  $D_{\delta}$  that is *contractive* under the action of CPTP operations, i.e. such that

$$D_{\delta}(\Lambda[\rho], \Lambda[\sigma]) \le D_{\delta}(\rho, \sigma), \tag{46}$$

for any two states  $\rho$  and  $\sigma$  and any CPTP operation A. Although contractivity may not be necessary to satisfy these properties, there is good reason to restrict to contractive distances. Distances are used to quantify the distinguishability between quantum states, and noninvertible CPTP operations are the representation of noise [77]. Intuitively, one expects noise to never increase the distinguishability of quantum states, hence the distances that contract through any CPTP operation are the mathematical realisation of this. We also note that, in a resource theoretic context, any contractive distance will induce a geometric measure of QCs that is monotonically non-increasing under the action of the free operations, regardless of their exact specification, purely by virtue of the fact that the free operations cannot create QCs. Table 2 summarises a selection of widely adopted contractive distances in quantum information theory, whose corresponding geometric QCs measures will be discussed in Section 3.3.

Finally, there is a subtlety regarding Requirement (iii) for the geometric measures of QCs. To each geometric measure of QCs one can associate a corresponding measure of entanglement, defined in terms of the distance to the separable states. Specifically, for

Measures and applications of quantum correlations

Distance	$D_\delta$	Formula	Contractive
Relative entropy	$D_{\rm RE}$	$\mathcal{S}(\rho \  \sigma) = \operatorname{Tr}(\rho \log \rho - \rho \log \sigma)$	Yes
( <i>p</i> <sup>th</sup> power) Schatten <i>p</i> -norm	$D_p$	$\ \rho - \sigma\ _p^p = \operatorname{Tr}( \rho - \sigma ^p)$	Only for $p = 1$
Infidelity	$D_F$	$1 - \left[ \operatorname{Tr} \left( \sqrt{\sqrt{\sigma} \rho \sqrt{\sigma}} \right) \right]^2$	Yes
Squared Bures	$D_{\mathrm{Bu}}$	$2\left[1 - \operatorname{Tr}\left(\sqrt{\sqrt{\sigma}\rho \sqrt{\sigma}}\right)\right]$	Yes
Squared Hellinger	$D_{\mathrm{He}}$	$2\left[1 - \operatorname{Tr}\left(\sqrt{\rho}\sqrt{\sigma}\right)\right]^{2}$	Yes

**Table 2.** Some distances  $D_{\delta}$  between two quantum states  $\rho$  and  $\sigma$ . Details on the definitions and discussions of contractivity can be found in [77, 78, 79].

a bipartite state  $\rho_{AB}$ , the geometric measure of entanglement given a particular choice of distance  $D_{\delta}$  is [16]

$$E^{G_{\delta}}(\rho_{AB}) := \inf_{\sigma_{AB} \in \mathcal{S}_{AB}} D_{\delta}(\rho_{AB}, \sigma_{AB}), \tag{47}$$

where  $S_{AB}$  is the set of separable states defined in Eq. (3). Because of the hierarchy of Eq. (12), as depicted in Fig. 2(a), it generally holds that

$$Q_{AB}^{G_{\delta}}(\rho_{AB}) \ge Q_{A}^{G_{\delta}}(\rho_{AB}) \ge E^{G_{\delta}}(\rho_{AB}).$$
(48)

For pure states  $|\psi\rangle_{AB}$ , one would naturally expect that

$$Q_{AB}^{G_{\delta}}(|\psi\rangle_{AB}) = Q_{A}^{G_{\delta}}(|\psi\rangle_{AB}) = E^{G_{\delta}}(|\psi\rangle_{AB}), \tag{49}$$

because the set of pure classical and pure separable states all reduce to the set of pure product states. However, since the infima in Eqs. (45) and (47) are, respectively, over all (not necessarily pure) classical and separable states, Eq. (49) does not hold trivially. Instead, one must show Requirement (iii) on a case-by-case basis for each choice of distance  $D_{\delta}$ .

3.2.2. Measurement induced geometric measures Another intuitive aspect of QCs, outlined by Definition 2, is that a quantumly correlated state is perturbed by any possible complete rank one LPM. If we consider a complete rank one LPM on subsystem A, we know that  $\pi_A[\rho_{AB}]$ as defined in Eq. (18) is necessarily different to  $\rho_{AB}$  if QCs are present. It is then meaningful to think that, provided one identifies the least perturbing LPM  $\pi_A$ , the more distinguished  $\pi_A[\rho_{AB}]$  is from  $\rho_{AB}$ , the more QCs were present originally in  $\rho_{AB}$ . By quantifying this distinguishability as the distance between  $\rho_{AB}$  and the closest (i.e., least perturbed) output state  $\pi_A[\rho_{AB}]$ , we can define the measurement induced geometric measures of QCs of a state  $\rho_{AB}$ . Given a distance  $D_{\delta}$ , these are defined as

$$Q_{A}^{M_{\delta}}(\rho_{AB}) := \inf_{\pi_{A}} D_{\delta}(\rho_{AB}, \pi_{A}[\rho_{AB}]),$$

$$Q_{AB}^{M_{\delta}}(\rho_{AB}) := \inf_{\pi_{AB}} D_{\delta}(\rho_{AB}, \pi_{AB}[\rho_{AB}]),$$
(50)

where  $M_{\delta}$  denotes the measurement induced geometric type for a corresponding distance  $D_{\delta}$ , and we have considered LPMs on both subsystems A and B as defined in Eq. (19) in the case of two-sided measures.

Once more, we can choose any distance on the set of quantum states to define a measure within this family, with some common examples listed in Table 2. The minimisation must be performed only over complete rank one LPMs, which represent the most fine grained

Informational quantity	$I_{\epsilon}$	Formula
(negative) von Neumann entropy	Is	$-\mathcal{S}(\rho_{AB}) = \operatorname{Tr}\left(\rho_{AB}\log\rho_{AB}\right)$
Mutual information	$I_I$	$I(\rho_{AB}) = S(\rho_A) + S(\rho_B) - S(\rho_{AB})$

**Table 3.** Some informational quantifiers  $I_{\epsilon}$  of a bipartite state  $\rho_{AB}$  with reduced states  $\rho_A = \text{Tr}_B(\rho_{AB})$  and  $\rho_B = \text{Tr}_A(\rho_{AB})$ .

projective measurements since each projector corresponds to a particular outcome and no outcomes are merged into a higher rank projector (more trivially, without restricting to rank one projectors one would always satisfy the infima in Eq. (50) by choosing as projectors, respectively,  $\mathbb{I}_A$  and  $\mathbb{I}_A \otimes \mathbb{I}_B$ ). We can use the fact that  $\pi_A[\rho_{AB}] \in \mathbb{C}_A$  and  $\pi_{AB}[\rho_{AB}] \in \mathbb{C}_{AB}$ , as we have already discussed in Eqs. (20) and (21), to see that

$$Q_A^{G_\delta}(\rho_{AB}) \le Q_A^{M_\delta}(\rho_{AB}), \quad \text{and} \quad Q_{AB}^{G_\delta}(\rho_{AB}) \le Q_{AB}^{M_\delta}(\rho_{AB}).$$
(51)

For certain choices of  $D_{\delta}$ , these inequalities are in fact saturated, and we will highlight when this occurs in the following (Sections 3.3 and 3.4). It is immediate to see that Requirement (i) for measurement induced geometric measures is satisfied because a state can be left invariant under a complete rank one LPM if, and only if, it is classical. Requirements (ii) and (iv) also hold for any contractive distance, however it is unknown whether this is true for Requirement (v) as well. Again, Requirement (iii) must be checked for each particular distance  $D_{\delta}$ .

3.2.3. Measurement induced informational measures An alternative viewpoint from which we can look at QCs is based on how much information about the state of a composite system can be extracted by accessing it locally. When only classical correlations are present, we can in principle extract all such information after a local measurement, while this is no longer the case when also QCs come into play. As discussed in Section 2.3.1, LGMs  $\Pi_A$  and  $\Pi_{AB}$  represent recording the measurement information about  $\rho_{AB}$  onto classical registers. It is intuitive, therefore, to adopt an information theoretic perspective and regard the minimal difference between the original information in the state  $\rho_{AB}$  and the postmeasurement information stored in the classical register as a measure of quantumness. Given an informational quantifier  $I_{\epsilon}$ , we can then define measurement induced informational (or simply informational) measures of QCs of a state  $\rho_{AB}$  as

$$Q_{A}^{\tilde{I}_{\epsilon}}(\rho_{AB}) := \inf_{\tilde{\Pi}_{A}} \left[ I_{\epsilon}(\rho_{AB}) - I_{\epsilon} \left( \tilde{\Pi}_{A}[\rho_{AB}] \right) \right],$$

$$Q_{AB}^{\tilde{I}_{\epsilon}}(\rho_{AB}) := \inf_{\tilde{\Pi}_{AB}} \left[ I_{\epsilon}(\rho_{AB}) - I_{\epsilon} \left( \tilde{\Pi}_{AB}[\rho_{AB}] \right) \right].$$
(52)

A list of the most prominent choices of  $I_{\epsilon}$  is presented in Table 3. We elaborate more on these choices in Section 3.5, also highlighting the relationships between informational and other types of measures of QCs.

The measures in Eq. (52) obey Requirement (i) since for all and only the classical states there is always an LPM (which is within the class of all LGMs) that leaves them unchanged. However, one should be careful when choosing  $I_{\epsilon}$  to ensure that  $Q_{AB}^{\tilde{I}_{\epsilon}}(\rho_{AB})$  and  $Q_{AB}^{\tilde{I}_{\epsilon}}(\rho_{AB})$  are non-negative (hence the choice of the negative von Neumann entropy in Table 3). Requirement (ii) holds for any  $I_{\epsilon}$  invariant under local unitaries, which in turn should always hold as a measure of information or correlations is not expected to be dependent upon the

local bases chosen. However, Requirements (iii), (iv) and (v) must be checked on a case-bycase basis for each selection of  $I_{\epsilon}$ . Notice also that for some choices of  $I_{\epsilon}$  one may need to impose further restrictions to the allowed LGMs in the optimisations appearing in Eqs. (52) in order to avoid trivial results, as we discuss in Section 3.5.1.

We can explicitly consider the special case of informational measures with optimisation restricted over complete rank one LPMs, hence defining

$$Q_A^{I_{\epsilon}}(\rho_{AB}) := \inf_{\Pi_A} \left[ I_{\epsilon} \left( \rho_{AB} \right) - I_{\epsilon} \left( \Pi_A [\rho_{AB}] \right) \right],$$
  

$$Q_{AB}^{I_{\epsilon}}(\rho_{AB}) := \inf_{\Pi_{AB}} \left[ I_{\epsilon} \left( \rho_{AB} \right) - I_{\epsilon} \left( \Pi_{AB} [\rho_{AB}] \right) \right].$$
(53)

It then holds by construction that

$$Q_A^{\tilde{l}_{\epsilon}}(\rho_{AB}) \le Q_A^{l_{\epsilon}}(\rho_{AB}), \quad \text{and} \quad Q_{AB}^{\tilde{l}_{\epsilon}}(\rho_{AB}) \le Q_{AB}^{l_{\epsilon}}(\rho_{AB}).$$
(54)

To be more precise, invoking Naimark's theorem as in Section 2.3.1, we recall that an LGM on  $\rho_{AB}$  can equivalently be performed by embedding  $\rho_{AB}$  into a larger space and carrying out an LPM. This means that

$$Q_A^{I_\epsilon}(\rho_{AB}) = \inf_{A'} Q_{A'}^{I_\epsilon}(\rho_{A'B}), \quad \text{and} \quad Q_{AB}^{I_\epsilon}(\rho_{AB}) = \inf_{A',B'} Q_{A'B'}^{I_\epsilon}(\rho_{A'B'}), \tag{55}$$

or in words that the one-sided LGM based informational measures are obtained as a minimisation of the corresponding one-sided LPM based informational measures over all embeddings of A into a larger system A', and similarly for the two-sided measures considering embeddings of both A and B in larger systems A' and B', respectively.

Taking care again to ensure non-negativity, the LPM subclass of informational measures obey Requirement (i) and Requirement (ii) if  $I_{\epsilon}$  is invariant under local unitaries, but all the other requirements must be checked for each  $I_{\epsilon}$  on a case-by-case basis. The meaning of the complete rank one LPM based informational measures in Eqs. (53) is even more intuitive. Indeed, by recalling that the output of the quantum-to-classical maps  $\Pi_A$  (resp.,  $\Pi_{AB}$ ) can be considered equivalent to the post-measurement states  $\pi_A$  (resp.,  $\pi_{AB}$ ), we have that such informational measures capture directly the minimum amount of change in the information content of the state  $\rho_{AB}$  itself (without the need for invoking a classical register) before and after a local von Neumann measurement. This reproduces the original spirit of the quantum discord [6], which can be classified indeed as an informational measure, as we show in Section 3.5.

3.2.4. Entanglement activation measures Definition 3 suggests that measures of QCs can be readily defined from measures of entanglement. In particular, the QCs content of a state  $\rho_{AB}$  can be quantified in terms of the minimum entanglement created between the system AB and an ancillary system, given by A' for the one-sided case, or by the composite A'B' for the two-sided case, in the pre-measurement state corresponding to a least perturbing complete rank one LPM on A or on both A and B, respectively. Given a specific bipartite entanglement monotone  $E^{\zeta}$ , this conceptual framework initially put forward in [36, 35] defines the entanglement activation measures of QCs of a state  $\rho_{AB}$ ,

$$Q_{A}^{E^{\zeta}}(\rho_{AB}) := \inf_{\{|a\rangle_{A}\}} E_{AB:A'}^{\zeta} \left( \rho_{ABA'}^{\prime \{|a\rangle_{A}\}} \right),$$

$$Q_{AB}^{E^{\zeta}}(\rho_{AB}) := \inf_{\{|a\rangle_{A},|b\rangle_{B}\}} E_{AB:A'B'}^{\zeta} \left( \rho_{ABA'B'}^{\prime \{|a\rangle_{A},|b\rangle_{B}\}} \right),$$
(56)

Measures and applications of quantum correlations

Entanglement measure	$E^{\zeta}$	Formula
Geometric measure(s)	$E^{G_{\delta}}$	$E^{G_{\delta}}(\rho_{AB}) = \inf_{\sigma_{AB} \in \mathfrak{S}_{AB}} D_{\delta}(\rho_{AB}, \sigma_{AB})$
Distillable entanglement	$E^d$	See Ref. [16, 3]
Entanglement negativity	$E^N$	$E^N(\rho_{AB}) = \ \rho_{AB}^{T_A}\ _1 - 1$
Entanglement of formation	$E^f$	$E^{f}(\rho_{AB}) = \inf \sum_{i} p_{i} \mathcal{S}(\operatorname{Tr}_{B}( \psi_{i}\rangle\langle\psi_{i} _{AB}))$
		$\{p_i,  \psi_i\rangle_{AB}\},$
		$\rho_{AB} = \sum_i p_i  \psi_i\rangle \langle \psi_i  _{AB}$

**Table 4.** Some entanglement monotones  $E^{\zeta}$  for a bipartite state  $\rho_{AB}$ .

where the optimisation is over the local bases of the pre-measurement interaction as detailed in Section 2.3.2, and the pre-measurement states  $\rho_{ABA'}^{\langle [a]_A \rangle}$  and  $\rho_{ABA'B'}^{\langle [a]_A \rangle [b]_B \rangle}$  are specifically defined in Eqs. (22) and (25); notice that we indicate the explicit bipartition X : Y with respect to which entanglement is being calculated with the notation  $E_{X:Y}^{\zeta}(\rho_{XY})$ .

One of the key strengths of this approach is that there is a huge variety of entanglement measures  $E^{\zeta}$  available in the quantum information literature [16] (we list a small selection of choices in Table 4): for each of those, Eqs. (56) define corresponding measures of more general one-sided and two-sided QCs. Crucially, it turns out that this correspondence respects the hierarchy of non-classical correlations illustrated in Figs. 1 and 2(a). Indeed, it was proven in [28] that

$$Q_{AB}^{E^{\zeta}}(\rho_{AB}) \ge Q_{A}^{E^{\zeta}}(\rho_{AB}) \ge E_{A:B}^{\zeta}(\rho_{AB}),$$
(57)

for all bipartite states  $\rho_{AB}$  and any  $E^{\zeta}$  monotonically non-increasing under LOCC. This follows because the original state  $\rho_{AB}$  can be obtained from the pre-measurement states in Eqs. (56) by means of an LOCC with respect to the system versus ancilla(e) split [28]. The inequalities in Eq. (57) all turn into equalities for pure states  $|\psi\rangle_{AB}$ , which means that Requirement (iii) is satisfied for all entanglement activation measures of QCs [28]. It is rightful to interpret Eq. (57) as a quantitative indication that QCs truly go *beyond* entanglement, in a very precise sense, and reduce to it in the case of pure states, as it should be. Requirement (i) holds by construction due to the characterisation of classical states according to Property 3, and Requirements (ii) and (iv) are satisfied as well for any entanglement activation measure of QCs defined from a valid entanglement monotone  $E^{\zeta}$ . Requirement (v) has not been investigated in general and remains to be checked on a case-by-case basis.

We discuss some specific examples of entanglement activation type measures of QCs in Section 3.6.

3.2.5. Unitary response measures As outlined by Definition 4, QCs give also rise to the unavoidable disturbance of a bipartite quantum state in response to the action of non-degenerate unitary operations on one of the subsystems. This qualitative aspect of QCs can be turned into a QCs quantifier by considering how far a given bipartite state  $\rho_{AB}$  is from being invariant under a non-degenerate local unitary operation, i.e., by considering the distinguishability between  $\rho_{AB}$  and its image after a least disturbing non-degenerate local unitary. For a given distance  $D_{\delta}$ , the (one-sided) unitary response measure of QCs of a state  $\rho_{AB}$  is then defined as [39, 40, 80, 41]

$$Q_A^{U_\delta^{\Gamma}}(\rho_{AB}) := \inf_{U_A^{\Gamma}} D_\delta(\rho_{AB}, U_A^{\Gamma}[\rho_{AB}]),$$
(58)

where the optimisation is over all local unitaries  $U_A^{\Gamma}$  with non-degenerate spectrum  $\Gamma$ , as defined in Eq. (28).

#### Measures and applications of quantum correlations

Among all possible choices of the spectrum  $\Gamma$ , a special role is played by the harmonic spectrum  $\Gamma^*$  given by the  $d_A$ -th complex roots of unity, with  $d_A$  being the dimension of subsystem A. The corresponding local unitary operations, also known as root-of-unity operations, will be denoted simply by  $U_A \equiv U_A^{\Gamma^*}$  (dropping the spectrum superscript). They may be expected to be the most informative ones that we can restrict to in order to quantify QCs, in the sense that the more the unitaries in the aforementioned optimisation are fully non-degenerate (with no spacing between any two eigenvalues becoming arbitrarily small) the better the corresponding response measures should be able to discern between different states [81, 39, 40, 41]. For this reason, we will primarily focus on the special case of unitary response measures with optimisation restricted to harmonic spectra,

$$Q_A^{U_\delta}(\rho_{AB}) := \inf_{U_A} D_\delta(\rho_{AB}, U_A[\rho_{AB}]).$$
<sup>(59)</sup>

The unitary response measures of QCs clearly satisfy Requirement (i) since any classicalquantum state is invariant under a non-degenerate local unitary [39, 40, 41]. Moreover, they satisfy Requirement (ii) and, provided  $D_{\delta}$  is contractive, Requirement (iv) [41], whereas it is still an open question if they comply with Requirement (v). For pure states  $|\psi\rangle_{AB}$ , any measure  $Q_A^{U_{\delta}}$  reduces to the corresponding "entanglement of response" [81],

$$E^{U_{\delta}}(|\psi\rangle_{AB}) := \inf_{U_{\delta}} D_{\delta}(|\psi\rangle \langle \psi|_{AB}, U_{A}[|\psi\rangle \langle \psi|_{AB}]), \tag{60}$$

which entails that Requirement (iii) is satisfied as well for any unitary response measure of QCs.

Examples of measures  $Q_A^{U_{\delta}}$  for typical distances  $D_{\delta}$  as given in Table 2, and their connections to geometric and measurement induced geometric measures will be explored in Section 3.7. We finally remark that, by virtue of the discussion in Section 2.3.3, the unitary response approach does not appear to be straightforwardly extendible to define two-sided measures of QCs, as the latter would fail Requirement (i), unlike the case of all the other types of measures covered in this review.

*3.2.6. Coherence based measures* Following Definition 5, QCs can be quantified in terms of measures of quantum coherence, minimised over all local bases. Measures of quantum coherence have been studied quite extensively in the last few years, see e.g. [42, 69, 43, 48, 49, 71, 44, 45, 46, 82, 83, 70, 47] and the very recent review [84]. While several connections between coherence, entanglement, and QCs have been pointed out quite recently, including quantitative interconversion schemes [85, 49, 86, 46, 47] somehow analogous to the entanglement activation framework [36, 35], a consistent definition of QCs quantifiers in terms of coherence has not been reported in full generality (to our knowledge), and will be introduced here.

For a single system, consider a measure of coherence  $C^{\eta}$  with respect to a fixed reference basis { $|i\rangle$ }, that is, a real, non-negative function  $C^{\eta}(\rho)$  which vanishes on, and only on, the corresponding set  $\mathcal{I}^{\{|i\rangle\}}$  of incoherent states (see Section 2.3.4), and is monotonically nonincreasing under a suitable set of incoherent operations. The incoherent operations may be, for instance, those defined in [43], which admit an operator sum representation where all Kraus operators map the set of incoherent states into itself, or a more restricted set such as the so-called covariant operations, which have been defined in the context of the resource theory of asymmetry [69]; we defer the reader to recent studies where merits and setbacks of various proposals for incoherent operations are discussed [70, 44, 71, 45, 72, 73]. We list a sample of coherence measures  $C^{\eta}$  of relevance in various resource theoretic contexts in Table 5.

Measures and applications of quantum correlations

Coherence measure	$C^\eta$	Formula
Relative entropy	$C^{\text{RE}}$	$C^{\text{RE}}(\rho) = \inf_{\chi \in \mathbb{J}^{[ \rho ]}} D_{\text{RE}}(\rho, \chi) = \mathcal{S}(\pi[\rho]) - \mathcal{S}(\rho)$
Infidelity	$C^F$	$C^F( ho) = 1 - \sup_{\chi \in \mathcal{I}^{(h)}} F( ho, \chi)$
$\ell_1$ norm	$C^{\ell_1}$	$C^{\ell_1}(\rho) = \ \rho\ _{\ell_1}^{\{ i\rangle\}} - 1 = \sum_{i \neq j}  \rho_{ij} $
Robustness	$C^{\operatorname{Ro}}$	$C^{\text{Ro}}(\rho) = \min_{\tau} \left\{ s \ge 0 \mid \frac{\rho + s\tau}{1+s} \in \mathcal{I}^{\{ i\rangle\}} \right\}$
Wigner-Yanase skew information	$C^{WY}$	$C^{WY}(\rho, K) = -\frac{1}{2} \operatorname{Tr}\left([\sqrt{\rho}, K]^2\right)$
Quantum Fisher information	$C^{\rm QF}$	$C^{\rm QF}(\rho, K) = 4 \sum_{m < n, q_m + q_n \neq 0} \frac{(q_m - q_n)^2}{q_m + q_n}  \langle \psi_m   K   \psi_n \rangle ^2$

**Table 5.** Some coherence measures for a state  $\rho$  with respect to a reference basis { $|i\rangle$ }. The first four [42, 43, 49, 45] are monotones under the more general incoherent operations of [43], while the last two [82, 83, 70], in which the reference basis is identified as the eigenbasis of the observable  $K = \sum_i k_i |i\rangle\langle i|$  with non-degenerate spectrum  $\Gamma = \{k_i\}$ , are monotones under the more restricted set of phase covariant operations defined in asymmetry theory [69]. Furthermore, in the first row we denote by  $\pi[\rho]$  the full dephasing in the reference basis,  $\pi[\rho] = \sum_i |i\rangle\langle i|\rho |i\rangle\langle i|$ , in the third row we evaluate the  $\ell_1$  norm for  $\rho$  written in the reference basis, in the fourth row we indicate by  $\tau$  an arbitrary state, while in the final row we write  $\rho$  in its spectral decomposition,  $\rho = \sum_n q_n |\psi_n\rangle\langle\psi_n|$ .

Given now a bipartite system AB in the state  $\rho_{AB}$ , and fixing local reference bases  $\{|a\rangle_A\}$  for subsystem A and  $\{|b\rangle_B\}$  for subsystem B as in Section 2.3.4, assume that to a chosen coherence measure  $C^{\eta}$  we can formally associate: a one-sided local coherence quantifier  $C_A^{\eta \mid |a\rangle_A}(\rho_{AB})$  with the requirements of vanishing on (and only on) the set  $\mathcal{J}_A^{\mid |a\rangle_A}$  of incoherent-quantum states defined in Eq. (34), and of being monotonic under (the chosen set of) local incoherent operations on A and arbitrary local operations on B; and a two-sided local coherence quantifier  $C_{AB}^{\eta \mid |a\rangle_A, |b\rangle_B}(\rho_{AB})$  with the requirements of vanishing on (and only on) the set  $\mathcal{J}_{AB}^{\mid |a\rangle_A, |b\rangle_B}$  of incoherent-incoherent states defined in Eq. (35), and of being monotonic under (the chosen set of) local incoherent operations on A and B. In particular, the two-sided quantifier can be defined as the standard extension of  $C^{\eta}$  to bipartite systems, choosing the product basis  $\{|a\rangle_A \otimes |b\rangle_B\}$  as reference basis for the composite AB [48, 49, 44]. Notice however that, depending on the definition of  $C^{\eta}$ , it may not be straightforwardly possible to construct both one-sided and two-sided corresponding quantifiers of local coherence; we will discuss this in more detail in Section 3.8.

For a given coherence quantifier  $C^{\eta}$ , we then define the *coherence based* measures of QCs of a state  $\rho_{AB}$  as

$$\begin{aligned}
\mathcal{Q}_{A}^{C^{\eta}}(\rho_{AB}) &:= \inf_{\{|a\rangle_{A}\}} C_{A}^{\eta \{|a\rangle_{A}\}}(\rho_{AB}), \\
\mathcal{Q}_{AB}^{C^{\eta}}(\rho_{AB}) &:= \inf_{\{|a\rangle_{A}, |b\rangle_{B}\}} C_{AB}^{\eta \{|a\rangle_{A}, |b\rangle_{B}\}}(\rho_{AB}),
\end{aligned}$$
(61)

where the minimisation is over the local reference bases  $\{|a\rangle_A\}$  of A and  $\{|b\rangle_B\}$  of B.

By construction, due to Property 5 and the constraints imposed on the local coherence quantifiers above, coherence based measures of QCs satisfy Requirements (i) and (iv), and also Requirement (ii), which holds manifestly in the two-sided case due to the involved minimisation over local bases on both parties and is implied by (iv) in the one-sided case. However, there are no general results yet concerning Requirements (iii) and (v), which have to be checked on a case-by-case basis.

In Section 3.8 we show how the coherence based approach encompasses some relevant quantifiers of QCs, and discuss their interplay with other types of measures.

3.2.7. Recoverability measures The last facet of QCs that we take into account is the following: when the QCs between two subsystems are small, almost all the information about the overall bipartite system is locally recoverable after performing a measurement on one of the subsystems. Since every entanglement-breaking channel can be written as a sequence of a measurement map followed by a preparation (recovery) map, as discussed in Section 2.3.5, such a facet of QCs can be quantitatively captured by how far the state  $\rho_{AB}$  of the overall bipartite system is from being a fixed point of an entanglement-breaking channel [50], along the lines of Definition 6. Given a distance  $D_{\delta}$ , we can then define the *recoverability* type measures of QCs as

$$Q_{A}^{R_{\delta}}(\rho_{AB}) := \inf_{\Lambda_{AB}^{\text{EB}}} D_{\delta}(\rho_{AB}, \Lambda_{A}^{\text{EB}}[\rho_{AB}]),$$

$$Q_{AB}^{R_{\delta}}(\rho_{AB}) := \inf_{\Lambda_{AB}^{\text{EB}}} D_{\delta}(\rho_{AB}, \Lambda_{AB}^{\text{EB}}[\rho_{AB}]),$$
(62)

where the optimisation is over the convex set of entanglement-breaking channels  $\Lambda_A^{\text{EB}}$  acting locally on *A* and  $\Lambda_{AB}^{\text{EB}}$  acting locally on *A* and on *B*, respectively for one-sided and two-sided measures.

The recoverability based measures comply with Requirement (i) as all and only the classical state are fixed points of some entanglement-breaking channel. They further satisfy Requirements (ii) and (iv) provided  $D_{\delta}$  is a contractive distance, while Requirements (iii) and Requirement (v) have to be tested for each  $D_{\delta}$ . Observe that, given an arbitrary state  $\rho_{AB}$ , the recoverability based measures of Eqs. (62) are in general smaller or equal than the corresponding measurement induced geometric measures of Eqs. (50) defined via the same  $D_{\delta}$ , since every LPM is in particular a local entanglement-breaking channel.

Instances of recoverability measures of QCs as defined in recent literature [50] are presented in Section 3.9.

#### 3.3. Geometric measures

Here we cover in detail geometric measures of QCs, as defined in Eqs. (45), for some representative choices of the distance  $D_{\delta}$ , including those summarised in Table 2.

3.3.1. Relative entropy The first definition of a geometric measure of QCs was given in [87, 88], where the relative entropy  $D_{\text{RE}}$  (see Table 2) was considered. Although not strictly a distance [77], the relative entropy (or Kullback-Leibler divergence) is a widely used tool in classical information theory to measure the distinguishability between two probability distributions, and it is naturally extended to quantum states [89]. The relative entropy is contractive [90, 91], therefore we need only to check Requirement (iii) for the corresponding QCs measures  $Q_A^{G_{\text{RE}}}$  and  $Q_{AB}^{G_{\text{RE}}}$ . The latter is verified, as for pure states  $|\psi\rangle_{AB}$  it holds

$$Q_A^{G_{\text{RE}}}(|\psi\rangle_{AB}) = Q_{AB}^{G_{\text{RE}}}(|\psi\rangle_{AB}) = E^{G_{\text{RE}}}(|\psi\rangle_{AB}) = \mathcal{S}(\rho_A) = \mathcal{S}(\rho_B), \tag{63}$$

where  $S(\rho_A) = S(\rho_B)$  amounts to the canonical measure of entanglement for the pure state  $|\psi\rangle_{AB}$ , known as the entanglement entropy [92, 77], with the marginal state  $\rho_A$  ( $\rho_B$ ) obtained from  $|\psi\rangle\langle\psi|_{AB}$  by partial trace over *B* (*A*).

In [93] it was shown that the closest classical state  $\chi_{AB}$  to any state  $\rho_{AB}$ , which solves the optimisation in Eqs. (45) for the relative entropy distance, is always the one obtained as the output of least perturbing complete rank one LPMs  $\pi_A[\rho_{AB}]$  and  $\pi_{AB}[\rho_{AB}]$  for the onesided and two-sided measures, respectively. This implies that the geometric and measurement induced geometric measures based upon the relative entropy distance coincide, i.e.

$$Q_A^{G_{\text{RE}}}(\rho_{AB}) = Q_A^{M_{\text{RE}}}(\rho_{AB}), \quad \text{and} \quad Q_{AB}^{G_{\text{RE}}}(\rho_{AB}) = Q_{AB}^{M_{\text{RE}}}(\rho_{AB}). \tag{64}$$

Furthermore, there is an agreement with the LPM informational measures based on the negative von Neumann entropy [94, 87] (see Table 3)

$$Q_{A}^{G_{RE}}(\rho_{AB}) = \inf_{\Pi_{A}} \left[ \mathcal{S}(\Pi_{A}[\rho_{AB}]) - \mathcal{S}(\rho_{AB}) \right] = Q_{A}^{I_{S}}(\rho_{AB}),$$
$$Q_{AB}^{G_{RE}}(\rho_{AB}) = \inf_{\Pi_{AB}} \left[ \mathcal{S}(\Pi_{AB}[\rho_{AB}]) - \mathcal{S}(\rho_{AB}) \right] = Q_{AB}^{I_{S}}(\rho_{AB}), \tag{65}$$

as well as with the entanglement activation measures based upon relative entropy and distillable entanglement [35, 36]

$$Q_{A}^{G_{\rm RE}}(\rho_{AB}) = Q_{A}^{E^{G_{\rm RE}}}(\rho_{AB}) = Q_{A}^{E^{d}}(\rho_{AB}), \quad \text{and} \quad Q_{AB}^{G_{\rm RE}}(\rho_{AB}) = Q_{AB}^{E^{G_{\rm RE}}}(\rho_{AB}) = Q_{AB}^{E^{d}}(\rho_{AB}), \quad (66)$$

and also with the coherence based measures in terms of relative entropy  $Q_A^{C^{\text{RE}}}(\rho_{AB})$  and  $Q_{AB}^{C^{\text{RE}}}(\rho_{AB})$ , defined later in Eq. (128) [93, 43, 95, 47].

There is clearly a quite remarkable convergence on the one-sided and two-sided measures  $Q_A^{G_{RE}}$  and  $Q_{AB}^{G_{RE}}$ , which are also known in the literature as "relative entropy of discord" and "relative entropy of quantumness" respectively [87, 88, 93, 36], across a range of different (generally inequivalent) categories of QCs with accordingly different physical interpretations.

As it happens customarily in quantum information theory, however, the best behaved quantities are often the most difficult to compute [96]. In the case of the geometric measures of QCs based on relative entropy,  $Q_A^{G_{RE}}$  and  $Q_{AB}^{G_{RE}}$ , analytical formulae are available only for special classes of states, such as Bell diagonal states of two qubits [97] and highly symmetric (i.e., Werner and isotropic) states of two qudits [98]. For both of these classes of states, one-sided and two-sided measures of QCs coincide.

3.3.2. Hilbert-Schmidt distance By using the 2-norm induced distance  $D_2$  (see Table 2), equal to the squared Hilbert-Schmidt distance, another geometric quantifier of one-sided QCs, corresponding to  $Q_A^{G_2}(\rho_{AB})$  as defined in Eq. (45), was proposed in [25], where a simple closed form of this quantity for arbitrary states of two qubits was also found. Extensions beyond two qubits were subsequently derived, including in particular analytical formulae for all states where subsystem A is a qubit while the dimension of subsystem B is arbitrary [99, 100, 101, 102], and an operational interpretation was provided by linking  $Q_A^{G_2}(\rho_{AB})$  to the task of remote state preparation [103] (see Section 4.1.5 for a critical discussion). It was later shown that [99]

$$Q_A^{G_2}(\rho_{AB}) = Q_A^{M_2}(\rho_{AB}), \tag{67}$$

while, if A is a qubit, it also holds that [39, 40]

$$Q_A^{G_2}(\rho_{AB}) = \frac{1}{4} Q_A^{U_2}(\rho_{AB}).$$
(68)

Experimental quantifications of  $Q_A^{G_2}(\rho_{AB})$  have been carried out [103, 104, 105, 106] and lower bounds provided [107, 108, 109]. The two-sided quantifier  $Q_{AB}^{G_2}$  was defined in [110].

The Hilbert-Schmidt based geometric measure of QCs enjoyed considerable popularity due to its analytical simplicity and became initially known just as the "geometric discord", but an important problem was subsequently identified, due to the fact that the Hilbert-Schmidt distance  $D_2$  is not contractive in general, as mentioned in Table 2. Although  $D_2$  is still unitarily invariant, which implies Requirement (ii), this means that Requirements (iv) and (v) are not automatically satisfied for  $Q_A^{G_2}$ . Indeed, [111] provided a counter argument to show that  $Q_A^{G_2}(\rho_{AB})$  does not obey Requirement (iv), which goes as follows. Consider appending an uncorrelated ancilla state  $\rho_C$  to the bipartite state  $\rho_{AB}$ , it then holds that

$$Q_A^{G_2}(\rho_{AB} \otimes \rho_C) = Q_A^{G_2}(\rho_{AB}) \operatorname{Tr}(\rho_C^2),$$
(69)

due to the multiplicativity of the squared Hilbert-Schmidt norm on tensor products. Since  $\text{Tr}(\rho_C^2)$  is just the purity of  $\rho_C$ , we know that adding/removing an impure ancilla will decrease/increase  $Q_A^{G_2}$ . However, addition and removal of an ancilla is a reversible quantum operation on the unmeasured subsystem *B*, for which we require the one-sided QCs never to increase. Thus,  $Q_A^{G_2}$  does not obey this crucial requirement. One proposed resolution to this particular problem was to rescale  $Q_A^{G_2}(\rho_{AB})$  by the purity of  $\rho_{AB}$  [96], but in this case it is still possible to find other counterexamples showing that Requirement (iv) is violated [112]. Another suggestion given in [113] was to exploit the link between the Hilbert-Schmidt geometric and measurement induced geometric measures in Eq. (67) and instead consider  $Q_A^{M_2}(\sqrt{\rho_{AB}})$ . While maintaining the analytical simplicity of  $Q_A^{G_2}(\rho_{AB})$ , this remedied modification no longer experiences the problem in Eq. (69), and for pure states  $|\psi\rangle_{AB}$  it reduces to  $Q_A^{M_2}(\rho_{AB})$  because  $\sqrt{|\psi\rangle}\langle\psi|_{AB} = |\psi\rangle\langle\psi|_{AB}$ . Finally, Requirement (iii) is satisfied for  $Q_A^{G_2}(\rho_{AB})$ , since for pure states one finds [100]

$$Q_A^{G_2}(|\psi\rangle_{AB}) = 1 - \text{Tr}(\rho_A^2) = 1 - \text{Tr}(\rho_B^2),$$
(70)

which is half of the entanglement monotone known as the I-tangle [114]. The latter reduces to the squared entanglement negativity [115, 116] if A is a qubit, where we define the entanglement negativity  $E^N(\rho_{AB})$  as in Table 4, with the suffix  $T_A$  denoting partial transposition with respect to subsystem A. More generally,  $2Q_A^{G_2}(\rho_{AB}) \ge [E^N(\rho_{AB})]^2$  for all two-qubit states  $\rho_{AB}$  [117], but this is not true anymore for higher dimensional states [118]. Geometric measures of QCs based on the Hilbert-Schmidt distance have been studied for two-mode Gaussian states in [119, 96] (specifically, in the measurement induced geometric approach); however, the only classical Gaussian states are product states [120, 121], which makes any geometric approach to the quantification of their QCs ambiguous (i.e., geometric quantum and total correlations collapse to the same quantity when restricting distances to the subsets  $\mathcal{C}_A$  or  $\mathcal{C}_{AB}$  and  $\mathcal{P}_A$  of Gaussian states, respectively).

3.3.3. Trace distance The trace distance, arising (modulo a  $\frac{1}{2}$  normalisation factor) from the 1-norm induced distance  $D_1$  (see Table 2), has also been adopted to define geometric measures of QCs as in Eqs. (45) [122, 123, 80, 78, 124]. Because  $D_p$  is only contractive for p = 1, the trace distance based geometric measure, alias "trace distance discord", is the only one of its class that directly obeys Requirements (ii), (iv), and (v), hence emerging as the most suitable choice out of all the Schatten  $D_p$  norm induced geometric measures. For pure states  $|\psi\rangle_{AB}$ , only when subsystem A is a qubit it has been shown in [80] that this measure of QCs reduces to the entanglement negativity [115, 116], i.e.

$$Q_A^{G_1}(|\psi\rangle_{AB}) = E^N(|\psi\rangle_{AB}),\tag{71}$$

while the behaviour of  $Q_A^{G_1}(|\psi\rangle_{AB})$  and  $Q_{AB}^{G_1}(|\psi\rangle_{AB})$  for arbitrary pure states is not known, leaving the fulfillment of Requirement (iii) as an open question in general. A closed formula

for  $Q_A^{G_1}(\rho_{AB})$  on two-qubit states with X-shaped density matrices (X states) and two-qubit quantum-classical states was provided in [59], where also the optimisation for general two-qubit states was greatly simplified. It was shown in [80] that, if subsystem A is a qubit, then

$$Q_A^{G_1}(\rho_{AB}) = Q_A^{M_1}(\rho_{AB}) = Q_A^{E^N}(\rho_{AB}) = \frac{1}{2}Q_A^{U_1}(\rho_{AB}).$$
(72)

Operationally, the trace distance is related to the minimum probability of error in the discrimination between states [125]. Experimental investigations of  $Q_A^{G_1}(\rho)$  have been carried out in a two-qubit nuclear magnetic resonance setup [104, 126].

3.3.4. Bures distance The squared Bures distance  $D_{Bu}$ , as well as the monotonically related infidelity  $D_F$  (see Table 2), can induce another pair of geometric measures of QCs defined as in Eqs. (45), that satisfy all of the five Requirements of Section 3.1 [35, 27, 127, 128, 129]. Indeed,  $D_{Bu}$  is contractive, and for pure states  $|\psi\rangle_{AB}$  it holds that

$$Q_{A}^{G_{\rm Bu}}(|\psi\rangle_{AB}) = Q_{AB}^{G_{\rm Bu}}(|\psi\rangle_{AB}) = E^{G_{\rm Bu}}(|\psi\rangle_{AB}) = 2(1 - \sqrt{s_{\rm max}}),\tag{73}$$

where  $s_{\text{max}}$  is the maximum Schmidt coefficient of  $|\psi\rangle_{AB}$  [127], which defines the geometric measure of entanglement [130]. An operational interpretation of  $Q_A^{G_{Bu}}(\rho_{AB})$  can be provided in terms of the optimal success probability of an ambiguous quantum state discrimination task [127], as discussed later in Section 4.2.2. Closed formulae for  $Q_A^{G_{Bu}}(\rho_{AB})$  have been supplied for Bell diagonal states of two qubits [27, 128].

3.3.5. Hellinger distance Similarly, using the squared Hellinger distance  $D_{\text{He}}$  (see Table 2) one obtains as well valid measures of geometric QCs obeying all the Requirements. An in-depth study of  $Q_A^{G_{\text{He}}}(\rho_{AB})$  was carried out in [56], where they simplified the problem of evaluating this measure for arbitrary bipartite states, and showed in particular that for pure states  $|\psi\rangle_{AB}$  it holds

$$Q_A^{G_{\text{He}}}(|\psi\rangle_{AB}) = 2\left(1 - \sqrt{\text{Tr}(\rho_A^2)}\right) = 2\left(1 - \sqrt{\text{Tr}(\rho_B^2)}\right),\tag{74}$$

which is a measure of entanglement [62]. An interesting link to the Hilbert-Schmidt based geometric measure was also established [56],

$$Q_A^{G_{\rm He}}(\rho_{AB}) = 2 - 2\sqrt{1 - Q_A^{G_2}(\sqrt{\rho_{AB}})},\tag{75}$$

which shows that, without compromising its reliability, the squared Hellinger distance based geometric measure maintains the same computational simplicity of the Hilbert-Schmidt distance based geometric measure corrected as in [113], and is thus amenable to a closed analytical formula for all bipartite states  $\rho_{AB}$  when subsystem A is a qubit. Geometric measures of QCs based on the Hellinger distance have been discussed for Gaussian states in [131]. Operationally, the Hellinger distance admits an interpretation in terms of asymptotic state discrimination [132, 56], as detailed in Section 4.2.2.

3.3.6. Hierarchy of geometric measures It is also interesting to compare the various geometric measures of QCs arising from different choices of distance  $D_{\delta}$ . We can use general inequalities between the distances in Table 2 to infer inequalities between the corresponding

geometric measures, i.e. if for two choices of distance  $\delta_1$  and  $\delta_2$  it holds that  $f(D_{\delta_1}(\rho, \sigma)) \leq g(D_{\delta_2}(\rho, \sigma))$  for any  $\rho$  and  $\sigma$  and for two increasing functions f and g, then we know that

$$f\left(\mathcal{Q}_{A}^{G_{\delta_{1}}}(\rho_{AB})\right) \leq g\left(\mathcal{Q}_{A}^{G_{\delta_{2}}}(\rho_{AB})\right), \quad \text{and} \quad f\left(\mathcal{Q}_{AB}^{G_{\delta_{1}}}(\rho_{AB})\right) \leq g\left(\mathcal{Q}_{AB}^{G_{\delta_{2}}}(\rho_{AB})\right). \tag{76}$$

The following inequalities are therefore useful [77, 79, 2, 127, 133, 56]

$$\frac{1}{r}D_{1}(\rho,\sigma)^{2} \leq D_{2}(\rho,\sigma) \leq D_{1}(\rho,\sigma)^{2},$$

$$\frac{1}{2}D_{1}(\rho,\sigma)^{2} \leq D_{\text{RE}}(\rho,\sigma),$$

$$D_{\text{Bu}}(\rho,\sigma) \leq D_{\text{He}}(\rho,\sigma) \leq D_{1}(\rho,\sigma) \leq \sqrt{1 - \left(1 - \frac{1}{2}D_{\text{Bu}}(\rho,\sigma)\right)^{2}},$$
(77)

where *r* is the rank of  $\rho - \sigma$ . These bounds are also applicable to the other types of measures of QCs involving a distance. One can further derive bounds between geometric measures of QCs that do not stem from fundamental inequalities between the corresponding distances. More generally, bounds amongst the different types of measures of QCs can be found, but these are sometimes tailored to the dimension of the relevant subsystems. A wealth of such inequalities are provided in [56].

3.3.7. Other distances There are other possible choices of  $D_{\delta}$  in Eqs. (45) that can be used to define geometric measures of QCs. First of all, note that both the Bures distance and Hellinger distance are closely related, in fact they become identical when considering the distance between pairs of commuting states. Indeed, both are examples of geodesic distances belonging to a family of monotone and Riemannian metrics defined on the set of quantum states, characterised by the Morozova-Čentsov-Petz theorem [134, 135, 77]. These infinitely many monotone and Riemannian metrics can each give a unique contractive geodesic distance that could in principle be used to define a geometric measure of QCs, but finding the geodesic distance in each case is a formidable problem and the information theoretic relevance of each distance would need to be considered.

Generalisations of the relative entropy could also be used to define geometric measures of QCs. Two recently suggested generalisations are the sandwiched relative Rényi entropies [136, 137] (see also [138])

$$D_{\mathcal{R}_{\alpha}}(\rho,\sigma) = \frac{\log \operatorname{Tr}\left[\left(\sigma^{\frac{1-\alpha}{2\alpha}}\rho\sigma^{\frac{1-\alpha}{2\alpha}}\right)^{\alpha}\right]}{\alpha-1}$$
(78)

and the sandwiched relative Tsallis entropies [139]

$$D_{\mathcal{T}_{\alpha}}(\rho,\sigma) = \frac{\operatorname{Tr}\left[\left(\sigma^{\frac{1-\alpha}{2\alpha}}\rho\sigma^{\frac{1-\alpha}{2\alpha}}\right)^{\alpha}\right] - 1}{\alpha - 1}.$$
(79)

These recent variations of the traditional relative Rényi and Tsallis entropies, designed to capture the noncommutative nature of the density matrix, are contractive quasi-distances for  $\alpha \in (1/2, \infty]$  and thus any geometric measure of QCs based upon either choice would naturally obey Requirements (i), (ii), (iv) and (v), with Requirement (iii) needing to be tested for each  $\alpha$ . Certain values of  $\alpha$  recover well known relative entropies, in particular when  $\alpha \rightarrow 1$  we have

$$D_{\mathcal{R}_1}(\rho,\sigma) = D_{\mathcal{T}_1}(\rho,\sigma) = D_{\text{RE}}(\rho,\sigma).$$
(80)

#### 3.4. Measurement induced geometric measures

As we have discussed in Section 3.3, the measurement induced geometric measures of QCs with minimisation over LPMs, defined by Eqs. (50), often have an intuitive crossover with the conventional geometric measures of QCs.

3.4.1. Hilbert-Schmidt and trace distance The measurement induced geometric type of measure was first introduced in [140] for the squared Hilbert-Schmidt distance and has subsequently been primarily studied in parallel with the geometric measures. In particular, due to the link in Eq. (67),  $Q_A^{M_2}(\rho_{AB})$  is not a valid measure of QCs as it fails Requirement (iv), and similar inconsistencies can be expected for  $Q_{AB}^{M_2}(\rho_{AB})$  due to the lack of contractivity of the Hilbert-Schmidt distance  $D_2$ .

While we have coincidence between the one-sided geometric measures and the onesided measurement induced geometric measures of QCs based on the trace distance for a qubit [80], as expressed in Eq. (72), it is still a relevant question to ask whether  $Q_A^{M_1}(\rho_{AB})$  and also  $Q_{AB}^{M_1}(\rho_{AB})$  are valid measures of QCs in general. As we have discussed in Section 3.2.2, Requirement (i) is naturally obeyed, and the contractivity of  $D_1$  implies Requirements (ii) and (iv), however it is not known whether Requirement (v) holds. Regarding Requirement (iii), it was proven in [141] by showing that  $Q_A^{M_1}(|\psi\rangle_{AB}) = Q_{AB}^{M_1}(|\psi\rangle_{AB})$  and both amount to a measure of entanglement.

3.4.2. Bures and Hellinger distance Two measurement induced geometric measures of QCs that do not have such a close link with the corresponding geometric measures are the ones induced by the squared Bures distance and the squared Hellinger distance (see Table 2). The one-sided versions, considered in [56], are valid measures of QCs obeying all the Requirements (i)–(iv), while Requirement (v) has not been tested yet. In particular, for pure states  $|\psi\rangle_{AB}$  it holds

$$Q_A^{M_{\text{Bu}}}(|\psi\rangle_{AB}) = Q_A^{G_{\text{He}}}(|\psi\rangle_{AB}) = 2\left(1 - \sqrt{\text{Tr}(\rho_A^2)}\right) = 2\left(1 - \sqrt{\text{Tr}(\rho_B^2)}\right), \quad (81)$$

$$Q_A^{M_{\rm He}}(|\psi\rangle_{AB}) = 2\left(1 - \text{Tr}(\rho_A^{3/2})\right) = 2\left(1 - \text{Tr}(\rho_B^{3/2})\right),\tag{82}$$

which are measures of entanglement [62]. We point again to [56] for a compendium of informative bounds.

Finally, the extensions to other distances suggested in Section 3.3.7 can likewise be considered for the measurement induced geometric measures.

#### 3.5. Measurement induced informational measures

This category encompasses some of the seminal and most studied measures of QCs.

3.5.1. von Neumann entropy based measures: Quantum deficit A simple choice to gauge the change of information due to local measurements is the (negative) von Neumann entropy in Table 3. By using  $I_S = -S$  in Eqs. (53) one can see that  $Q_A^{I_S}(\rho_{AB})$  and  $Q_{AB}^{I_S}(\rho_{AB})$  quantify the global information lost by (rank one) LPMs.

the global information lost by (rank one) LPMs. The measures  $Q_A^{I_S}(\rho_{AB})$  and  $Q_{AB}^{I_S}(\rho_{AB})$  were introduced in [87], being called the "oneway quantum deficit" and "zero-way quantum deficit", respectively. They have fundamental operational interpretations in thermodynamics, as we discuss in Section 4.3.1. The coincidence between these measures and the relative entropy based geometric measures, Eq. (65), was highlighted in [87], and the coincidence with the distillable entanglement
based entanglement activation measures, Eq. (66), was highlighted in [36, 35]. The one-sided measure  $Q_A^{I_S}(\rho_{AB})$  was also independently studied [94] and referred to as "thermal discord" in [12]. All the Requirements are satisfied for these measures as discussed in Section 3.3.1.

The corresponding LGM based informational measures in terms of von Neumann entropy,  $Q_A^{\tilde{I}_S}(\rho_{AB})$  and  $Q_{AB}^{\tilde{I}_S}(\rho_{AB})$ , were considered in [142, 10], with optimisation restricted to local rank one POVMs. These measures are less used in the literature and Requirements (iv) and (v) were not tested, while Requirement (iii) was proven in [142].

A final word of clarification is needed regarding the restriction to rank one measurements for von Neumann entropy based informational measures of QCs. Notice that, if one allowed instead the optimisation over all LGMs (or even all LPMs), then the optimal measurement could be the trivial 'measurement' of not doing anything (i.e., the identity), resulting in illdefined quantifiers. The restriction to rank one measurements is therefore necessary to arrive at informative (and non-negative) QCs measures, and is anyway an intuitive restriction since rank one measurements represent the most fine grained type of measurements [143].

3.5.2. Mutual information based measures: Quantum discord Since we are interested in quantum *correlations*, it seems appropriate to consider as an informational gauge the mutual information  $I_{I} = I$ , which is itself a measure of correlations. Specifically, resulting from a generalisation of the classical mutual information to the quantum setting, the mutual information in Table 3 can be seen as the canonical quantifier of the total correlations between subsystem A and B in a state  $\rho_{AB}$ . We can think of the total correlations remaining in the states  $\tilde{\Pi}_A[\rho_{AB}]$  or  $\tilde{\Pi}_{AB}[\rho_{AB}]$  after a (general) LGM on subsystem A or on both subsystems A and B as effectively classical, because these are the correlations that are extracted from  $\rho_{AB}$  by a local measurement and can be stored in a classical register. Hence, to find the corresponding QCs one should subtract these classical correlations from the original total correlations of  $\rho_{AB}$ . Since different local measurements may lead to the extraction of different amounts of classical correlations, and given that we aim to quantify only the genuinely quantum share of the total correlations, we need to minimise this difference (or equivalently maximise the extractable classical correlations) over all LGMs, hence justifying the form of the mutual information based measures of QCs in Eqs. (52) and Eqs. (53). These measures will therefore be understood as quantifying, in general, the minimum amount of correlations lost in the state  $\rho_{AB}$  by the act of a maximally informative local measurement on one or both subsystems.

It turns out that the one-sided mutual information based measure of QCs coincides with the original definition of "quantum discord", which was first conceived in [6, 7] by interpreting the disagreement (or discord, indeed) between two possible extensions of classically identical definitions of the mutual information to the quantum setting. In [6] Ollivier and Zurek considered  $Q_A^{I_I}(\rho_{AB})$ , i.e., minimisation over LPMs, while in [7] Henderson and Vedral considered  $Q_A^{I_I}(\rho_{AB})$ , i.e., minimisation over LGMs. Having clarified the fundamental meaning of quantum discord within the broader scope of the informational approach to QCs, for completeness we will now provide a summary of the original analysis behind it by focusing on the more general case of LGMs, although the same holds analogously for LPMs.

Consider two classical random variables A and B with a joint probability distribution  $p_{ab}$ and marginal probability distributions  $p_a = \sum_{b \in B} p_{ab}$  and  $p_b = \sum_{a \in A} p_{ab}$ . The correlations between A and B are given by the mutual information

$$I(A:B) := \mathcal{H}(A) + \mathcal{H}(B) - \mathcal{H}(AB), \tag{83}$$

with  $\mathcal{H}(A) = -\sum_{a \in A} p_a \log p_a$  and  $\mathcal{H}(B) = -\sum_{b \in B} p_b \log p_b$  the Shannon entropies of the individual random variables A and B, and  $\mathcal{H}(AB) = -\sum_{a \in A} \sum_{b \in B} p_{ab} \log p_{ab}$  the joint Shannon

entropy of A and B [144, 145]. An equivalent expression for the mutual information is given by

$$\mathcal{J}(A:B) := \mathcal{H}(B) - \mathcal{H}(B|A), \tag{84}$$

where

$$\mathcal{H}(B|A) := \sum_{a \in A} p_a \mathcal{H}(B|a) \tag{85}$$

is the conditional Shannon entropy, i.e. the average of  $\mathcal{H}(B|a) = -\sum_{b \in B} p_{b|a} \log p_{b|a}$  over  $a \in A$ , with  $p_{b|a} := p_{ab}/p_a$  the conditional probability distribution of *B* given result *a*. By substituting the definition of  $p_{b|a}$  into the above expression of  $\mathcal{H}(B|A)$ , we can easily see that

$$\mathcal{H}(B|A) = \mathcal{H}(AB) - \mathcal{H}(A), \tag{86}$$

establishing the equivalence of  $\mathcal{I}(A : B)$  and  $\mathcal{J}(A : B)$ .

By extending the above to the quantum setting, the joint probability distribution  $p_{ab}$  generalises to a state  $\rho_{AB}$  of the composite system and the marginal probability distributions  $p_a$  and  $p_b$  become states of the respective subsystems, i.e.  $\rho_A = \text{Tr}_B(\rho)$  and  $\rho_B = \text{Tr}_A(\rho)$ . The first definition I(A : B) of the mutual information then naturally extends to

$$I(\rho_{AB}) := \mathcal{S}(\rho_A) + \mathcal{S}(\rho_B) - \mathcal{S}(\rho_{AB}), \qquad (87)$$

with the von Neumann entropy S replacing the Shannon entropy. Extending the second definition  $\mathcal{J}(A : B)$  of the mutual information is less simple because it is not unambiguously clear how to deal with the conditional entropy  $\mathcal{H}(B|A)$ . The most direct way would be to emulate Eq. (86) with the conditional entropy

$$\mathcal{S}_{B|A}(\rho_{AB}) := \mathcal{S}(\rho_{AB}) - \mathcal{S}(\rho_{A}). \tag{88}$$

However, to truly capture the conditioning of "*B* given *A*" in the quantum setting, one must first extract information on *A* by performing local measurements. Hence, following Eq. (85), the conditional entropy can be interpreted as the average entropy of the post-measurement ensemble of states of subsystem *B* following an LGM on *A* with subnormalised maps  $\{(\tilde{\pi}_A)_a\}$ . Invoking the definitions in Section 2.3.1, one can then define the LGM based conditional von Neumann entropy as

$$\mathcal{S}_{B|A}^{\tilde{\pi}_{A}}(\rho_{AB}) := \sum_{a} p_{a} \mathcal{S}(\rho_{B|a}), \qquad (89)$$

which gives an LGM based quantum version of the classical mutual information as

$$\tilde{\mathcal{J}}_{A}^{\tilde{\pi}_{A}}(\rho_{AB}) := \mathcal{S}(\rho_{B}) - \mathcal{S}_{B|A}^{\tilde{\pi}_{A}}(\rho_{AB}).$$
(90)

As argued before,  $I(\rho_{AB})$  in Eq. (87) represents the total correlations, while the maximisation of the quantity in Eq. (90) over all LGMs,

$$\tilde{\mathcal{J}}_A(\rho_{AB}) := \sup_{\tilde{\pi}_A} \left[ \tilde{\mathcal{J}}_A^{\tilde{\pi}_A}(\rho_{AB}) \right], \tag{91}$$

can be regarded as a measure of one-sided classical correlations (with respect to subsystem *A*). Their difference yields the (LGM based) quantum discord with respect to subsystem *A*,

$$\tilde{\mathcal{D}}_A(\rho_{AB}) := \mathcal{I}(\rho_{AB}) - \tilde{\mathcal{J}}_A(\rho_{AB}) = \inf_{\tilde{\pi}_A} \left[ \mathcal{I}(\rho_{AB}) - \tilde{\mathcal{J}}_A^{\tilde{\pi}_A}(\rho_{AB}) \right],$$

which may be equivalently seen as the difference between two quantum versions of the classical conditional entropy

$$\tilde{\mathcal{D}}_{A}(\rho_{AB}) = \inf_{\tilde{\pi}_{A}} \left[ \mathcal{S}_{B|A}^{\tilde{\pi}_{A}}(\rho_{AB}) - \mathcal{S}_{B|A}(\rho_{AB}) \right].$$
(92)

To see that this quantity coincides with the informational QCs measure that we label  $Q_A^{I_I}(\rho_{AB})$ , as defined in Eqs. 52, we can write explicitly (see e.g. [33])

$$\widetilde{\mathcal{J}}_{A}^{\widetilde{\pi}_{A}}(\rho_{AB}) = \mathcal{S}(\rho_{B}) - \mathcal{S}_{B|A}^{\widetilde{\pi}_{A}}(\rho_{AB}) = \mathcal{S}(\rho_{B}) - \sum_{a} p_{a}\mathcal{S}(\rho_{B|a}) \\
= \mathcal{S}(\rho_{B}) - \sum_{a} p_{a} \log p_{a} - \mathcal{S}\left(\sum_{a} p_{a} |a\rangle \langle a|_{A'} \otimes \rho_{B|a}\right) \\
= \mathcal{I}\left(\sum_{a} p_{a} |a\rangle \langle a|_{A'} \otimes \rho_{B|a}\right) = \mathcal{I}\left(\widetilde{\Pi}_{A} [\rho_{AB}]\right),$$
(93)

where in the second line we use the joint entropy theorem [2], and in the last we use Eq. (16).

The two-sided informational measure based upon mutual information has also been extensively studied in the literature. In [97], a modification of  $Q_{AB}^{I_I}(\rho_{AB})$  was considered by omitting the optimisation over local measurements and called the "measurement induced disturbance". Local projections fixed in the respective eigenbases of the reduced states were used instead in its definition, i.e. if  $\rho_A = \sum_a p_a |a\rangle \langle a|_A$  and  $\rho_B = \sum_b p_b |b\rangle \langle b|_B$  then the chosen LPM would consist of projectors  $\{(\pi_A)_a \otimes (\pi_A)_b\}$  with  $(\pi_A)_a[\rho] = |a\rangle \langle a|_A \rho |a\rangle \langle a|_A$  and  $(\pi_B)_b[\rho] = |b\rangle \langle b|_B \rho |b\rangle \langle b|_B$ . The motivation behind this choice is so that the local projection leaves the reduced states of  $\rho_{AB}$  invariant, simplifying the measurement induced disturbance to the difference in global von Neumann entropies and thus leading to an upper bound to  $Q_{AB}^{I_S}(\rho_{AB})$ . However, ignoring the optimisation over local measurements generally causes an overestimation of the QCs and also gives rise to pathologies such as the measurement induced disturbance being undefined on states whose marginals have a degenerate spectrum [143], or even approaching a maximum on some classical-classical states where QCs vanish instead [146]. To overcome this problem, the measure  $Q_{AB}^{I_I}(\rho_{AB})$  including an optimisation over LPMs on A and B was studied in [146] as the "ameliorated measurement induced disturbance", while the more general LGM based measure  $Q_{AB}^{I_{I}}(\rho_{AB})$ , sometimes referred to as "symmetric quantum discord" had been considered already in [24, 143]. In particular, the authors of [147, 148, 149] defined the (two-sided) "classical mutual information"

$$\tilde{\mathcal{J}}_{AB}(\rho_{AB}) := \sup_{\tilde{\Pi}_{AB}} \mathcal{I}(\tilde{\Pi}_{AB}[\rho_{AB}]), \tag{94}$$

as the maximum classical correlations obtainable from a quantum state  $\rho_{AB}$  by purely local processing. The quantity in Eq. (94) amounts to the generalisation of Eq. (91) when LGMs  $\tilde{\Pi}_{AB} = \tilde{\Pi}_A \otimes \tilde{\Pi}_B$  on both A and B are considered as in Eq. (17). In this way, the two-sided mutual information based measure of QCs can then be written simply as

$$Q_{AB}^{I_{I}}(\rho_{AB}) = \mathcal{I}(\rho_{AB}) - \tilde{\mathcal{J}}_{AB}(\rho_{AB}), \qquad (95)$$

that is, as the difference between total and classical mutual information, in analogy to Eq. (92).

Non-negativity of  $Q_A^{I_I}(\rho_{AB})$  follows directly from the strong subadditivity of the von Neumann entropy [150] (see also [6, 151, 29]). This implies non-negativity of all the other mutual information based informational measures of QCs, see Eq. (98) in the following for clarification. As all informational measures, Requirements (i) and (ii) are satisfied. Furthermore, Requirement (iv) was proven to hold in [35, 111], and Requirement (iii) holds as well, since for pure states  $|\psi\rangle_{AB}$  we have [77, 33, 142]

$$Q_A^{I_I}(|\psi\rangle_{AB}) = Q_{AB}^{I_I}(|\psi\rangle_{AB}) = Q_A^{\tilde{I}_I}(|\psi\rangle_{AB}) = Q_{AB}^{\tilde{I}_I}(|\psi\rangle_{AB}) = \mathcal{S}(\rho_A) = \mathcal{S}(\rho_B), \tag{96}$$

which is the entanglement entropy. However, it is still an open question whether Requirement (v) is obeyed for this class of measures.

Regarding computability, the main obstacle in calculation of the mutual information based informational measures of QCs is the non-trivial optimisation over LGMs or LPMs. Indeed, the computational complexity of this optimisation is NP-complete [152], falling into the hardest class of problems within NP and with no efficient way currently known to provide a solution, so that the running time of any present algorithm grows exponentially with the dimension of the Hilbert space. Nevertheless, in the LGM case it is sufficient to optimise only over local POVMs that are extremal and of rank one (see Section 2.3.1), as proven in [151, 153, 154]. Notice that, unlike the case of the von Neumann  $I_S$  based informational measures in Section 3.5.1, the mutual information  $I_I$  based measures defined in this section can be optimised in principle over *all* LGMs or LPMs, without the need to restrict their definitions a priori, yet the optimal measurements always happen to consist of rank one operators [151, 142, 143].

Comparisons between minimisation over LGMs and LPMs have been carried out, in particular for the quantum discord, and even for two-qubit states there are many cases where  $Q_A^{I_I}(\rho_{AB}) < Q_A^{I_I}(\rho_{AB})$ , although the two quantities are typically quite close [155, 142, 156, 157]. Nevertheless, there has been some analytical progress in the calculation of these quantities, including: the evaluation of  $Q_A^{\tilde{I}_I}(\rho_{AB})$  in [158] for all two-qubit states of rank 2, and in [159, 156, 160, 161, 157, 98, 162] for certain two-qubit X states (note that the original result of [159] was claimed to apply to all two-qubit X states, but this claim was later shown to be incorrect in [156, 160, 163]), and in [98] for highly symmetric two-qudit states; and the evaluation of  $Q_A^{I_T}(\rho_{AB})$  in [97] for two-qubit Bell diagonal states (in which case  $Q_A^{I_T}(\rho_{AB}) = Q_{AB}^{I_T}(\rho_{AB})$ ), in [164, 165, 166, 167] for two-qubit X states, in [101] for qubit-qudit extended X states, in [168] for certain two-qutrit states, in [169, 170] for two-qubit reduced states of ground states of spin chains, and in [171, 172] by providing an efficient numerical algorithm for general two-qubit states. The quantum discord  $Q_A^{I_I}(\rho_{AB})$  has been generalised to continuous variable Gaussian states in [173, 120], and a closed computable formula obtained for all two-mode Gaussian states with optimisation restricted to Gaussian LGMs [120]; the latter measurements turn out to be in fact optimal (even among non-Gaussian measurements) for a large class of two-mode Gaussian states [174]. Two-sided measures of QCs based on the mutual information  $Q_{AB}^{I_I}(\rho_{AB})$  have also been studied for Gaussian states [175], showing that in this case Gaussian LGMs are generally not optimal.

We will detail some of the most concrete and direct operational interpretations of the quantum discord and other mutual information based informational measures of QCs in Section 4. We will also discuss the curious connection between  $Q_A^{I_I}(\rho_{AB})$  and the entanglement activation type measures in Section 3.6. Finally, let us mention an interesting and useful inequality involving the quantum discord  $Q_A^{I_I}$  and the entanglement of formation  $E^f$  (see Table 4), in arbitrary tripartite states  $\rho_{ABC}$  (with equality on pure states), given by [176, 177]

$$Q_A^{I_I}(\rho_{AB}) \ge \mathcal{S}(\rho_A) - \mathcal{S}(\rho_{AB}) + E_{B:C}^f(\rho_{BC}).$$
(97)

3.5.3. Hierarchy of informational measures and their regularisation The following wheel of inequalities summarises the hierarchical structure of the informational measures of QCs introduced so far, when all evaluated on the same arbitrary bipartite state  $\rho_{AB}$ :

The vertical inequalities arise because an optimisation over LGMs necessarily includes an optimisation over LPMs. The horizontal inequalities are shown in [143, 87]. Finally, the radial inequalities are explored in [142].

From an information theoretic perspective, it is often relevant to consider also an asymptotic scenario in which many independent and identically distributed (i.i.d.) copies of a quantum state are available. Given a real, non-negative function  $f(\rho)$  on the set of quantum states, the corresponding *regularisation* (if the limit exists) is defined as

$$f(\rho)_{\infty} := \lim_{N \to \infty} \frac{1}{N} f(\rho^{\otimes N}).$$
(99)

Regularisation has been used extensively in entanglement theory [3], in particular to define additive entanglement measures (like the entanglement cost) from non-additive ones (like the entanglement of formation), since for any  $f(\rho)$  the regularised  $f(\rho)_{\infty}$  is by construction additive on tensor product states.

One can then define accordingly regularised versions of all of the measures of QCs mentioned in this review. In particular, for the informational type of measures, a remarkable convergence occurs upon regularisation, collapsing the whole left half of the hierarchy in Eq. (98). Specifically, it was shown in [178] that the regularisations of LGM and LPM versions of both the mutual information and the von Neuman entropy based informational QCs measures all reduce to the same quantity, i.e.

$$Q_{A}^{\tilde{I}_{I}}(\rho_{AB})_{\infty} = Q_{A}^{I_{I}}(\rho_{AB})_{\infty} = Q_{A}^{\tilde{I}_{S}}(\rho_{AB})_{\infty} = Q_{A}^{I_{S}}(\rho_{AB})_{\infty}.$$
 (100)

This means in particular that the regularised one-way information deficit and the regularised one-sided quantum discord coincide in general. The quantity in Eq. (100) can equivalently be written as

$$Q_A^{\tilde{I}_I}(\rho_{AB})_{\infty} = I(\rho_{AB}) - \tilde{\mathcal{J}}_A(\rho_{AB})_{\infty}, \qquad (101)$$

which is obtained from Eq. (92) upon regularisation, taking into account that the mutual information is already additive,  $\mathcal{I}(\rho_{AB})_{\infty} = \mathcal{I}(\rho_{AB})$ . The complementary quantity  $\tilde{\mathcal{J}}_A(\rho_{AB})_{\infty}$  in Eq. (100), which is the regularised version of the one-sided measure of classical correlations [7] defined in Eq. (91), is known as "distillable common randomness" [179, 10].

*3.5.4. Other informational measures* There are natural extensions of the two informational measures suggested in Table 3. Generalisations of the von Neumann entropy are considered in [180], in particular the Rényi entropies

$$\mathcal{R}_{\alpha}(\rho) := -\frac{\log \operatorname{Tr}(\rho^{\alpha})}{\alpha - 1}$$
(102)

and the Tsallis entropies

$$\mathcal{T}_{\alpha}(\rho) := \frac{1 - \operatorname{Tr}(\rho^{\alpha})}{\alpha - 1},\tag{103}$$

with, as  $\alpha \to 1$ ,

$$\mathcal{R}_1(\rho) = \mathcal{T}_1(\rho) = \mathcal{S}(\rho). \tag{104}$$

These generalised entropies can give extensions of the von Neumann entropy based informational measures of QCs by choosing in Eqs. (53) the informational quantities  $I_{\mathcal{R}_{\alpha}} =$  $-\mathcal{R}_{\alpha}$  and  $I_{\mathcal{T}_{\alpha}} = -\mathcal{T}_{\alpha}$ . Indeed, in [180, 181] the authors consider  $Q_A^{I_{\mathcal{R}_{\alpha}}}(\rho_{AB})$ ,  $Q_{AB}^{I_{\mathcal{R}_{\alpha}}}(\rho_{AB})$ ,  $Q_A^{I_{\mathcal{R}_{\alpha}}}(\rho_{AB})$ , and even more general forms of entropy. These quantities are all non-negative because a complete rank one LPM always increases the Rényi and Tsallis entropies when the result is not known. Furthermore,  $I_{\mathcal{R}_{\alpha}}(\rho_{AB})$  and  $I_{\mathcal{T}_{\alpha}}(\rho_{AB})$  are invariant under local unitaries and hence Requirements (i) and (ii) hold. For pure states  $|\psi\rangle_{AB}$ , one has

$$Q_A^{I_{\mathcal{R}_a}}(|\psi\rangle_{AB}) = Q_{AB}^{I_{\mathcal{R}_a}}(|\psi\rangle_{AB}) = \mathcal{R}_a(\rho_A) = \mathcal{R}_a(\rho_B), \qquad (105)$$

$$Q_A^{l\tau_\alpha}(|\psi\rangle_{AB}) = Q_{AB}^{l\tau_\alpha}(|\psi\rangle_{AB}) = \mathcal{T}_\alpha(\rho_A) = \mathcal{T}_\alpha(\rho_B), \qquad (106)$$

which are both valid generalisations of the entanglement entropy [77], thus validating Requirement (iii). However, Requirement (iv) does not hold for the family of informational measures corresponding to the Tsallis entropies for any choice of  $\alpha \neq 1$  [182], because these measures can arbitrarily change with the addition and removal of an impure ancilla, as in Eq. (69). Indeed, when  $\alpha = 2$  the Tsallis entropies reduces to the linear entropy  $T_2(\rho) = 1 - \text{Tr}(\rho^2)$  and it holds that [183]

$$Q_A^{I_{\tau_2}}(\rho_{AB}) = Q_A^{M_2}(\rho_{AB}) = Q_A^{G_2}(\rho_{AB}),$$
(107)

which is the Hilbert-Schmidt based geometric measure of QCs. It was then suggested in [182] to rescale these measures by a type of generalised purity, analogous to the suggestion of [96], but it is still not known whether Requirement (iv) may be recovered in this way. The family of informational measures corresponding to the Rényi entropies are left unchanged by the addition and removal of an impure ancilla [184], but it is still unknown whether the more general Requirement (iv) holds. Finally, it is not known whether Requirement (v) is obeyed either.

One can also look at generalisations of the mutual information adopted as informational quantifier  $I_I$ . Recall that the mutual information  $\mathcal{I}(\rho_{AB})$ , defined in Eq. (87), can be equivalently rewritten as the distance from  $\rho_{AB}$  to the set of product states according to the relative entropy [89, 93], i.e.

$$I(\rho_{AB}) = \inf_{\sigma_{AB} \in \mathcal{P}_{AB}} D_{\text{RE}}(\rho_{AB}, \sigma_{AB}), \tag{108}$$

where  $\mathcal{P}_{AB}$  is the set of product states, and the minimisation is achieved in general by  $\sigma_{AB} = \rho_A \otimes \rho_B$ , the product of the marginals of  $\rho_{AB}$ . One could instead consider in Eq. (53) the informational quantities

$$I_{I_{\alpha}^{\mathcal{R}}}(\rho_{AB}) = \inf_{\substack{\chi_{AB} \in \mathcal{P} \\ \chi_{AB} \in \mathcal{P}}} D_{\mathcal{R}_{\alpha}}(\rho_{AB}, \chi_{AB}),$$

$$I_{I_{\alpha}^{\mathcal{T}}}(\rho_{AB}) = \inf_{\substack{\chi_{AB} \in \mathcal{P} \\ \chi_{AB} \in \mathcal{P}}} D_{\mathcal{T}_{\alpha}}(\rho_{AB}, \chi_{AB}),$$
(109)

arising from the sandwiched relative Rényi and Tsallis entropies defined in Eqs. (78) and (79), to define generalised mutual information based measures of QCs  $Q_A^{I_{I_{\alpha}}^{\mathcal{R}}}(\rho_{AB})$  and  $Q_A^{I_{I_{\alpha}}^{\mathcal{T}}}(\rho_{AB})$ with optimisation over LPMs, as was done in [185]. The corresponding two-sided LPM and one-sided and two-sided LGM measures may be defined as well. Being non-negative, zero on classical states and local unitarily invariant, all these quantities obey Requirements (i) and (ii). However, it is unknown whether Requirements (iii), (iv), and (v) hold, and one may need to check in each case for particular values of  $\alpha$  (beyond the standard  $\alpha \rightarrow 1$ ).

We finally highlight that similar approaches to defining measures of QCs by generalising entropies have been reported [186, 187, 188, 189, 190, 191].

#### 3.6. Entanglement activation measures

The entanglement activation approach to QCs, leading to the measures in Eqs. (56), was defined almost simultaneously in [36] and [35]. In particular, [36] formalised the problem as an adversarial game, named the "activation protocol", in which the goal is to establish entanglement between a given principal system and a register of ancillary systems against an adversary who could scramble the local bases on the principal system before the control-NOT interactions. The authors focused on fully symmetric measures of QCs (i.e., generalisations of the two-sided measures to a multipartite setting) as prerequisites for the necessary activation of bipartite entanglement between system and ancillae in the output state of the protocol. Their approach was later generalised to one-sided and generally partial quantumness measures in [28]. Independently, [35] studied an essentially equivalent protocol, focusing primarily on the one-sided case, yet providing the fundamental interpretation in terms of the von Neumann model of local measurements, as described in Section 2.3.2.

In the following, we cover a few particular cases of entanglement activation measures, corresponding to some relevant choices of  $E^{\zeta}$  as given in Table 4. The case of the entaglement of formation  $E^f$  is not treated explicitly here, but details are available in [12].

3.6.1. Distillable entanglement and relative entropy Both [36] and [35] reached the conclusion that the entanglement activation measures of QCs based upon relative entropy  $E^{\text{RE}}$  and distillable entanglement  $E^d$  (see Tables 2 and 4) coincide with each other and reduce to their geometric counterparts, see Eq. (66), and to the informational quantum deficits, as amply discussed in Section 3.3.1. This is a consequence of the fact that the pre-measurement states as defined in the entanglement activation approach belong to the special class of so-called maximally correlated states, for which  $E^d = E^{\text{RE}}$  [192]; more generally, this observation is crucial to establish Requirement (i) for all the QCs measures  $Q_A^{E\zeta}$  and  $Q_{AB}^{E_{\zeta}}$  even when using entanglement monotones  $E^{\zeta}$  that may vanish on some non-separable states, such as the distillable entanglement  $E^d$ .

It is interesting to note that the entanglement activation approach can also be adapted to recover the original measure of quantum discord  $Q_A^{I_I}(\rho_{AB})$  optimised over LPMs, as anticipated in Section 3.5.2 [35]. This requires a modification of the one-sided definition in Eqs. (56), to consider the minimisation of the *partial* entanglement in the pre-measurement state,

$$Q_A^{\Delta E^{\zeta}}(\rho_{AB}) := \inf_{\{|a\rangle_A\}} \left[ E_{AB:A'}^{\zeta} \left( \rho_{ABA'}^{\langle |a\rangle_A\}} \right) - E_{A:A'}^{\zeta} \left( \operatorname{Tr}_B(\rho_{ABA'}^{\langle |a\rangle_A\}}) \right) \right], \tag{110}$$

that is, the difference given by the entanglement created between the whole system AB and the ancilla A', minus the entanglement created between the probed subsystem A and the ancilla

A' in their reduced state, discarding the other subsystem B. It then turns out that [35]

$$Q_A^{\Delta E^{\zeta}}(\rho_{AB}) = Q_A^{I_I}(\rho_{AB}), \qquad (111)$$

with  $E^{\zeta}$  being either  $E^{\text{RE}}$  or  $E^d$ , thus providing an alternative interpretation to the quantum discord [6].

Note that one could define more general measures of QCs  $Q_{AB}^{\Delta E^{\zeta}}$  based on the partial entanglement activation approach of Eq. (110) by considering other choices for  $E^{\zeta}$ , however in this case one needs  $E^{\zeta}$  to be convex and non-increasing on average under stochastic LOCC (which are stronger requirements than standard LOCC monotonicity) [16], in order to generate valid measures of QCs, in particular to fulfill Requirement (iv) [35].

3.6.2. Negativity of quantumness The "negativity of quantumness" was defined in [36] and studied in detail in [80]. One-sided and two-sided versions of this measure,  $Q_A^{E^N}(\rho_{AB})$  and  $Q_{AB}^{E^N}(\rho_{AB})$ , are obtained by choosing the entanglement negativity  $E^N$  (see Table 4) [115, 116] in Eqs. (56). As already noticed in [36, 80], the negativity of quantumness amounts to the minimum coherence in all local bases quantified by the  $\ell_1$  norm (see Table 5). We can formalise this equivalence using the notation of Section 3.2.6. Namely, for a state  $\rho_{AB}$  and fixing local orthonormal reference bases { $|a\rangle_A$ } and { $|b\rangle_B$ }, let us define

$$C_{A}^{\ell_{1}\{|a\rangle_{A}\}}(\rho_{AB}) := \sum_{ij} \|_{A} \langle a_{i}| \rho_{AB} |a_{j}\rangle_{A} \|_{1} - 1,$$

$$C_{AB}^{\ell_{1}\{|a\rangle_{A},|b\rangle_{B}\}}(\rho_{AB}) := \|\rho_{AB}\|_{\ell_{1}}^{\{|a\rangle_{A} \otimes |b\rangle_{B}\}} - 1,$$
(112)

where we have explicitly indexed the local basis  $\{|a\rangle_A\} \equiv \{|a_i\rangle_A\}_{i=1}^{d_A}$  in the one-sided case, while the notation in the two-sided case indicates that the  $\ell_1$  norm is taken for the matrix representation of  $\rho_{AB}$  with respect to the product basis  $\{|a\rangle \otimes |b\rangle\}$  (recall that the  $\ell_1$  norm, or taxicab norm, is basis dependent). The minimisation of Eqs. (112) over all local reference bases as in Eq. (61) defines then the one-sided and two-sided  $\ell_1$  coherence based measures of QCs, which coincide with the corresponding versions of the negativity of quantumness [80],

$$Q_{A}^{C^{\ell_{1}}}(\rho_{AB}) := \inf_{\{|a\rangle_{A}\}} C_{A}^{\ell_{1}} {}^{\{|a\rangle_{A}\}}(\rho_{AB}) = Q_{A}^{E^{N}}(\rho_{AB}),$$

$$Q_{AB}^{C^{\ell_{1}}}(\rho_{AB}) := \inf_{\{|a\rangle_{A},|b\rangle_{B}\}} C_{AB}^{\ell_{1}} {}^{\{|a\rangle_{A},|b\rangle_{B}\}}(\rho_{AB}) = Q_{AB}^{E^{N}}(\rho_{AB}).$$
(113)

The two-sided negativity of quantumness  $Q_{AB}^{E^N}(\rho_{AB})$  can also be interpreted as a geometric measure as in Eq. (45) and as a measurement induced geometric measure as in Eq. (50),

$$Q_{AB}^{E^{N}}(\rho_{AB}) = Q_{AB}^{G_{\ell_{1}}}(\rho_{AB}) = \inf_{\substack{\chi_{AB} \in \mathcal{C}_{AB}}} D_{\ell_{1}}(\rho_{AB}, \chi_{AB}),$$

$$Q_{AB}^{E^{N}}(\rho_{AB}) = Q_{AB}^{M_{\ell_{1}}}(\rho_{AB}) = \inf_{\substack{\pi_{AB}}} D_{\ell_{1}}(\rho_{AB}, \pi_{AB}[\rho_{AB}]),$$
(114)

where we adopt the  $\ell_1$  distance

$$D_{\ell_1}(\rho, \sigma) := \|\rho - \sigma\|_{\ell_1}^{\{|i\rangle\}}, \tag{115}$$

evaluated with respect to the basis  $\{|a\rangle_A \otimes |b\rangle_B\}$ , which corresponds in the first relation to the eigenbasis of the classical-classical state  $\chi_{AB}$ , and in the second relation to the product of local bases associated with the complete rank one LPM as defined in Eq. (19).

In the special case of subsystem A being a qubit, the one-sided negativity of quantumness  $Q_{AB}^{E^N}(\rho_{AB})$  relates instead to the geometric, measurement induced, and unitary response measures based on the trace distance [80], as summarised in Eq. (72).

Closed formulae for the negativity of quantumness of two-qubit Bell diagonal states and two-qudit highly symmetric states were obtained in [80]. Two optical experiments recently investigated the entanglement activation framework [38, 193]. In particular, in [38] a procedure was devised and tested to verify the 'if and only if' in Eq. (27) based on a finite net of data, and the quantitative equivalence  $Q_A^{M_1}(\rho_{AB}) = Q_A^{G_1}(\rho_{AB})$  was demonstrated for two qubits. The experiment in [193] started instead with a classically correlated state, in which QCs were then created by local noise (using dissipative non-LCPO maps [64]), and finally activated into entanglement with ancillae through the protocol described in Section 2.3.2. Both experiments measured QCs via the negativity of quantumness. Finally, the negativity of quantumness has been studied for Gaussian states in [121], where it was also shown that non-Gaussian pre-measurement interactions are necessary for the entanglement activation of QCs in Gaussian states. Alternative schemes inspired by the activation protocol have been proposed for continuous variable systems [194, 195].

## 3.7. Unitary response measures

The unitary response measures were first introduced in [196, 81] to quantify entanglement. For pure states  $|\psi\rangle_{AB}$ , the entanglement of response was in fact defined as the distance from  $|\psi\rangle_{AB}$  to its image through a least disturbing local unitary with spectrum given by the  $d_A$ -th complex roots of unity, as in Eq. (60).

3.7.1. Hilbert-Schmidt distance The unitary response measure of QCs  $Q_A^{U_2}$  was then defined by resorting to the squared Hilbert-Schmidt distance in [39, 40] and was shown to be equivalent to the Hilbert-Schmidt geometric measure of QCs  $Q_A^{G_2}$  when subsystem A is a qubit, as recalled in Eq. (68). Consequently, due to the problem of Eq. (69) highlighted in [111], the response measure  $Q_A^{U_2}$  does not satisfy Requirement (iv) and does not qualify as a valid measure of QCs.

3.7.2. Trace distance The trace distance response measure of QCs  $Q_A^{U_1}$  was studied in [80] and connected both to the trace distance geometric measure of QCs and to the negativity of quantumness defined via the entanglement activation approach, when the subsystem A is a qubit, see Sections 3.3.3 and 3.6.2.

3.7.3. Bures and Hellinger distance: Discord of response In [41] the unitary response measures  $Q_A^{U_{\delta}}$ , therein referred to as "discord of response", were defined more generally as in Eq. (59) by resorting to any contractive distance  $D_{\delta}$ , with a particular focus on the squared Bures distance (see Table 2). Such a study was then complemented in [56] with a comprehensive comparison between several distance-based measures of QCs, including notable advances for the evaluation of the unitary response measures  $Q_A^{U_{Bu}}$  and  $Q_A^{U_{He}}$  based respectively on the squared Bures and the squared Hellinger distances (see Table 2). In particular, for pure states  $|\psi\rangle_{AB}$ , it holds [56]

$$Q_A^{U_{\mathrm{Bu}}}(|\psi\rangle_{AB}) = 2\left(1 - \sqrt{1 - E^{U_F}(|\psi\rangle_{AB})}\right),\tag{116}$$

$$Q_A^{U_{\rm He}}(|\psi\rangle_{AB}) = 2E^{U_F}(|\psi\rangle_{AB}), \qquad (117)$$

Measures and applications of quantum correlations

where

$$E^{U_F}(|\psi\rangle_{AB}) = \inf_{U_A} \left[ 1 - \left| \left\langle \psi \right| \left( U_A \otimes \mathbb{I}_B \right) \left| \psi \right\rangle_{AB} \right|^2 \right]$$
(118)

is the entanglement of response defined as in Eq. (60) with the infidelity (see Table 2) chosen as distance function and with optimisation restricted to root-of-unity local unitaries  $U_A$ . In the case of mixed states, closed formulae for  $Q_A^{U_{Bu}}$  and  $Q_A^{U_{He}}$  were first derived in [27, 41] for the special case of two-qubit Bell diagonal states. More generally, for arbitrary states  $\rho_{AB}$  when A is a qubit, these two measures can be linked analytically to their geometric counterparts by the simple relation [56]

$$Q_A^{U_\delta}(\rho_{AB}) = 4Q_A^{G_\delta}(\rho_{AB}) - \left[Q_A^{G_\delta}(\rho_{AB})\right]^2,$$
(119)

with  $\delta$  = Bu, He. In particular, by chaining this with Eqs. (75) and (67), we get the simple relation

$$Q_A^{U_{\rm He}}(\rho_{AB}) = 4Q_A^{G_2}(\sqrt{\rho_{AB}}) = 4Q_A^{M_2}(\sqrt{\rho_{AB}}), \qquad (120)$$

which shows that the unitary response measure of QCs based on the squared Hellinger distance is equivalent to the remedied Hilbert-Schmidt geometric measure defined in [113], hence is analytically computable for all states  $\rho_{AB}$  when A is a qubit. In this specific case, the Hellinger unitary response measure  $Q_A^{U_{\text{He}}}(\rho_{AB})$  is also equivalent to the coherence based measure defined in terms of the Wigner-Yanase skew information  $Q_A^{C^{WY}}(\rho_{AB})$ , introduced in [54] as "local quantum uncertainty" (see Section 3.8.4). Namely,

$$Q_A^{U_{\rm He}}(\rho_{AB}) = 2Q_A^{C^{\rm WY}}(\rho_{AB}),$$
 (121)

for all states  $\rho_{AB}$  when A is a qubit. In this case, such an observation independently provides a closed formula for  $Q_A^{U_{\text{He}}}(\rho_{AB})$  [54], equivalent to the one given in [113, 56].

The Bures and Hellinger unitary response measures have been studied for Gaussian states with the restriction to Gaussian unitaries in [197], within the operational context of quantum reading (see Section 4.2.2).

*3.7.4. Chernoff distance: Discriminating strength* Finally, in [58] the restriction to a harmonic spectrum was lifted by defining the "discriminating strength" as follows

$$Q_{A}^{U_{C}^{\Gamma}}(\rho_{AB}) := 1 - \max_{U_{A}^{\Gamma}} C(\rho_{AB}, U_{A}^{\Gamma}[\rho_{AB}]), \qquad (122)$$

where

$$C(\rho, \sigma) := \min_{0 \le s \le 1} \operatorname{Tr}(\rho^s \sigma^{1-s})$$
(123)

is a measure of indistinguishability between the states  $\rho$  and  $\sigma$ , which coincides with the Uhlmann fidelity *F* [198] (see Table 2),

$$F(\rho,\sigma) := \left[ \operatorname{Tr} \left( \sqrt{\sqrt{\sigma}\rho \sqrt{\sigma}} \right) \right]^2, \qquad (124)$$

when either  $\rho$  or  $\sigma$  is pure. The quantity in Eq. (123) enters in the definition of the quantum Chernoff bound [132]

$$\xi(\rho,\sigma) := -\log C(\rho,\sigma). \tag{125}$$

In Eq. (122), the optimisation is over all local unitaries  $U_A^{\Gamma}$  on subsystem A with nondegenerate spectrum  $\Gamma$ , as in Eq. (58). The discriminating strength, whose defining operational interpretation in the context of channel discrimination will be discussed in Section 4.2.2, can be thus regarded as a unitary response measure of QCs based on the Chernoff distance  $D_{\rm C}(\rho, \sigma) := 1 - {\rm C}(\rho, \sigma)$ . The choice of the spectrum  $\Gamma$  in order to maximise the measure  $Q_A^{U_{\rm C}^{\rm c}}$  was analysed as well in [58]. When subsystem A is a qubit (in which case any spectrum leads to an equivalent measure up to a normalisation factor, hence the harmonic spectrum can be assumed without any loss of generality [54]), it was further shown in [58] that the discriminating strength coincides up to a constant multiplicative factor with the local quantum uncertainty [54], and thus with the Hellinger unitary response measure of QCs, via Eq. (121). In continuous variable systems, the discriminating strength was investigated in [199] by restricting to a minimisation over Gaussian unitaries, and a closed expression was derived for a subclass of two-mode Gaussian states, exploiting the analytical formula for the quantum Chernoff distance between a pair of two-mode Gaussian states obtained in [200].

At present, the status of Requirement (v) for all the above mentioned unitary response measures of QCs remains unknown.

### 3.8. Coherence based measures

DE(IA)

3.8.1. Relative entropy As with many other resources in quantum information theory, the relative entropy can be used to define a coherence quantifier  $C^{\text{RE}}$  which is a full monotone in any possible resource theory of coherence [42, 43], and is easily computable for all quantum states as indicated in Table 5. Following the approach outlined in Section 3.2.6, we can define basis-dependent quantifiers of local coherence with respect to the relative entropy by adopting a geometric approach [95, 47], and simplify their expression by resorting to the equivalence with the measurement induced geometric approach [93]. We have then

$$C_{A}^{\text{RE} \{|a\rangle_{A}\}}(\rho_{AB}) := \inf_{\chi_{AB} \in \mathcal{I}_{A}^{[|a\rangle_{A}\}}} D_{\text{RE}}(\rho_{AB}, \chi_{AB}) = D_{\text{RE}}(\rho_{AB}, \pi_{A}[\rho_{AB}]),$$

$$C_{AB}^{\text{RE} \{|a\rangle_{A},|b\rangle_{B}\}}(\rho_{AB}) := \inf_{\chi_{AB} \in \mathcal{I}_{AB}^{[|a\rangle_{A},|b\rangle_{B}\}}} D_{\text{RE}}(\rho_{AB}, \chi_{AB}) = D_{\text{RE}}(\rho_{AB}, \pi_{AB}[\rho_{AB}]),$$
(126)

where, as usual, the LPMs  $\pi_A$  and  $\pi_{AB}$  project the subsystems in the chosen reference bases  $\{|a\rangle_A, |b\rangle_B\}$  as in Eqs. (18) and (19). The coherence based measures of QCs defined in terms of relative entropy are then obtained upon minimisation over said local bases as in Eqs. (61), and as anticipated return the relative entropy based geometric measures of QCs,

$$Q_A^{C^{\text{RE}}}(\rho_{AB}) = \inf_{\{|a\rangle_A\}} C_A^{\text{RE}}(|a\rangle_A)(\rho_{AB}) = Q_A^{G_{\text{RE}}}(\rho_{AB}),$$

$$Q_{AB}^{C^{\text{RE}}}(\rho_{AB}) = \inf_{\{|a\rangle_A, |b\rangle_B\}} C_{AB}^{\text{RE}}(|a\rangle_A, |b\rangle_B)(\rho_{AB}) = Q_{AB}^{G_{\text{RE}}}(\rho_{AB}),$$
(127)

discussed in Section 3.3.1 together with all their equivalent interpretations in terms of the other types of QCs quantifiers.

*3.8.2. Infidelity* Exploiting the results of [35, 49], one can also derive coherence based measures of QCs in terms of infidelity (see Table 5), which turn out to match their geometric counterparts, analogously to the case of relative entropy. Compactly, we have indeed

$$Q_{A}^{C^{F}}(\rho_{AB}) := 1 - \sup_{\{|a\rangle_{A}\}} \sup_{\chi_{AB} \in \mathcal{I}_{A}^{\{|a\rangle_{A}\}}} F(\rho_{AB}, \chi_{AB}) = Q_{A}^{G_{F}}(\rho_{AB}),$$

$$Q_{AB}^{C^{F}}(\rho_{AB}) := 1 - \sup_{\{|a\rangle_{A}, |b\rangle_{B}\}} \sup_{\chi_{AB} \in \mathcal{I}_{A}^{\{|a\rangle_{A}, |b\rangle_{B}\}}} D_{F}(\rho_{AB}, \chi_{AB}) = Q_{AB}^{G_{F}}(\rho_{AB}),$$
(128)

where the infidelity based geometric measures are monotonically related to the corresponding measures defined in terms of squared Bures distance (see Table 2) and enjoy accordingly all their properties, discussed in Section 3.3.4.

3.8.3.  $\ell_1$  norm The  $\ell_1$  norm defines a very intuitive and easy to compute quantifier of coherence (see Table 5), which is also a full monotone in all possible resource theories of quantum coherence [43]. A consistent pathway to obtain QCs measures from the  $\ell_1$  norm of coherence has been presented in Section 3.6.2, where the ensuing coherence based quantifiers have been shown to reproduce exactly the entanglement activation measures based on negativity. In particular, the two-sided measure  $Q_{AB}^{C\ell_1}(\rho_{AB})$ , alias the two-sided negativity of quantumness  $Q_{AB}^{E^N}(\rho_{AB})$  [80], exhibits a particularly appealing form, as it amounts to the total of the moduli of the off-diagonal elements of the density matrix  $\rho_{AB}$  (traditionally referred to as "coherences"), minimised over all local product bases. This is valid for arbitrarily large multipartite systems as well, with respect to the product of orthonormal bases on each subsystem [36].

3.8.4. Asymmetry: Local quantum uncertainty and interferometric power Beyond redefinition of existing measures, the coherence based approach can be used to define original measures of QCs of high operational significance, which do not follow straightforwardly from other approaches. This is the case in particular for two coherence quantifiers defined in the context of the resource theory of asymmetry, namely the Wigner-Yanase skew information and the quantum Fisher information (see Table 5). Both quantities are monotones under phase covariant operations [69], but not under more general incoherent operations [43], as proven in [83, 82] for the skew information, and in [70] for the quantum Fisher information. These two metrics belong to the family of monotone and Riemannian metrics discussed in Section 3.3.7 with geodesics corresponding respectively to the Hellinger and the Bures distances (see Table 2) [77]. In [54] and [57], valid measures of QCs have been successfully defined based on these two asymmetry monotones, as we now review within a unified approach.

Given a measure of coherence in the context of quantum asymmetry  $C^{\eta\Gamma}(\rho, K^{\Gamma})$ , defined with respect to a fixed reference basis  $\{|i\rangle\}$  identified as the eigenbasis of the observable  $K^{\Gamma} = \sum_{i} k_{i} |i\rangle\langle i|$  with non-degenerate spectrum  $\Gamma = \{k_{i}\}$ , then the corresponding one-sided measure of QCs of a bipartite state  $\rho_{AB}$  can be defined adapting Eq. (61) as

$$Q_A^{C^{\eta} \Gamma}(\rho_{AB}) := \inf_{K_A^{\Gamma}} C^{\eta \Gamma}(\rho_{AB}, K_A^{\Gamma} \otimes \mathbb{I}_B), \qquad (129)$$

where the minimisation is over all local observables  $K_A^{\Gamma}$  on subsystem A with non-degenerate spectrum  $\Gamma$ . Notice the resemblance with the case of unitary response measures in Eq. (58). Similarly to such a case, in fact, it is not immediate to extend Eq. (129) to define a corresponding two-sided measure of QCs, as including an optimisation over observables  $K_B^{\Gamma}$ on B as well might nullify the resulting two-sided quantifier even on maximally entangled states, like in the instance of Eq. (32).

Using Eq. (129), we can now define the "local quantum uncertainty"  $Q_A^{C^{WY}\Gamma}(\rho_{AB})$  as the coherence based measure of QCs induced by the Wigner-Yanase skew information [54], and the "interferometric power"  $Q_A^{C^{QF}\Gamma}(\rho_{AB})$  as the coherence based measure of QCs induced by the quantum Fisher information [57]. More precisely, each of these expressions defines a family of measures, dependent upon the fixed spectrum  $\Gamma$  of the observables entering the definition. The original names derive from their physical meaning. As the skew information  $C^{WY}(\rho, K)$  can be regarded as a quantifier of the genuinely quantum share of the uncertainty (out of the total variance) associated to the measurement of K on the state  $\rho$ , which is due to the fact that  $[\rho, K] \neq 0$  [201], the QCs measure  $Q_A^{C^{WY}}$  consequently quantifies the minimum quantum uncertainty affecting the measurement of a local observable on subsystem A in the state  $\rho_{AB}$ ; in this way, QCs are clearly linked to non-commutativity, as classical-quantum states are all and only the states which can commute with a local observable on A [54]. On the other hand, the interferometric power  $Q_A^{C^{QF}\Gamma}(\rho_{AB})$  recognises this non-commutativity as an advantageous feature to maintain a guaranteed precision in a metrological task, as we will detail in Section 4.2.1.

Beyond the specific interpretations, however, both measures enjoy similar properties, due to their common origin. In particular, they both obey all Requirements (i)–(iv) [54, 57], while Requirement (v) is currently being tested, with some partial progress achieved for  $Q_A^{C^{QF}\Gamma}(\rho_{AB})$  [67]. By construction, it holds

$$Q_A^{C^{WY}\Gamma}(\rho_{AB}) \le \frac{1}{4} Q_A^{C^{QF}\Gamma}(\rho_{AB}), \qquad (130)$$

with equality on pure states (for this reason, the  $\frac{1}{4}$  factor is usually included in the definition of the interferometric power for normalisation) [57, 202]. Closed analytical formulae are available for both measures on all bipartite states  $\rho_{AB}$  when subsystem A is a qubit [54, 57], demonstrating that these measures reconcile the typically contrasting requirements of computability and reliability [96]. In this case (dropping the spectrum superscript without loss of generality as discussed in Section 3.2.5), the local quantum uncertainty  $Q_A^{CWY}(\rho_{AB})$  is also equivalent to the unitary response measure based on the Hellinger distance  $Q_A^{GV}(\rho_{AB})$  [54] (which in turn reduces to the remedied Hilbert-Schmidt geometric measure  $Q_A^{G2}(\sqrt{\rho_{AB}})$  [113]), and to the discriminating strength  $Q_A^{U_C^r}(\rho_{AB})$  [58], as discussed in Section 3.7. Surprisingly, a similar equivalence does not hold between the interferometric power  $Q_A^{C^{OF}}(\rho_{AB})$  and the unitary response measure based on the Bures distance, and it is still unknown whether the former may admit any alternative interpretation in terms of a geometric or a unitary response type measure. The interferometric power has been extended to continuous variable systems, and a closed formula derived for all two-mode Gaussian states restricting to an optimisation over Gaussian observables K with harmonic spectrum [202].

Finally, the interferometric power was measured experimentally in a two-qubit nuclear magnetic resonance ensemble realised in chloroform, as part of an implementation of black box quantum phase estimation with noisy resources (See 4.2.1) [57].

3.8.5. Other coherence quantifiers Given that the formalism of coherence based measures of QCs has been introduced only in the present paper in its full generality, there have been no further records (to the best of our knowledge) to define other quantifiers of this type. However, the procedure outlined in Section 3.2.6 could be applied to any valid coherence measure in principle, with the analysis in Section 3.8.4 being specifically useful for monotones arising from the resource theory of asymmetry specialised to coherence. In this context, let us mention that the authors of [203] defined the "quantum variance", a related quantifier of quantum uncertainty yielding a computable and experimentally accessible lower bound to both Wigner-Yanase skew information and quantum Fisher information, and discussed its applications to measuring QCs and to exploring the associated quantum coherence length in many-body systems.

One suggestion for the future could be to define and study the measures of QCs based on the recently introduced "robustness of coherence" [45] (see Table 5), which may be expected to enjoy some appealing operational interpretations in a channel discrimination setting. This is left open for a keen reader (or co-author). Another possibility could be to study the measures of QCs induced by Tsallis generalisations of the relative entropy of coherence [204], and investigate their relationship with corresponding geometric and informational measures defined via the same entropies.

# 3.9. Recoverability measures

The recoverability approach to QCs, as outlined in Section 3.2.7, has been explored only quite recently; for this reason, there is only one measure which has been considered in explicit detail so far [50], as we now review.

3.9.1. Surprisal of measurement recoverability The "surprisal of measurement recoverability"  $Q_A^{R_F}$  has been originally defined by resorting to the (negative logarithm of) Uhlmann fidelity [50], corresponding the one-sided definition in Eqs. (62) with  $D_{\delta}(\rho, \sigma) = -\log F(\rho, \sigma)$ , where the fidelity  $F(\rho, \sigma)$  is given by Eq. (124). Explicitly,

$$Q_A^{R_F}(\rho_{AB}) := -\log \sup_{\Lambda_A^{\text{EB}}} F(\rho_{AB}, \Lambda_A^{\text{EB}}[\rho_{AB}]), \qquad (131)$$

where the optimisation is over the convex set of entanglement-breaking channels  $\Lambda_A^{\text{EB}}$  acting on subsystem *A* 

The motivation behind this definition arose from an alternative angle from which the quantum discord (see Section 3.5.2) can be looked at, introduced for the first time in [111]. Indeed, by using the fact that any LGM  $\tilde{\Pi}_A$  on subsystem A can be written as a unitary  $U_{A \to A'C}^{\tilde{\Pi}_A}$  from A to a composite system A'C followed by discarding C, i.e.,  $\tilde{\Pi}_A[\rho_{AB}] = \text{Tr}_C(U_{A \to A'C}^{\tilde{\Pi}_A}[\rho_{AB}])$ , one then gets

$$Q_A^{\tilde{I}_I}(\rho_{AB}) = \inf_{\tilde{\Pi}_A} \mathcal{I}_{A'}(U_{A \to A'C}^{\tilde{\Pi}_A}[\rho_{AB}]), \tag{132}$$

where we denote by  $I_A(\rho_{ABC})$  the conditional quantum mutual information of the tripartite state  $\rho_{ABC}$  with respect to subsystem A, defined as

$$I_A(\rho_{ABC}) := \mathcal{S}(\rho_{AB}) + \mathcal{S}(\rho_{AC}) - \mathcal{S}(\rho_A) - \mathcal{S}(\rho_{ABC}).$$
(133)

The latter quantity, often denoted as  $I(B; C|A)_{\rho}$  in information theory literature [205], captures the correlations present between *B* and *C* from the perspective of *A* in the i.i.d. resource limit, where an asymptotically large number of copies of the tripartite state  $\rho_{ABC}$  are shared between the three parties.

In the quest for developing a version of the conditional quantum mutual information which could be relevant for the finite resource regime, in [50] the surprisal of the fidelity of recovery of a tripartite quantum state  $\rho_{ABC}$  with respect to subsystem A was defined as follows,

$$F_A^R(\rho_{ABC}) := -\log \sup_{\Lambda_{A \to AC}^R} F(\rho_{ABC}, \Lambda_{A \to AC}^R[\rho_{ABC}]),$$
(134)

where the optimisation is over all recovery maps  $\Lambda_{A\to AC}^R$  from A to the composite system AC. The surprisal of measurement recoverability was then introduced in [50] in complete analogy with quantum discord by simply substituting the conditional quantum mutual information with the surprisal of the fidelity of recovery,

$$Q_{A}^{R_{F}}(\rho_{AB}) := \inf_{\tilde{\Pi}_{A}} F_{A'}^{R}(U_{A \to A'C}^{\tilde{\Pi}_{A}}[\rho_{AB}]),$$
(135)

and only afterwards proven to be equivalent to Eq. (131).

For bipartite pure states  $|\psi\rangle_{AB}$ , the surprisal of measurement recoverability is equal to an entanglement measure,

$$Q_A^{\kappa_F}(|\psi_{AB}\rangle) = -\log \operatorname{Tr}(\rho_A^2), \qquad (136)$$

where  $\rho_A$  is the marginal state of subsystem A, thus satisfying Requirement (iii). The validity of Requirement (v) is still unknown.

In general, it was shown in [50] using the results of [206] that the surprisal of measurement recoverability is a lower bound to the quantum discord,

$$Q_A^{R_F}(\rho_{AB}) \le Q_A^{\tilde{I}_I}(\rho_{AB}).$$
(137)

Furthermore, [207] developed a hierarchy of efficiently computable and faithful lower bounds to  $Q_A^{\tilde{I}_T}$  that converge exactly to  $Q_A^{R_F}$ , thus showing in particular that the surprisal of measurement recoverability can be bounded numerically by a semidefinite program, and admits an operational interpretation in the task of local broadcasting, as we detail in Section 4.1.1.

It will be interesting in the future to study two-sided versions  $Q_{AB}^{R_F}(\varphi_{AB})$  of the surprisal of measurement recoverability, as well as other recoverability measures corresponding to different choices of distance  $D_{\delta}$  in Eqs. (62), and explore their interplay with the other types of measures of QCs collected in this review.

Further related measures can be obtained by alternative generalisations of the conditional quantum mutual information, e.g. in terms of Rényi entropies [208].

## 3.10. Summary of measures

We close this Section by summarising in Table 6 the key characteristics of the majority of QCs measures discussed in this review, including a synopsis of Requirements (i)-(v) and interdependency relations between measures of different types. When more than two measures coincide, as in the case of the relative entropy based measures, due to space constraints we only note the equality with the first occurring measure aside each of the further ones; the chain of identities can then be reconstructed scanning the 'Notes' column vertically.

Note that similar (and more comprehensive) tables can be found for entanglement measures in [209].

Notation	Nomenclature	Notes	(i)	(ii)	(iii)	(iv)	(v)
$Q_A^{G_{ m RE}}$	relative entropy of discord		$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$
$Q_{AB}^{G_{\mathrm{RE}}}$	relative entropy of quantumness		$\checkmark$	$\checkmark$	$\checkmark$	_	$\checkmark$
$\dot{Q}_A^{G_2}$	Hilbert-Schmidt geometric discord		$\checkmark$	$\checkmark$	$\checkmark$	×	?
$Q_{AB}^{G_2}$	2-s Hilbert-Schmidt geometric discord		$\checkmark$	$\checkmark$	$\checkmark$	_	?
$Q_{\scriptscriptstyle A}^{G_1}$	trace distance geometric discord		$\checkmark$	$\checkmark$	?	$\checkmark$	$\checkmark$
$Q_{AB}^{G_1}$	2-s trace distance geometric discord		$\checkmark$	$\checkmark$	?	_	$\checkmark$
$Q_A^{G_{\rm Bu}}$	Bures geometric discord		$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$
$\tilde{Q}_{AB}^{\tilde{G}_{Bu}}$	2-s Bures geometric measure		$\checkmark$	$\checkmark$	$\checkmark$	_	$\checkmark$
$\tilde{O}_{A}^{G_{\text{He}}}$	Hellinger geometric discord		$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$
$\tilde{Q}_{AB}^{G_{\text{He}}}$	2-s Hellinger geometric discord		$\checkmark$	$\checkmark$	$\checkmark$	_	$\checkmark$
$\frac{\sim_{AD}}{O_A^{M_{\rm RE}}}$	1-s relative entropy m.i.g. measure	$= O_{A}^{G_{\text{RE}}}$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$
$\widetilde{O}^{A}_{AB}$	2-s relative entropy m.i.g. measure	$= \widetilde{O}_{AB}^{A}$	$\checkmark$	$\checkmark$	$\checkmark$	_	$\checkmark$
$\widetilde{O}_{AB}^{M_2}$	1-s Hilbert-Schmidt m.i.g. measure	$= O_{4}^{G_2}$	$\checkmark$	$\checkmark$	$\checkmark$	×	?
$\widetilde{O}_{AB}^{M_2}$	2-s Hilbert-Schmidt m.i.g. measure	$\boldsymbol{z}_A$	$\checkmark$	$\checkmark$	$\checkmark$	_	?
$O^{M_1}$	1-s trace distance mig measure	$d_{A=2} O_{G_{1}}^{G_{1}}$				1	?
$O^{M_1}$	2-s trace distance milding measure	$- \mathfrak{L}_A$			• ./	-	?
$\mathcal{Q}_{AB}^{AB}$	$1$ -s Bures m i $\sigma$ measure		· ./		·	1	?
$\mathcal{Q}_A^{M_{\mathrm{Bu}}}$	$2_{\rm s}$ Bures mig measure		./	./	./	×	• ?
$\mathcal{Q}_{AB}$ $\mathcal{Q}^{M_{\text{He}}}$	1 s Hellinger mig measure		1	1	1	/	• ?
$\mathcal{Q}_A$ $\mathcal{Q}^{M_{\mathrm{He}}}$	2 s Hellinger mig measure		~	~	× /	V	י ר
$\underline{\mathcal{Q}_{AB}}_{O^{I_S}}$	2-s Heiniger III.i.g. measure	$- O^{G_{\text{RE}}}$	<u> </u>	<u> </u>	<u> </u>		
$\mathcal{Q}_{A}^{I_{S}}$	one-way quantum deficit	$= Q_A$ $= Q_{GRE}^G$	~	~	~	$\checkmark$	~
$\mathcal{Q}_{AB}_{AB}$	zero-way quantum dencu	$= Q_{AB}$	V,	V	V	-	√ 0
$Q_A^{3}$	1-s von Neumann (LGM) informational measure		$\checkmark$	$\checkmark$	$\checkmark$	?	?
$Q^{\prime s}_{AB}$	2-s von Neumann (LGM) informational measure		$\checkmark$	$\checkmark$	$\checkmark$	_	?
$Q_A^{\prime_I}$	(LPM) quantum discord		$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	?
$Q_{AB}^{\prime_{I}}$	ameliorated measurement induced disturbance		$\checkmark$	$\checkmark$	$\checkmark$	_	?
$Q_A^{I_I}$	(LGM) quantum discord		$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	?
$Q_{AB}^{\tilde{I}_I}$	(LGM) symmetric quantum discord		$\checkmark$	$\checkmark$	$\checkmark$	_	?
$Q_{A}^{E^{G_{\text{RE}}}}$	1-s relative entropy activation measure	$= Q_A^{G_{\text{RE}}}$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$
$\tilde{O}_{AB}^{E^{G_{\mathrm{RE}}}}$	2-s relative entropy activation measure	$= O_{AB}^{G_{RE}}$	$\checkmark$	$\checkmark$	$\checkmark$	_	$\checkmark$
$\mathcal{Z}_{AB}^{AB}$	1-s distillable entanglement activation measure	$= O_{+}^{G_{\text{RE}}}$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$
$\widetilde{O}_{L}^{E^{d}}$	2-s distillable entanglement activation measure	$= O_{A}^{G_{RE}}$	$\checkmark$	$\checkmark$	1	_	$\checkmark$
$O^{E^N}_{E^N}$	1-s negativity of quantumness	$d_A=2 O^{G_1}$				1	?
$\mathcal{Q}_{A}^{E^{N}}$	2-s negativity of quantumness	$- \mathfrak{L}_A$	./	./	./	-	?
$\frac{\varphi_{AB}}{O^{U_2}}$	Hilbert-Schmidt discord of response	$d_A=2$ $AO^{G_2}$	./		•	~	$\frac{\cdot}{2}$
$\mathcal{Q}_A$ $\mathcal{O}^{U_1}$	Trace distance discord of response	$= 4Q_A$ $d_A=2 2 Q_{G_1}^{G_1}$	× /	× /	~		ว
$Q_A$	Durace distance discord of response	$= 2Q_A$	~	~	~	~	: 9
$Q_A^{\mu}$		$a = g(Q_A^{-1})$	V	V	V	V	<i>!</i>
$Q_{A_{U^{\Gamma}}}^{\circ}$	Hellinger discord of response	$a_A = g(Q_A^{\circ})$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	?
$Q_A^{o_c}$	discriminating strength	$\stackrel{d_A=2}{=} Q_A^{U_{\text{He}}}$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	?
$Q_{A_{-}}^{C^{G_{\mathrm{RE}}}}$	1-s relative entropy coherence measure	$= Q_A^{G_{\rm RE}}$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$
$Q^{C^{G_{ m RE}}}_{AB}$	2-s relative entropy coherence measure	$= Q_{AB}^{G_{\text{RE}}}$	$\checkmark$	$\checkmark$	$\checkmark$	_	$\checkmark$
$Q_A^{ar{C}^{\ell_1}}$	1-s $\ell_1$ norm coherence based measure	$=Q_A^{E^N}$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	?
$Q_{AB}^{\widetilde{C}^{\ell_1}}$	2-s $\ell_1$ norm coherence based measure	$=Q_{AB}^{\widetilde{E}^{N}}$	$\checkmark$	$\checkmark$	$\checkmark$	_	?
$Q_{\Lambda}^{C^{W_{Y}}}$	local quantum uncertainty	$\stackrel{d_A=2}{=} 2 Q_A^{U_{\text{He}}}$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	?
$Q^{C_{\rm QF}}$	interferometric power	A	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	?
$\overline{Q_{A}^{R_{F}}}$	Surprisal of measurement recoverability		$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	?
-A							

**Table 6.** Summary of measures of QCs, categorised by types as in Table 1. Additional legend: 'm.i.g.' stands for 'measurement induced geometric'; '1-s' means one-sided, '2-s' means two-sided; the superscript ' $d_A = 2$ ' indicates equality when A is a qubit; and  $g(x) = 4x - x^2$ .

#### 4. Applications and operational interpretations of quantum correlations

After having introduced so many different measures of QCs, even though guided by physical threads inspired by Fig. 3, one could honestly wonder how exactly to navigate through the maze of Table 6, or, in other words: *"Which measure should I use?"* 

It is desirable for any measure of QCs (or of any other *resource*) to be directly linked to the figure of merit in the performance of an operational task. This allows one to justify the quantitative ordering on the set of states imposed by a specific measure of QCs, i.e. if a given state has more QCs than another, we know that this is precisely because this state performs better in the corresponding operational task. In the following we provide a selective review of some of the most concrete and physically relevant applications and operational interpretations of measures of QCs.

### 4.1. Quantum information and communication

Much weight has been placed by the international community behind successfully performing quantum communication, which involves any information transmission task that cannot be achieved perfectly using classical resources, due to the fundamental role of communication primitives in the landscape of future quantum technologies, including quantum cryptography and computing [210]. QCs beyond entanglement have been shown to play a clear role in delineating possibilities and limitations for some well established protocols in quantum information and communication, including broadcasting, distribution of entanglement, quantum state merging and redistribution.

4.1.1. Local broadcasting Copying information is a natural ability in our classical realm. Indeed, the classical information you are reading right now is one of (hopefully) many copies. In the quantum realm, however, unconditional copying is not allowed, due to the no-cloning theorem [211, 212]. The no-cloning theorem says that it is impossible to create an identical copy of an arbitrary unknown pure quantum state of a system by using a composite unitary acting upon the system and an ancilla. The best that we can do is to reliably clone *orthogonal* pure states, which effectively corresponds to copying of classical information. Such an impossibility was subsequently generalised to mixed states and general transformations leading to the so called no-broadcasting theorem [213].

These no-go theorems enshrine classicality in single systems. For composite systems, the task of "local broadcasting" was outlined in [24]. Consider a composite bipartite state  $\rho_{AB}$  shared between laboratory A and B, with each laboratory also possessing an ancilla, respectively A' and B', in the composite state  $\sigma_{A'B'} = |0\rangle \langle 0|_{A'} \otimes |0\rangle \langle 0|_{B'}$ . The objective is for A and B to perform local operations only (and no communication) on the joint state  $\rho_{AB} \otimes \sigma_{A'B'}$ , in order to produce a joint state  $\rho_{AA'BB'} = (\Lambda_{AA'} \otimes \Lambda_{BB'})[\rho_{AB} \otimes \sigma_{A'B'}]$  which contains two (possibly correlated) copies of  $\rho_{AB}$ ,

$$\operatorname{Tr}_{A'B'}(\rho_{AA'BB'}) = \operatorname{Tr}_{AB}(\rho_{AA'BB'}) = \rho_{AB}.$$
(138)

It was shown in [24] that this task can only be performed if  $\rho_{AB} \in \mathcal{C}_{AB}$ , which is a very intuitive result saying that only classically correlated states can be locally broadcast. More generally, it was shown that, even if the task was relaxed to that of obtaining two (possibly different) reduced states but with the same total correlations as measured by the mutual information between *A* and *B* (or *A'* and *B'*), then the correlations themselves could only be broadcast if  $\rho_{AB}$  is classically correlated,

$$\rho_{AB} \in \mathcal{C}_{AB} \iff \exists \rho_{AA'BB'} \text{ such that } I(\operatorname{Tr}_{A'B'}(\rho_{AA'BB'})) = I(\operatorname{Tr}_{AB}(\rho_{AA'BB'})) = I(\rho_{AB}),$$

which is called the no-local-broadcasting theorem [24]. One-sided local (or unilocal) broadcasting was then considered in [214]. Here, the idea is for only *A* to have access to an ancilla  $|0\rangle \langle 0|_{A'}$  and to perform local operations to obtain states  $\rho_{AA'B} = (\Lambda_{AA'} \otimes \mathbb{I}_B)[\rho_{AB} \otimes |0\rangle \langle 0|_{A'}]$  such that

$$\mathcal{I}(\mathrm{Tr}_{A'}(\rho_{AA'B})) = \mathcal{I}(\mathrm{Tr}_{A}(\rho_{AA'B})) = \mathcal{I}(\rho_{AB}).$$
(139)

As intuitively expected, this is shown to be possible if and only if  $\rho_{AB} \in C_A$ . Partial broadcasting of QCs was further discussed in [215].

By design it appears straight away that QCs are relevant in the local broadcasting paradigm: indeed, the (im)possibility of local or unilocal broadcasting provides another qualitative characterisation of classical versus quantum states, that could be added to Fig. 3. Can such a paradigm provide natural operational interpretations to measures of QCs then? The answer is affirmative.

To begin with, precise quantitative links to one-sided informational measures of QCs and unilocal broadcasting were provided recently by [216], who consider a bipartite system in the state  $\rho_{AB}$  and the process of party A redistributing their correlations with B to N ancillae  $\{A_i\}_{i=1}^N$ by local operations  $\Lambda_{A \to A_1 \dots A_N}$ , with the aim to maintain as many correlations as possible between each  $A_i$  and the other party B. Let us define the reduced state of  $A_i$  and B after the redistribution as

$$\rho_{A_iB} = \operatorname{Tr}_{\otimes_{j\neq i}A_j}[(\Lambda_{A\to A_1\dots A_N} \otimes \mathbb{I}_B)[\rho_{AB}]].$$
(140)

Due the no-unilocal broadcasting theorem [24, 214], for any  $\rho_{AB} \notin C_A$  we know that some correlations will be generally lost in the redistribution process, i.e., each part  $A_i$  will have less correlations with *B* than the initial correlations between *A* and *B*, or in formulae,  $I(\rho_{A_iB}) \leq I(\rho_{AB}), \forall i = 1, ..., N$ . The loss of correlations when redistributing to the ancilla  $A_i$  can be averaged over all the ancillae, and minimised over all the local redistributing operations, to arrive at the minimum average loss of correlations,

$$\overline{\Delta}_{A}^{(N)}(\rho_{AB}) := \min_{\Lambda_{A \to A_{1} \dots A_{N}}} \frac{1}{N} \sum_{i=1}^{N} \left[ \mathcal{I}(\rho_{AB}) - \mathcal{I}(\rho_{A_{i}B}) \right].$$
(141)

Quite remarkably, in the limit of infinitely many ancillae, it was shown that [216]

$$\lim_{N \to \infty} \overline{\Delta}_A^{(N)}(\rho_{AB}) = Q_A^{\tilde{I}_I}(\rho_{AB}), \qquad (142)$$

that is, the asymptotic minimum average loss of correlations after attempting to unilocally broadcast any quantum state  $\rho_{AB}$  is given by its QCs content as measured by the quantum discord. Eq. (142) yields one of the most prominent operational interpretations of the quantum discord, and generalises the similar work of [217] whose result was limited to pure states only. Borrowing the title of [217], this means that *quantum discord cannot be shared*, while only the classical correlations quantified by Eq. (90) can be arbitrarily redistributed to many parties on one side.

These results have intimate connections to "quantum Darwinism" [218]. The idea behind quantum Darwinism is to explain the emergence of classical reality by observing a quantum system indirectly through its environment. The only information objectively accessible about the quantum system is that proliferated into many sub-parts of the environment. Thus, it is clear that  $\overline{\Delta}_A^{(N)}(\rho_{AB})$ , hence the quantum discord of  $\rho_{AB}$  in the limit of asymptotically many subparts of the environment, represents the portion of correlations inaccessible to the classical realm [216]. Curiously, no rigorous result analogous to Eq. (142) has been yet established in the twosided setting of redistributing correlations in a quantum state  $\rho_{AB}$  both on  $A \rightarrow \{A_i\}_{i=1}^N$  and on  $B \rightarrow \{B_j\}_{j=1}^N$ . It is reasonable to expect that in this case one would recover the two-sided informational measure  $Q_{AB}^{\tilde{I}_I}(\rho_{AB})$  as the asymptotic minimum average of the correlation gap  $I(\rho_{AB}) - I(\rho_{A_iB_i})$ . We conjecture this to be the case, following the intuition that  $Q_{AB}^{\tilde{I}_I}(\rho_{AB})$ should be related to the amount of correlations lost when performing local broadcasting [24], since the maximum share of correlations in a state  $\rho_{AB}$  that can be broadcast, i.e., transferred to classical registers from both subsystems, is given by the classical mutual information of  $\rho_{AB}$ , Eq. (94). We hope our conjecture will be settled in the near future.

Another way to obtain quantitative results is to look at how similar a locally broadcast copy can be to the original state  $\rho_{AB}$  [207]. Focusing again on the unilocal broadcasting setting of *A* attempting to locally copy  $\rho_{AB}$  to *N* ancillae  $\{A_i\}_{i=1}^N$  and arriving at the reduced states  $\rho_{A_iB}$  in Eq. (140), it is clear that for any  $A_i$ 

$$\max_{\Lambda_{A \to A_1 \dots A_N}} F(\rho_{AB}, \rho_{A_i B}) \le 1,$$
(143)

where *F* is the Uhlmann fidelity, Eq. (124). Equality holds if and only if  $\rho_{AB}$  can be locally broadcast, or equivalently if  $\rho_{AB} \in C_A$ : in this case, for  $\rho_{AB} = \sum_i p_i |i\rangle \langle i|_A \otimes \rho_B^{(i)}$ , a channel implementing the transformation  $|i\rangle_A \rightarrow |i\rangle_A^{\otimes N}$  would be suitable for broadcasting. This transformation is a type of *N*-Bose-symmetric channel  $\Lambda_{A\rightarrow A_1...A_N}^{B_N}$  mapping *A* onto the fully symmetric subspace of  $A_1...A_N$ , i.e. returning a density operator invariant under any permutation  $\varpi(A_1...A_N)$  of the ancillae [219]. It is then interesting to consider [207]

$$Q_A^{B_N}(\rho_{AB}) := -\log \sup_{\Lambda_{A \to A_1 \dots A_N}^{B_N}} F(\rho_{AB}, \rho_{A_1B})$$
(144)

as a figure of merit quantifying the quality of *N*-copy local broacasting (note that, due to the symmetry of  $\Lambda_{A\to A_1...A_N}^{B_N}$ , it is enough to just consider  $\rho_{A_1B}$ ). In the case  $N \to \infty$ , the composition of  $\Lambda_{A\to A_1...A_N}^{B_N}$  followed by the partial trace over all but ancilla  $A_1$  is equivalent to the action of an entanglement breaking channel, Eq. (39), or simply  $\rho_{A_1B}$  is the output of an entanglement breaking channel acting on  $\rho_{AB}$ . This means that  $Q_A^{B_\infty}(\rho_{AB}) = Q_A^{R_F}(\rho_{AB})$ and we recover the surprisal of measurement recoverability  $Q_A^{R_F}(\rho_{AB})$  from Eq. (131). Hence,  $Q_A^{R_F}(\rho_{AB})$  has an operational interpretation in terms of the minimum distinguishability between  $\rho_{AB}$  and its unilocally broadcast copies in the asymptotic setting. Furthermore, it has been suggested that each  $Q_A^{B_N}(\rho_{AB})$  is a valid measure of one-sided QCs in its own right [207], obeying at least Requirements (i), (ii) and (iv). These quantities are simple to calculate numerically as the solution of a semidefinite program, and display a natural hierarchy [207, 206]

$$Q_A^{\mathsf{B}_2}(\rho_{AB}) \le Q_A^{\mathsf{B}_N}(\rho_{AB}) \le Q_A^{R_F}(\rho_{AB}) \le Q_A^{\tilde{I}_I}(\rho_{AB}),\tag{145}$$

hence providing computable lower bounds to quantum discord.

In summary, the ability to locally broadcast a correlated quantum state is fundamentally (and inversely) linked to the QCs present in the state, in both qualitative and various quantitative senses.

4.1.2. *Entanglement distribution* The entanglement activation framework presented in Section 3.2.4 provides already an operational setting to interpret measures of QCs in a state

 $\rho_{AB}$  in terms of their potential for the creation of entanglement with ancillary systems. Under some conditions, this entanglement can be swapped back to the original system *AB* [220]. More generally, this means that all the measures  $Q_A^{E\zeta}$  and  $Q_{AB}^{E\zeta}$  of QCs of the entanglement activation type, some of which are listed in Table 6, may be regarded as having an intrinsically operational character [36].

Here we discuss the role of QCs for a different task, namely "entanglement distribution". Distributing entanglement between spatially separated laboratories is a key starting point in many quantum information protocols. Consider two laboratories A and B, who want to increase their shared entanglement. From the fundamental paradigm of entanglement, it is clear that such a task cannot be achieved using only LOCC. Instead, it is necessary to send a quantum particle C from one laboratory to the other (we will assume the particle is sent from A to B), i.e. to utilise *quantum* communication in combination with local operations.

More rigorously, we can look at the state  $\rho_{ABC}$  of the composite system consisting of the two parties *A* and *B* and the carrier particle *C*. It is then relevant to consider two bipartitions of the three parties. Bipartition AC : B represents the carrier particle at laboratory *A*, while bipartition A : BC represents the carrier particle after having been sent to laboratory *B*. We then turn to the entanglement  $E_{AC:B}(\rho_{ABC})$  and  $E_{A:BC}(\rho_{ABC})$ , which represent the entanglement before and after the carrier particle has been sent (through a noiseless quantum channel), where  $E_{X:Y}$  is a measure of bipartite entanglement for the bipartition X : Y. The objective is to achieve  $E_{A:BC}(\rho_{ABC}) > E_{AC:B}(\rho_{ABC})$ . Laboratory *B* can then attempt to perform joint operations on both their system and the carrier particle to localise the entanglement onto *B* and hence increase the entanglement between *A* and *B*.

The question is whether we can place a minimum cost on the increase in entanglement  $E_{A:BC}(\rho_{ABC}) - E_{AC:B}(\rho_{ABC})$ . One might guess that the cost is related to  $E_{AB:C}(\rho_{ABC})$ , i.e. so that the increase in entanglement is limited by the entanglement carried by particle *C* with the two laboratories. In fact, entanglement across the *AB* : *C* split turns out not to be necessary, and the task of entanglement distribution can also be accomplished with a separable carrier, as first shown in [221]. Instead, [222, 223] suggest that more general QCs between *AB* and *C*, defined with respect to one-sided measurements on *C*, may represent appropriately the minimum cost. They showed that, when resorting to the geometric measures of QCs (see Section 3.2.1), it holds

$$Q_C^{G_\delta}(\rho_{ABC}) \ge E_{A:BC}^{G_\delta}(\rho_{ABC}) - E_{AC:B}^{G_\delta}(\rho_{ABC}), \tag{146}$$

where  $E_{X:Y}^{G_{\delta}}(\rho_{ABC})$  is the corresponding geometric measure of entanglement defined in Eq. (47) for the bipartition X : Y. This inequality, which in the case of measures based on the relative entropy represents a strengthening of the standard subadditivity inequality of von Neumann entropy, tells us that QCs on the *C* side are in general necessary for entanglement distribution. In fact, whenever QCs are not present, the transmission of the carrier particle *C* effectively reduces to classical communication. Examples of states  $\rho_{ABC}$  were given in [222] for when this bound is tight, and it was argued in [223] that entanglement distribution with separable states may be more efficient than using entangled carriers if local generation of entanglement at the sender laboratory is expensive.

However, the usefulness of a state for entanglement distribution is not determined only by its QCs. In [224] examples of states with nonzero QCs yet that cannot be used for entanglement distribution were provided, and additional conditions to ensure usefulness for this task were derived in terms of the dimension of *C* and the number of product terms in the decomposition of the separable  $\rho_{ABC}$  along the partition AB : C. This situation was likened to entanglement, whereby only some entangled states are useful for distillation (i.e. the states being not bound entangled) [3]. It was also suggested in [225] that the quantity  $Q_C^{G_\delta}(\rho_{ABC})$  may not a relevant resource for entanglement distribution because laboratory *A* initially holds the carrier *C* and can thus alter the QCs by local operations, potentially making Eq. (146) a rather loose bound.

Finally, a number of experiments have successfully demonstrated entanglement distribution with separable states in discrete and continuous variable setups [226, 227, 228].

4.1.3. Quantum state merging Imagine that our two laboratories A and B each sample from discrete random variables, with a joint probability distribution  $p_{ab}$  where  $a \in A$  and  $b \in B$  are the possible values found at each laboratory, see Section 3.5.2 for further details. With N repeated samples, they arrive at a sequence of values  $\{(a_1, b_1), (a_2, b_2), \ldots, (a_N, b_N)\}$ . However, neither laboratory knows the results of the other. If laboratory B wants to transmit their results to A, how much information must they send? Shannon's noiseless coding theorem tells us that for large N, B can send as little as  $\mathcal{H}(B)$  bits per sample for A to successfully reconstruct the complete sequence [144, 2]. Such a scheme, however, does not exploit any correlations between the two random variables. If A has some prior information on B, i.e. if I(A : B) > 0, then can B exploit this to send fewer than  $\mathcal{H}(B)$  bits per sample? The answer is yes, only the partial information that A is missing about B needs to be transmitted, which can be achieved with B sending only  $\mathcal{H}(B|A) = \mathcal{H}(AB) - \mathcal{H}(A) (\leq \mathcal{H}(B))$  bits of information per sample, provided N is large [229]. Importantly, this may be completed without B having any knowledge of the prior information that laboratory A has.

An analogous problem may be studied in the quantum setting. Now consider a quantum source that emits a bipartite pure state  $|\psi_i\rangle_{AB}$  with probability  $p_i$  and distributes this state to laboratories A and B. The laboratories know the statistics of the source and hence the mixed state  $\rho_{AB} = \sum_i p_i |\psi_i\rangle \langle \psi_i|_{AB}$ , but not the ensemble  $\{p_i, |\psi_i\rangle_{AB}\}$  realising it. Suppose that the laboratories sample from this source N times and have the sequence of (unknown) pure states  $\{|\psi_{i_1}\rangle_{AB}, |\psi_{i_2}\rangle_{AB}, \dots, |\psi_{i_N}\rangle_{AB}\}$ . For each sample of the source, how can B transmit their share of the distributed pure state to A? Both classical and quantum communication (and local operations) are at B's disposal. However, classical communication is a much simpler task than quantum communication, which requires sending of sensitive quantum information down noisy channels. Hence, we allow for unlimited and free classical communication and concern ourselves with finding the most efficient way in terms of quantum communication for B to transmit their share of the unknown state to A. With free classical communication we are able to carry out quantum communication via teleportation [230], using a bank of maximally entangled Bell states  $|\Phi\rangle_{AB} = \frac{1}{\sqrt{2}} (|0\rangle_A |0\rangle_B + |1\rangle_A |1\rangle_B$  shared between A and B. The question then becomes: what is the minimum entanglement, quantified by shared Bell states (or "ebits"), required per copy of  $\rho_{AB}$  for B to transmit their state to A?

This question was tackled in [231, 232]. Here it was shown that for any ensemble  $\{p_i, |\psi_i\rangle_{AB}\}$  realising the mixed state  $\rho_{AB}$  and for  $N \to \infty$ , the rate of ebit consumption is at least max $\{0, S_{B|A}(\rho_{AB})\}$ , with  $S_{B|A}(\rho_{AB}) = S(\rho_{AB}) - S(\rho_A)$  the conditional entropy of Eq. (88) (not to be confused with the measurement based quantity in Eq. (89)), representing the partial information needed for *B* to transmit their state to *A*. However, it is a curious fact that the quantum conditional entropy can be negative for some  $\rho_{AB}$  (consider, for example, any pure entangled state). In that case, *B* may transmit their state to *A* using only local operations and classical communication. Even better, it is then possible to *create* shared ebits at a rate no more than  $-S_{B|A}(\rho_{AB})$ . Thus, the negative quantum conditional entropy can be understood as the potential for future quantum communication between *A* and *B* in the form of teleportation.

The optimal protocol allowing B to transmit their state to A is called "quantum state

merging", and can be equivalently thought of in the following way [232]. Imagine that a third reference laboratory *C* is present so that the composite state of the three systems  $|\psi\rangle_{ABC}$  is pure. State merging is the task of performing LOCC and optimally drawing from (or contributing to) the bank of shared ebits to transform the state  $|\psi\rangle_{ABC}^{\otimes N}$  into the state  $|\psi\rangle_{ADC}^{\otimes N}$ , where *D* is an ancilla in laboratory *A* designed to hold the state of *B*. The fidelity between  $|\psi\rangle_{ABC}^{\otimes N}$  and  $|\psi\rangle_{ADC}^{\otimes N}$  must be high, tending to unity as  $N \to \infty$ , which means that the local state of the reference laboratory *C* is effectively unchanged throughout the process. Working within this setting is useful in our following exposition on the role of QCs in state merging.

One link with QCs was provided in [233]. They noted the fact that entanglement between A and B is destroyed during the state merging process and so considered the total entanglement consumption  $E_{A:B}^{f}(\rho_{AB}) + S_{B|A}(\rho_{AB})$ , where  $E_{A:B}^{f}(\rho_{AB})$  is the entanglement of formation between A and B (see Table 4). Here,  $E_{A:B}^{f}(\rho_{AB})$  quantifies the entanglement present before merging (which is entirely consumed), and  $S_{B|A}(\rho_{AB})$  quantifies the entanglement gained or lost after merging. By manipulating Eq. (97) [176, 177], given that the global state of the tripartite system ABC is pure, one finds that this total entanglement consumption is equal to the mutual information based measure of QCs, i.e. the quantum discord, of  $\rho_{BC} = \text{Tr}_A(|\psi\rangle \langle \psi|_{ABC})$  with one-sided measurements on C

$$E_{A:B}^{f}(\rho_{AB}) + \mathcal{S}_{B|A}(\rho_{AB}) = Q_{C}^{\bar{I}_{I}}(\rho_{BC}).$$
(147)

Hence, the QCs between B and C, with measurements on C, can be quantitatively understood as the total entanglement consumption in quantum state merging. The authors of [233] also considered the corresponding asymptotic total entanglement consumption, i.e.

$$E_{A:B}^{f}(\rho_{AB})_{\infty} + \mathcal{S}_{B|A}(\rho_{AB}) = Q_{C}^{I_{I}}(\rho_{BC})_{\infty}, \qquad (148)$$

where  $E_{A:B}^{f}(\rho_{AB})_{\infty}$  is the entanglement cost [3], thus providing an interpretation to the regularised measure of QCs given in Eq. (101).

These findings give an operational motivation for the asymmetry present in the one-sided measures of QCs, that is, in general  $Q_C^{\bar{I}_I}(\rho_{BC}) \neq Q_B^{\bar{I}_I}(\rho_{BC})$  because the total entanglement consumed for *B* to merge with *A* is not necessarily the same as for *C* to merge with *A*. A link to dense coding was also provided in [233]. In dense coding, one can use quantum communication to transmit classical information at a faster rate than with classical communication, and it was shown that  $Q_C^{\bar{I}_I}(\rho_{AC}) - Q_C^{\bar{I}_I}(\rho_{BC})$ , with  $\rho_{AC} = \text{Tr}_B(|\psi\rangle \langle \psi|_{ABC})$ , quantifies exactly the difference in the quantum advantage of dense coding for *C* to communicate with *A* as opposed to *B*.

An alternative operational view of QCs in terms of state merging was found in [234]. Here it was shown that the regularised QCs  $Q_A^{\tilde{I}_I}(\rho_{AB})_{\infty}$  of Eq. (101) represent the minimum increase in the cost of state merging (in terms of the rate of ebit consumption) if one first performs local measurements on each copy of A. Specifically, if  $S_{B|A}(\rho_{AB})$  is the cost of state merging before and  $S_{B|A}(\tilde{\Pi}_A[\rho_{AB}])_{\infty}$  is the cost of state merging after each LGM on A, then

$$\inf_{\tilde{\Pi}_{L}} \mathcal{S}_{B|A}(\tilde{\Pi}_{A}[\rho_{AB}])_{\infty} - \mathcal{S}_{B|A}(\rho_{AB}) = \mathcal{Q}_{A}^{I_{I}}(\rho_{AB})_{\infty}.$$
(149)

This operational link was then extended in [235, 236] by considering what is known as the fully quantum Slepian-Wolf protocol [237], which is the "mother of all protocols" since it is so general that it contains as special cases quantum state merging, dense coding, teleportation and entanglement distillation [2, 205]. Again, the regularised QCs  $Q_A^{\tilde{I}_I}(\rho_{AB})_{\infty}$  represent

the minimum drop in performance of the fully quantum Slepian-Wolf protocol after local measurements on A.

Finally, in [238] the primitive of "information concentration" was studied, a variation of state merging in which two parties A and B aim to maximise their mutual information by means of LOCC operations performed by B and a third party C. The figure of merit for this protocol was found to be a tripartite generalisation of the quantum discord.

4.1.4. *Quantum state redistribution* The quantum state redistribution protocol consists of a sender holding two systems A and C, a receiver holding instead only one system B, and a reference system R that is inaccessible to both sender and receiver, all sharing a fourpartite pure state  $|\psi_{ACBR}\rangle$  [239]. The objective is to redistribute the quantum information in such a way that is instead the receiver who holds C, while retaining the purity of the overall four-partite state. In order to accomplish this task, sender and receiver are only allowed to perform local quantum operations, consume or generate ebits, and the sender can give only qubits to the receiver. More precisely, contrarily to the quantum teleportation protocol [230], they are not allowed to classically communicate beyond what can be encoded in qubits. It turns out that this transfer can occur perfectly in the asymptotic limit of many copies provided that the amount of qubits that the sender gives to the receiver, also called communication cost, is at least half the conditional quantum mutual information  $\mathcal{I}_B(\rho_{CBR})$  of the reduced state  $\rho_{CBR} = \text{Tr}_A(|\psi_{ACBR}\rangle\langle\psi_{ACBR}|)$  between C and R with respect to B; see [239] for more details about the optimal protocol. Interestingly, this protocol is self-dual under time reversal, i.e., it can be reversed by generating the same amount of entanglement and spending the same communication cost. In particular, switching A and B has no effect on the communication cost, indeed for any pure state  $|\psi_{ACBR}\rangle$  it happens that  $\mathcal{I}_B(\rho_{CBR}) = \mathcal{I}_A(\rho_{ACR})$ , with  $\rho_{ACR} = \text{Tr}_B(|\psi_{ACBR}\rangle\langle\psi_{ACBR}|)$ . We can thus think of the quantity  $I_B(\rho_{CBR}) = I_A(\rho_{ACR})$ as twice the minimal communication cost to transfer C between A and B, regardless of the direction of the transfer, while retaining the purity of the global four-partite state  $|\psi_{ACBR}\rangle$ .

The quantum state redistribution protocol provides us with another operational interpretation for the quantum discord, as first highlighted in [240]. Let us consider two laboratories holding, respectively, the quantum systems *A* and *B*, which share the bipartite quantum state  $\rho_{AB}$ . Suppose that a maximally informative LGM  $\tilde{\Pi}_A$  is performed on *A*, with *A'* denoting the corresponding classical output. As already mentioned in Section 3.9.1, such LGM can be written as a unitary  $U_{A \to A'C}^{\tilde{\Pi}_A}$  from *A* to the composite system *A'C* followed by discarding *C*, i.e.,

$$\tilde{\Pi}_{A}[\rho_{AB}] = \operatorname{Tr}_{C}(U_{A \to A'C}^{\tilde{\Pi}_{A}}[\rho_{AB}]).$$
(150)

Moreover, we also know that the quantum discord  $Q_A^{\tilde{I}_I}(\rho_{AB})$  between A and B, Eq. (132), is exactly given by the conditional quantum mutual information  $I_{A'}(\rho_{A'CB})$  of the state  $\rho_{A'CB} = U_{A \to A'C}^{\tilde{\Pi}_A}[\rho_{AB}]$  between C and B with respect to A', i.e.,

$$Q_{A}^{I_{I}}(\rho_{AB}) = \mathcal{I}_{A'}(\rho_{A'CB}).$$
(151)

Now, let *R* be a reference system purifying the state  $\rho_{A'CB}$ , i.e., such that  $\rho_{A'CB} = \text{Tr}_R(|\psi_{A'CBR}\rangle\langle\psi_{A'CBR}|)$ . By looking at Eq. (150), we can then think of the measurement  $\Pi_A$  as a process whereby the environment *C* is lost and given to the reference *R*. One can then ask the following question. What is the optimal quantum communication cost needed to send system *C* from *R* to laboratory *A* in such a way that the action of the LGM  $\Pi_A$  can be undone? As already mentioned, this is given by half the conditional quantum mutual information  $I_{A'(CB)}$  between *C* and *B* with respect to *A'*, by assuming that laboratory *B* 

plays no role in the protocol. Therefore, Eq. (151) provides us with the promised operational interpretation of quantum discord as twice the optimal communication cost needed to restore the environment of a measurement so that it is no longer lost, i.e., to restore the coherence lost in a measurement [240]. Due to the symmetry under time reversal of such optimal protocol, the quantum discord between *A* and *B* can also be seen as the communication cost needed to send system *C* back to the reference *R*, i.e., as the amount of quantum information lost in the measurement process [240], which neatly captures the physical motivation behind the original definition of discord [6].

Building upon this operational interpretation, in [241] it has been shown that the quantum discord  $Q_A^{I_T}(\rho_{AB})$  of a bipartite quantum state  $\rho_{AB}$  is equal to the minimal rate of noise that one needs to apply to subsystem A in such a way that: (i) the resulting state  $\tilde{\rho}_{AB}$  is locally recoverable after an LGM acts on A and (ii) the post-measurement state corresponding to  $\tilde{\rho}_{AB}$  after such an LGM is indistinguishable from the post-measurement state corresponding to  $\rho_{AB}$  after a maximally informative LGM. A state satisfying properties (i) and (ii) is also said to be approximately einselected, where "einselection" is the abbreviation for environment-induced superselection and is a process whereby the interaction between a quantum system and the environment is such that only the eigenstates corresponding to particular observables of the system, so-called pointer states, persist in the system [37]. In other words, quantum discord is equal to the optimal cost of simulating einselection. This perhaps stands as a physical interpretation of quantum discord that is even more in line with its own original definition as a measure of the decrease of correlations after einselection is complete [6].

4.1.5. Remote state preparation In quantum teleportation [230, 242], the objective is to transmit an unknown quantum state from laboratory *A* to laboratory *B*. To achieve such a feat, one must make use of classical communication and some distributed entanglement. In fact, to teleport a qubit it is necessary and sufficient to use two bits of classical communication and one ebit [230]. Instead, in "remote state preparation", the objective is to transmit a *known* qubit state. It was shown by [243, 244] that in the many copy setting, this is achievable with the use of one classical bit of communication and one shared ebit per copy, and that in general it is possible to trade-off these two quantities: one can pay with more classical bits to save on ebits, and vice versa.

One special case is remote state preparation of the qubit states of the form  $|\psi\rangle = \frac{1}{\sqrt{2}}(|0\rangle + e^{i\phi}|1\rangle)$ , which lie on the equator of the Bloch sphere, as discussed in [245, 246]. In this case, remote state preparation can be performed without resorting to the many copy limit. Consider the shared ebit  $\frac{1}{\sqrt{2}}(|01\rangle - |10\rangle)$ , which can be equivalently written (up to a global phase) as  $\frac{1}{\sqrt{2}}(|\psi\psi^{\perp}\rangle - |\psi^{\perp}\psi\rangle)$ , where  $|\psi^{\perp}\rangle = \frac{1}{\sqrt{2}}(|0\rangle - e^{i\phi}|1\rangle)$  is the qubit state orthogonal to  $|\psi\rangle$ . If *A* performs an LPM on their half of the ebit in the basis  $\{|\psi\rangle, |\psi\rangle^{\perp}\}$ , then it is clear that *B* will either get the state  $|\psi^{\perp}\rangle$ , if the LPM by *A* resulted in  $|\psi\rangle$ , or the state  $|\psi\rangle$ , if *A* found  $|\psi^{\perp}\rangle$ . If *B* has  $|\psi^{\perp}\rangle$ , then  $|\psi\rangle = \sigma_3 |\psi^{\perp}\rangle$ , where  $\sigma_3$  is the third Pauli matrix. However, *B* can only know whether to apply the rotation from the measurement outcome of *A*, and hence remote state preparation of equatorial states in this single copy case requires one shared ebit and one bit of classical communication.

If one has access to an ebit as a resource in the above protocol, then remote state preparation can be achieved perfectly. Instead, if one uses a more general resource state, such as a mixed state of two qubits  $\rho_{AB}$ , then an error is introduced into the remote state preparation. Consider the state to be prepared with Bloch vector  $\vec{m}$  lying on the equatorial plane, and the resultant state after the aforementioned protocol with Bloch vector  $\vec{n}$ . In [103], a quadratic payoff function  $P_{\vec{m},\vec{n}}^2 := (\vec{m} \cdot \vec{n})^2$ , which captures how well the target  $\vec{m}$  and the actual  $\vec{n}$  overlap and is linked to the fidelity between the corresponding states, was adopted to quantify the performance of the protocol. Precisely, for a given resource state  $\rho_{AB}$ , a figure of merit for remote state preparation can be defined by first maximising  $P_{\vec{m},\vec{n}}$  over all output states  $\vec{n}$  (or equivalently all possible LPMs performed by A), then averaging the result over all pure states  $\vec{m}$  on the equator, and finally minimising the latter quantity over all possible choices of the reference north pole state on the Bloch sphere, since the orientation of the axes can be arbitrary, as long as this is pre-agreed between A and B. The resulting quantity, that we denote by  $\overline{P}_A^2(\rho_{AB})$ , can be interpreted as the remote state preparation fidelity achievable with the shared state  $\rho_{AB}$  in a worst case scenario. Interestingly, it was shown in [25] that, for a class of two-qubit states  $\rho_{AB}$  (i.e. the Bell diagonal states), this quantity coincides with (twice) the Hilbert-Schmidt based geometric measure of QCs,

$$\overline{P}_A^2(\rho_{AB}) = 2Q_A^{G_2}(\rho_{AB}), \tag{152}$$

thus apparently providing an operational interpretation to the latter, even though in a specialised setting. The theoretical result was supported experimentally using a photonic implementation, where certain separable states with nonzero QCs were shown to perform remote state preparation better than some entangled states with smaller QCs, according to the figure of merit given by Eq. (152).

A refinement of this protocol has been presented in [247], where it is argued, however, that the figure of merit  $\overline{P}_A^2(\rho_{AB})$  can be misleading, since it can be surpassed simply by employing the trivial protocol of B randomly preparing a pure equatorial state, regardless of the communication from A and the shared quantum resource  $\rho_{AB}$ . Instead, the suggested figure of merit should derive from the (non-quadratic) payoff-function  $P_{\vec{m},\vec{n}} = \vec{m} \cdot \vec{n}$ . Generalisations of the encoding and decoding strategies of A and B are also outlined in [247]. Instead of an LPM on the shared resource  $\rho_{AB}$ , A can perform a more general two-outcome LGM, and send the result using one bit of classical communication to B. B is then allowed to utilise any local operations to recover the best approximation of  $\vec{m}$ . In this more general setting, it is proved that it is impossible to use a shared separable state to outperform an entangled state in remote state preparation, hence providing a very critical analysis of the role of QCs beyond entanglement in this communication task. However, their role can be resurrected when certain reasonable restrictions are placed upon the local operations that B is able to perform when trying to recover  $\vec{m}$ . Two important cases are highlighted where it is possible for quantumly correlated separable states to outperform entangled ones: (i) B does not share a local Bloch reference frame with A, and (ii) B is restricted to unital operations. Even stronger, under condition (ii) and in the case of the shared resource state  $\rho_{AB}$  being Bell diagonal, the remote state preparation figure of merit recovers again a link to the Hilbert-Schmidt based geometric measure of QCs.

A further criticism of the role of QCs in remote state preparation was given by [248]. Here, they discriminate between the QCs that can be created by local operations and those that cannot, and it is argued that only the latter should be considered. In this sense, it is then shown that there are quantumly correlated states that are useless for remote state preparation, while there are states without QCs that can be made useful. Furthermore, it is discussed in [249] that an increase in QCs does not necessarily imply an increase in the quality of remote state preparation, even though in some cases a quantitative link is found between the fidelity of remote state preparation using noisy cluster states and the negativity of quantumness  $Q_A^{E^N}$ defined in Section 3.6.2.

Finally, in [102] a version of remote state preparation is discussed for a system consisting

of a qubit and a harmonic oscillator. It is found that the geometric measure  $Q_A^{G_2}(\rho_{AB})$ , where A is the qubit, can provide a lower bound to the payoff of remote state preparation under certain restrictions. These restrictions are on the receiving oscillator B: if B is restricted to only local unitary corrections following the measurement by A then  $Q_A^{G_2}(\rho_{AB})$  presents a limit on the performance of remote state preparation, whereas if B is free to perform arbitrary (non-unitary) local operations then  $Q_A^{G_2}(\rho_{AB})$  becomes no longer relevant, and the payoff reduces to an entanglement measure. This lends support to the findings of [247].

4.1.6. Quantum cryptography The idea behind "quantum key distribution" is to harness quantum mechanics to tackle the classically problematic task of distributing cryptographic keys securely, without a third party eavesdropper discovering what has been communicated [250]. In fact, the distributors and receivers of a key encoded in a *quantum* system can detect an attempt at eavesdropping (up to a sensitivity threshold) due to the disturbance induced by measurements on their system. From the outset, this problem seems tailored to utilising the resource of QCs and in particular the inherent non-orthogonality of bipartite states with nonzero QCs. Consequently, it was shown in [251] that quantumly correlated states are necessary for quantum key distribution in the device-dependent setting — which corresponds to the distributing and receiving parties being able to trust their devices.

In the practically relevant case of distribution of secure keys over a lossy channel with transmissivity  $\eta$ , such as a free-space link or an optical fibre, the secret key capacity *K* has been very recently determined and found to coincide with the maximum quantum discord that can be distributed to the remote parties through such a lossy channel [252]. In the following we briefly explain this result, which connects QCs to the ultimate limits of quantum communication in continuous variable systems. Further details are available in [252].

In general, a quantum cryptographic protocol [250] involves two distant laboratories, a sender A and a receiver B, separated by a quantum channel. The sender prepares ensembles of (non-orthogonal) input states and transmits them to the receiver, who can measure the outputs. They can resort to unlimited classical communication, which allows them to extract a key through error correction and privacy amplification. The secret key capacity K is defined as the maximum number of secret key bits per channel use which can be distributed over the quantum channel, obtained upon maximising over all possible input states of the sender A and all possible output measurements by the receiver B (in general, the input-output local operations may be adaptive, i.e., assisted by two-way classical communication [252]). An achievable protocol can be represented in terms of N ebits of entanglement distributed over the channel, followed by LOCC on both parties. Let us denote by  $\rho_{AB}^{\infty}$  the output state of A and B, in the limit  $N \rightarrow \infty$  (i.e., in the limit of an ideal maximally entangled distributed state, also known as an EPR state [17]). The result in [252] then shows that, in the important practical case of a bosonic lossy channel with transmissivity  $\eta$ , it holds

$$K(\eta) = Q_A^{I_I}(\rho_{AB}^{\infty}) = -\log_2(1-\eta),$$
(153)

where in particular Gaussian LGMs are found to be optimal to calculate the quantum discord  $Q_A^{\tilde{I}_I}(\rho_{AB}^{\infty})$  of the asymptotic two-mode Gaussian state  $\rho_{AB}^{\infty}$ , due to the results of [120, 174].

The further exploration of the role and power of QCs in quantum key distribution protocols certainly has potential [253, 251, 252], but its scope exceeds the present review, and we point the interested reader to [250, 254] for an overview on quantum cryptography.

4.1.7. Quantum locking of classical correlations Let us begin in the classical setting by considering a generic measure of correlations between two random variables A and B. We

might expect that this measure cannot increase by more than *m* bits if one were to carry out *m* bits of classical communication. Indeed, this intuitive property holds in particular for the mutual information I(A : B) of Eq. (83) [148]. How does this feature extends to the quantum setting, where quantum communication is also possible? Suppose we apply a scheme of local operations and either *n* qubits of quantum communication or 2*n* bits of classical communication, acting on a bipartite state  $\rho_{AB}$  shared between laboratory *A* and *B*. We may now require that a quantifier of correlations cannot increase by more than 2*n* bits after this scheme, since we can view each qubit of quantum communication as effectively two bits of classical communication due to dense coding. This property holds indeed for the quantum mutual information  $I(\rho_{AB})$  of Eq. (87) [147].

Interestingly, this property does not hold for the classical correlations  $\tilde{\mathcal{J}}_{AB}$  of Eq. (94), representing the correlations available to laboratories A and B through local measurements. Consider the bipartite (2N + 1)-qubit state  $\rho_{AB}$  with A having N qubits and B having N + 1,

$$\rho_{AB} = \frac{1}{2^{N+1}} \sum_{i=0}^{2^{N-1}} \sum_{j=0}^{1} (U_j | i \rangle \langle i | U_j^{\dagger})_A \otimes (| i \rangle \langle i | \otimes | j \rangle \langle j |)_B$$
(154)

where  $U_0 = \mathbb{I}$  and  $|\langle i'|U_1|i\rangle|^2 = 1/2^N$ , i.e. with  $\{U_1|i\rangle\}$  producing a mutually unbiased basis with respect to the basis  $\{|i\rangle\}$  [255]. For example, we may fix  $\{|i\rangle\}$  to be the computational basis and choose  $U_1 = H^{\otimes N}$ , with H the Hadamard gate. This state may be prepared by having laboratory B pick a random N-bit string with label i and sending to A either  $U_0 |i\rangle = |i\rangle$  or  $U_1 |i\rangle$ , again at random based on the value  $j \in \{0, 1\}$ . The total correlations are here  $I(\rho_{AB}) = N$ , but the classical correlations are  $\tilde{\mathcal{J}}_{AB}(\rho_{AB}) = N/2$ . Hence, some of the correlations are *locked*, i.e. inaccessible to A and B through local measurements. However, if B sends the result j to laboratory A, then A can reverse  $U_j$ . Laboratory A then measures in the computational basis so that they together share the N-bit string i, and therefore N + 1 bits of classical correlations (including the communicated bit). They have thus *unlocked* N/2 bits of classical correlations with only one bit of classical communication.

This curious phenomenon is a truly quantum effect, yet can occur in separable states like in Eq. (154). Clearly then, entanglement is not a figure of merit in locking of classical correlations, so what about more general QCs? The first hints that QCs play a role were given in [143, 256]. There it was noted that both the two-sided mutual information based measure of QCs  $Q_{AB}^{I_{I}}(\rho_{AB})$  and the measurement induced disturbance (see Section 3.5.2) are equal to N/2 in the above example, exactly the amount of locked classical correlations.

The link with QCs was placed on a firmer footing in [257]. In their setting, laboratory *B* samples from a random source of numbers *i* and wants to send the result to *A*, but has to use a noisy quantum channel so that *A* eventually receives  $\rho_A^{(i)}$ . Together *A* and *B* share a quantum-classical state  $\rho_{AB} \in C_B$ , just like the one in Eq. (154), and the objective is for *A* to infer the value of *i*. The classical correlations tell us how well *A* can do this. If laboratory *B* can then send a key to reveal the value of *i* to *A*, it turns out that the amount of classical correlations unlocked in doing so is exactly the quantum discord  $Q_A^{\tilde{I}_I}(\rho_{AB})$  [257]. Furthermore, in the asymptotic limit of many copies of  $\rho_{AB}$ , it was shown that the regularised QCs measure  $Q_A^{\tilde{I}_I}(\rho_{AB})_{\infty}$  of Eq. (101) quantifies the quantum advantage for *A* to successfully infer the message from *B* when compared to a corresponding classical protocol, in terms of the key length per copy required to unlock the correlations;  $Q_A^{\tilde{I}_I}(\rho_{AB})_{\infty}$  then also gives the amount of classical correlations unlocked per copy in the quantum protocol [257].

### 4.2. Quantum metrology and discrimination

Precision measurements are of key relevance in all quantitative sciences and underpin many technological applications, such as navigation, sensing, and medical imaging [258]. It is important therefore to identify the resources which can lead to a precision enhancement in suitable implementations. Multipartite entanglement has been extensively investigated as a resource for quantum enhanced measurements, see e.g. [259, 260, 261]. Here we review the role played by QCs in overcoming classical limitations for certain practical tasks of estimation and discrimination.

4.2.1. Black box quantum metrology Quantum metrology studies how to harness quantum mechanical effects to enhance the precision in estimating physical quantities not amenable to a direct observation [262, 263, 258]. A relevant class of problems in quantum metrology can be formalised in terms of phase estimation in an interferometric setup, which is akin to the scheme in Fig. 3(c). The estimation procedure then consists of the following steps. An input state  $\rho_{AB}$  enters a two-arm channel, in which the idler subsystem *B* is unaffected while the probe subsystem *A* undergoes a local unitary transformation  $U_A$ , so that the output density matrix can be written as  $\rho_{AB}^{\varphi} := (U_A \otimes \mathbb{I}_B)\rho_{AB}(U_A \otimes \mathbb{I}_B)^{\dagger}$ , with  $U_A = e^{-i\varphi H_A}$ , where  $\varphi$  is the parameter one aims to estimate and  $H_A$  is a (non-degenerate) Hamiltonian generator of the local unitary dynamics. Finally, the information on  $\varphi$  is recovered by constructing an unbiased estimator  $\hat{\varphi}$  obtained by classical processing of the data resulting from (possibly joint) measurements of suitable observables on the output state  $\rho_{AB}^{\varphi}$ .

For any input state  $\rho_{AB}$  and known generator  $H_A$ , and provided *n* i.i.d. iterations of the probing procedure are implemented, the maximum achievable precision is determined theoretically by the quantum Cramér-Rao bound [264], which says that the mean square error  $\Delta^2 \hat{\varphi}$  of any unbiased estimator scales as

$$n \,\Delta^2 \tilde{\varphi} \ge \frac{1}{C^{\text{QF}}(\rho_{AB}, H_A \otimes \mathbb{I}_B)},\tag{155}$$

where  $C^{QF}$  is the quantum Fisher information defined in Table 5, which quantifies how sensitive the output state is to an infinitesimal change of the encoded parameter  $\varphi \rightarrow \varphi + \delta \varphi$ . There always exists an optimal measurement strategy at the output stage which makes the inequality in Eq. (155) asymptotically tight for  $n \gg 1$ , which means that the quantum Fisher information  $C^{QF}(\rho_{AB}, H_A \otimes \mathbb{I}_B)$  can be regarded as the figure of merit determining the optimal precision achievable in the estimation of a parameter unitarily encoded by the local generator  $H_A$  when using an input state  $\rho_{AB}$ . Therefore, under the assumption of complete prior knowledge of  $H_A$ , it is clear that coherence in the eigenbasis of  $H_A$  (or, more precisely, asymmetry with respect to the group of transformations with unitary representation  $U_A = e^{-i\varphi H_A}$ ) [70] is the essential resource for the estimation. Since maximal coherence in a known reference basis can be achieved by a superposition state of subsystem A only, there is no need for any correlated idler ancilla B at all in this conventional setting.

In [57], the rules of the game were changed by considering a "black box" paradigm in which the eigenbasis of the generator  $H_A$  is completely unknown a priori (while only its spectrum is known). We can imagine e.g. a referee controlling the local dynamics on A (i.e. like an examiner who is setting an exam), who decides the basis of  $H_A$  (i.e. the question to ask) only after the experimenters selected the input state  $\rho_{AB}$  (i.e. completed their preparation), and then discloses the choice (i.e. asks the question) so that optimal measurements can still be performed at the output stage (i.e. the best possible answer can be returned, given the initial preparation). Assuming the referee is fully adversarial (i.e. the unlucky case that the examinees get asked the question they are least prepared for), the meaningful figure of merit for the protocol has to be defined in a worst case scenario, by considering the minimum achievable precision over all possible bases of  $H_A$ . In formulae, this corresponds to minimising the quantum Fisher information,

$$\inf_{H_A} C^{\rm QF}(\rho_{AB}, H_A \otimes \mathbb{I}_B) =: \mathcal{Q}_A^{C^{\rm QF}}(\rho_{AB}), \qquad (156)$$

over all local generators  $H_A$  of fixed spectral class. This is exactly the definition of the interferometric power  $Q_A^{C^{QF}}(\rho_{AB})$  discussed in Section 3.8.4, whose operational meaning is now very clear: the degree of QCs of any state  $\rho_{AB}$  measured by the interferometric power amounts to the minimum precision that the state  $\rho_{AB}$  guarantees for the estimation of a parameter  $\varphi$  encoded in a local unitary dynamics on A, in the worst case scenario in which the eigenbasis of the generator of said unitary is initially completely unknown. The more the QCs content of the state  $\rho_{AB}$  according to  $Q_A^{C^{QF}}$ , the more the state  $\rho_{AB}$  will be useful for phase estimation on A with respect to any possible non-trivial generator (i.e., in our analogy, the better the candidates will be confident to respond to any possible exam question, securing a higher base mark).

This connection was investigated experimentally in [57] in a nuclear magnetic resonance implementation of black box quantum metrology, in which classical-quantum states were shown to be unable to estimate the parameter in case of a most adverse setting of  $H_A$ , while states with nonzero QCs were found to successfully accomplish the task in all tested settings, with precision bounded from below by their interferometric power. The paradigm of black box metrology was also extended to the technologically relevant case of optical interferometry with Gaussian states and local Gaussian unitary dynamics in [202], in particular elucidating the roles of QCs, entanglement, and state mixedness in order to maximise the performance in the phase estimation task.

While the above results provide a concrete scenario in which QCs beyond entanglement are found to play centre stage, in practice one may want to assess instead the versatility of input states  $\rho_{AB}$  in terms of their average metrological performance, rather than their worst case scenario only. One can then introduce alternative figures of merit e.g. by replacing the minimum with an average according to the Haar measure in Eqs. (129). Such a study has been carried out in [265] by defining the average local Wigner-Yanase skew information  $\overline{Q}_A^{C_{WY}}(\rho_{AB})$  (see Table 5). Unlike the minimum, which defines the local quantum uncertainty  $Q_A^{C_{WY}}(\rho_{AB})$ , the average local skew information is found not to be a measure of QCs anymore. In particular, it vanishes only on states of the form  $\frac{I_A}{d_A} \otimes \tau_B$ , that is, tensor product states between a maximally mixed state on A, and an arbitrary state on B [265]. This entails that, to ensure a reliable discrimination of local unitaries on average (rather than in the worst case), the input states need to have two resource ingredients: local purity of the probing subsystem A, and correlations (of any nature) with the ancilla. The interplay between the average and the minimum performance, as well as a study on the role of entanglement, are detailed in [265].

Let us finally mention an earlier study of QCs in metrology. The authors of [266] investigated unitary phase estimation using *N*-qubit probe states (with phase transformation applied to each qubit) initialised in mixed states with a) no correlations; b) only classical correlations; or c) non-classical correlations (QCs or entanglement); all classes of probe states having the same (tunable) degree of purity for fairness of comparison. They found that uncorrelated and classically correlated probes resulted in a quantum Fisher information scaling linearly with *N*, while quantum strategies allowed for a quadratic scaling  $N^2$ , as expected in quantum metrology [263]. For the particular classes of probe states considered in their work for part c), they observed that the enhancement, compared to case a), persisted

even when the mixedness was so high that entanglement disappeared. Therefore they argued that QCs (which were found to increase with N according to a measure based on the relative entropy distance from the set of N-qubit fully classical states, i.e. the multipartite extension of the relative entropy of quantumness) may be responsible for this enhancement. It is however still unclear whether these conclusions are special to the selected classes of states, or can be further extended to more general settings.

4.2.2. State discrimination and quantum illumination There exist direct links between measures of QCs and the task of ambiguous quantum state discrimination, which in turn plays a key role both in quantum communication and cryptography [267]. In this protocol, a family of *n* known states  $\{\rho_i\}_{i=1}^n$  encodes a message. A sender randomly selects the states from this family via a probability distribution  $\{p_i\}_{i=1}^n$  and gives them one by one to a receiver, whose task is to identify them and thus decode the message. To do this, the receiver performs a generalised measurement  $\{\mu_i\}_{i=1}^n$  with *n* outcomes on each of the states given to them by the sender and concludes that the received state is  $\rho_j$  when the measurement outcome is the *j*<sup>th</sup> one. Since the states  $\{\rho_i\}_{i=1}^n$  need not be orthogonal, there is in general no measurement that can perfectly distinguish them, so that the best the receiver can do is to perform a measurement minimising the probability of equivocation. Such optimal measurement is the one maximising the so-called success probability  $P_{succ} = \sum_{i=1}^n p_i P_{i|i}$ , where  $P_{i|i} = \text{Tr}(\mu_i[\rho_i])$  is the probability of getting the *i*<sup>th</sup> result provided that the given state is the *i*<sup>th</sup> one.

In [127, 33] the case of  $\{\rho_i\}_{i=1}^n$  being states of a bipartite system *AB*, with *n* equal to the dimension of subsystem *A*, has been studied in order to investigate the role played by QCs in ambiguous quantum state discrimination. We note that *A* and *B* must not be confused with the sender and the receiver. They are simply the subsystems of the bipartite quantum system that the sender gives to the receiver each time. Therefore, the receiver can then perform a generalised measurement on the whole bipartite quantum system in order to decode the message. In this case, the maximal fidelity between the state after encoding according to the receiver,  $\rho_{AB} = \sum_{i=1}^{n} p_i \rho_i$ , and the set  $\mathcal{C}_A$  of classical-quantum states, is proven to be exactly the maximal success probability in the ambiguous quantum state discrimination of the states  $\{\rho_i\}_{i=1}^n$  with prior probabilities  $\{p_i\}_{i=1}^n$ , i.e.,

$$\max_{\chi_{AB}\in\mathcal{C}_A} F(\rho_{AB}, \chi_{AB}) = P_{\text{succ}},$$
(157)

that implies

$$Q_A^{G_{\text{Bu}}}(\rho_{AB}) = \sqrt{2\left(1 - \sqrt{P_{\text{succ}}}\right)}.$$
(158)

In particular, this holds due to the fact that the optimal LGM  $\{\mu_i\}_{i=1}^n$  that maximises the success probability is found to be a complete rank one LPM on subsystem A. This provides us with the following direct link between the one-sided Bures geometric measure of QCs  $Q_A^{G_{Bu}}(\rho_{AB})$  and the considered protocol: the more the state after encoding according to the receiver,  $\rho_{AB} = \sum_{i=1}^n p_i \rho_i$ , is close to be classical-quantum, the more successful the receiver will be in the discrimination task by performing a local von Neumann measurement on subsystem A.

"Quantum illumination" [268, 269, 270] stands as a paradigmatic example of a quantum state discrimination task. The conventional illumination protocol consists of a probe qudit, in a pure state  $\rho_{\phi} = |\phi\rangle\langle\phi|$ , sent into a distant noisy region to detect the possible presence of a target object. If the target is not there, the probe is completely lost and the sender detects only environmental noise described by the maximally mixed state  $\rho_E = \mathbb{I}/d$ . Even if the target is actually present, there is only a small probability  $\eta$  that the probe is reflected back and then detected by the sender, whereas the majority of time the probe is lost and again only random

noise is detected. Overall, by assuming that the presence and absence of the target happen with the same prior probability, such a conventional protocol is equivalent to a quantum state discrimination task between the state  $\rho_c^{(0)} := \rho_E$ , corresponding to the absence of the target, and the state  $\rho_c^{(1)} := \eta \rho_{\phi} + (1 - \eta)\rho_E$ , corresponding to the presence of the target, with equal a priori probabilities  $p_0 = p_1 = 1/2$ . More precisely, the better the sender can discriminate between the states  $\rho_c^{(0)}$  and  $\rho_c^{(1)}$ , the more confident they will be in inferring whether or not the target is actually there.

In [271], the accuracy of the discrimination between two states  $\rho_0$  and  $\rho_1$  with a priori probabilities  $p_0$  and  $p_1$ , respectively, has been quantified by the corresponding Shannon distinguishability, i.e.,  $D_{\mathcal{H}}(\rho_0, \rho_1) := \max_{\mathcal{M}} \mathcal{J}(X : X_o^{\mathcal{M}})$ , where  $\mathcal{J}(X : X_o^{\mathcal{M}})$  is the classical mutual information, Eq. (84), between the variable X with probability distribution  $\{p_0, p_1\}$  and the variable  $X_o^{\mathcal{M}}$  obtained as output by performing the measurement  $\mathcal{M}$ . Therefore, the best case scenario within the conventional approach to illumination can be quantitatively identified by a figure of merit  $I_c^{\max} := \max_{\phi} D_{\mathcal{H}}(\rho_c^{(0)}, \rho_c^{(1)})$ .

In a quantum illumination protocol assisted by non-classical correlations, the sender improves their strategy by maximally entangling the probe A with an idler ancilla B (which is retained at the sending station) by creating the pure global state  $\rho_{\psi_{AB}} = |\psi\rangle\langle\psi|_{AB}$ , where  $|\psi\rangle_{AB} = (1/\sqrt{d}) \sum_k |k\rangle_A \otimes |k\rangle_B$  and  $\{|k\rangle\}$  is a qudit orthonormal basis (d being the dimension of both probe A and ancilla B). Now, the corresponding equivalent quantum state discrimination task is between the state  $\rho_{AB}^{(0)} := \rho_E \otimes \rho_B$  with a priori probability  $p_0 = 1/2$ , corresponding to the absence of the target, and the state  $\rho_{AB}^{(1)} := \eta \rho_{\psi_{AB}} + (1 - \eta)\rho_E \otimes \rho_B$  with a priori probability  $p_1 = 1/2$ , corresponding to the presence of the target, where  $\rho_B = \text{Tr}_A(\rho_{\psi_{AB}}) = \mathbb{I}/d$  represents the reduced state of the ancilla B. Such quantum illumination protocol thus outperforms its conventional counterpart when it is easier to distinguish  $\rho_{AB}^{(0)}$  from  $\rho_{AB}^{(1)}$  for any conventional input  $\rho_{\phi}$ , i.e., when  $I_q > I_c^{\text{max}}$  with  $I_q := D_{\mathcal{H}}(\rho_{AB}^{(0)}, \rho_{AB}^{(1)})$ . The corresponding advantage can be quantified by the difference  $\Delta I := I_q - I_c^{\text{max}}$ . It is interesting to pin down the exact origin of such a quantum advantage, since the entanglement initially present between probe and idler can all be destroyed due to the effects of the noise acting on the probe in the target region [272].

In [271] an answer has been provided by showing a direct equality between the performance gain in quantum illumination over any conventional illumination protocol and the amount of quantum discord between the probe and its ancilla which is consumed to detect the target, thus linking the expenditure of QCs to the resilient enhancement of quantum illumination. More precisely:

$$\Delta I = \Delta Q_A^{I_I},\tag{159}$$

where  $\Delta Q_A^{\tilde{I}_I} := Q_A^{\tilde{I}_I}(\bar{\rho}_{\psi_{AB}}) - \overline{Q}_A^{\tilde{I}_I}$ , with  $\bar{\rho}_{\psi_{AB}} := p_0 \rho_{\psi_{AB}}^{(0)} + p_1 \rho_{\psi_{AB}}^{(1)}$  being the state of probe and idler ancilla before detecting the target and  $\overline{Q}_A^{\tilde{I}_I} := p_0 Q_A^{\tilde{I}_I}(\rho_{\psi_{AB}}^{(0)}) + p_1 Q_A^{\tilde{I}_I}(\rho_{\psi_{AB}}^{(1)})$  being the average quantum discord between the probe and the idler ancilla after detecting the target.

Another direct connection between QCs and ambiguous quantum state discrimination has been provided by [273], where the following protocol has been considered. Two laboratories A and B share some known quantumly correlated state  $\rho_{AB}$ . Laboratory A privately encodes the value  $x_i$  of a random variable X with probability  $p_i$  onto their subsystem through the application of a corresponding local unitary  $U_A^{(i)}$ . Then, A gives their subsystem to B, who is asked to decode the encoded value of the random variable X by carrying out some measurement  $\mathcal{M}$  on the whole bipartite system. This is clearly an ambiguous quantum discrimination between the states  $\rho_i := U_A^{(i)}[\rho_{AB}]$  with a priori probabilities  $p_i$ . In analogy with [271], the accuracy of laboratory B's performance is quantified by resorting to the classical mutual information  $\mathcal{J}(X : X_o^{\mathcal{M}})$  between the encoded variable X and the decoded variable  $X_o^{\mathcal{M}}$ . Now, let use denote by  $I'_q$  the best possible performance that B can achieve when they can carry out any measurement, and by  $I'_c$  the best possible performance that B can achieve when they are restricted only to a single local measurement on each of A and B, followed by classical post-processing. The quantity  $\Delta I' := I'_q - I'_c$  thus represents the advantage provided by coherent interactions between A and B over no interaction, in performing local measurements within the decoding task.

In [273] it has been shown that there always exists an optimal choice of local unitaries  $U_A^{(i)}$  (for instance, the Pauli operators if  $\rho_{AB}$  is a two-qubit state) such that

$$\Delta I' = \Delta Q_A^{I_I},\tag{160}$$

where  $\Delta Q_A^{\tilde{I}_I} := Q_A^{\tilde{I}_I}(\rho_{AB}) - Q_A^{\tilde{I}_I}(\rho'_{AB})$ , with  $\rho'_{AB} := \sum_i p_i U_A^{(i)}[\rho_{AB}]$  being the state after encoding by party A, thus proving that the quantum discord consumed during encoding yields exactly the extra quantum advantage that B can gain by allowing coherent interactions. This was also demonstrated experimentally in [273] in a continuous variable optical setup based on Gaussian states and Gaussian encoding operations.

The last operational interpretation of QCs within ambiguous quantum state discrimination that we review is the one provided in [58]. Here, the setting is that of discriminating among n copies of two states  $\rho_0$  and  $\rho_1$ . In particular, by assuming that the a priori probabilities of getting  $\rho_0$  and  $\rho_1$  are equal, as it happens e.g. in a quantum illumination protocol, it is known that for  $n \gg 1$  the minimal probability of error in distinguishing between these two states, using the optimal discrimination strategy due to Helstrom [125], scales approximately as  $e^{-n\xi(\rho_0,\rho_1)} = C(\rho_0,\rho_1)^n$ , where  $\xi(\rho_0,\rho_1)$  is the quantum Chernoff bound defined in Eq. (125) while  $C(\rho_0, \rho_1)$  is defined in Eq. (123). Moreover, the complementary Chernoff distance  $D_{\rm C}(\rho_0,\rho_1) = 1 - {\rm C}(\rho_0,\rho_1)$  turns out to be nothing but the measure of distinguishability adopted in the definition of the discriminating strength in Eq. (122). When  $\rho_0 := \rho_{AB}$ and  $\rho_1 := U_A^{\Gamma}[\rho_{AB}]$ , this coincidence entails that such a measure of QCs quantifies operationally the guaranteed accuracy in the asymptotic discrimination between the state  $\rho_{AB}$  and the transformed state  $U_A^{\Gamma}[\rho_{AB}]$ , in the worst case scenario in which only the spectrum  $\Gamma$  of the unitary operation  $U_A^{\Gamma}$  is known, and a minimisation is considered over all such unitaries with fixed spectrum  $\Gamma$ . This task bears some analogy with the black box metrology protocol discussed in Section 4.2.1, indeed both settings provide direct operational interpretations for QCs quantifiers in worst case scenarios, with the difference that there the goal was to estimate the parameter imprinted by a local unitary  $U_A$  on the probe subsystem A, while here the goal is just to determine whether a local unitary  $U_A$  was applied or not to the probe subsystem A. This task has also been considered in continuous variable systems in [199] by restricting  $\rho_{AB}$ and  $U_A$  to Gaussian states and operations, respectively.

An analogous analysis applies to establish the operational role played by the Hellinger discord of response  $Q_A^{U_{\text{He}}}$  in quantum reading of classical digital memories [274, 275], as demonstrated in [197] in a continuous variable setting restricted to Gaussian states and operations. The authors of [197] further provided an operational interpretation for the trace distance based Gaussian discord of response in terms of (one minus) the maximum error probability in quantum reading of a classical memory. The study in [197] includes as well a detailed characterisation of the possible enhancements to QCs and to the corresponding quantum advantage due to thermal noise in the considered task.

Finally, we notice that the presence and use of QCs has also been investigated in other tasks related to quantum illumination, such as ghost imaging [276, 277].

### 4.3. Further physics applications

One of the reasons why entanglement theory has gained so large an acclaim can be traced to its successful cross-fertilisation into other important areas of physics beyond quantum information theory, such as condensed matter, atomic and optical physics, statistical mechanics, and cosmology. As QCs capture all trademark effects of quantum mechanics, including and beyond entanglement, it seems fit to investigate further the role and applicability they can have in different branches of physics, not limited to those where the benefits of entanglement can already be appreciated. Here we will focus on two particular domains.

4.3.1. Quantum thermodynamics The concept of QCs arises naturally in the thermodynamics of quantum systems, or "quantum thermodynamics" [278, 279], particularly in the context of work extraction [280, 10] from a bipartite quantum system in contact with a thermal reservoir, a fundamental task for which entanglement is not necessary [281]. Specifically, informational measures of QCs exactly quantify the difference between the maximal work that can be extracted by a Maxwell's demon, who is an entity that perfectly knows the state of the composite system and is able to perform any global operation on it, and the maximal work that can be extracted by two goblins, who are less powerful entities whose knowledge about the state of the composite system can be limited and are able to perform just some subclass of LOCC. A different quantifier of QCs arises depending on the limitations on the goblins' knowledge and the subclass of LOCC that they are able to perform. Such QCs quantifiers are also referred to as "demon discords" [142].

On the one hand, due to the Landauer's principle, the maximal work that a Maxwell's demon can extract from a bipartite quantum system *AB* in a state  $\rho_{AB}$  and in contact with a thermal reservoir at temperature *T* is given by

$$W_{\text{demon}}(\rho_{AB}) := kT(\log d_{AB} - \mathcal{S}(\rho_{AB})), \tag{161}$$

where k is Boltzmann's constant and  $d_{AB} = d_A d_B$  is the dimension of the composite system AB.

On the other hand, if the goblins perfectly know the state  $\rho_{AB}$  and one goblin (operating on *A*) can perform any complete rank one LPM  $\Pi_A$  and communicate classically the result to the other goblin (operating on *B*), then the maximal work that they can extract is given by

$$W_{\text{goblins}}(\rho_{AB}) := kT \Big( \log d_{AB} - \min_{\Pi_A} \mathcal{S}(\Pi_A[\rho_{AB}]) \Big), \tag{162}$$

in such a way that in this case

$$\Delta W(\rho_{AB}) := W_{\text{demon}}(\rho_{AB}) - W_{\text{goblins}}(\rho_{AB}) = kT Q_A^{I_S}(\rho_{AB}), \tag{163}$$

where  $Q_A^{I_S}$  is the thermal discord introduced in [94], which also coincides with the one-way quantum deficit introduced in [87], as already mentioned in Section 3.5.1. In fact, the same conclusions are drawn when the goblins are restricted to the so-called closed local operations and classical communication (closed LOCC), i.e., LOCC that cannot change the total number of particles and consist only of local unitary operations and local projective measurements, with one-way communication allowed from A to B [87]. Furthermore, if each of the goblins performs a complete rank one LPM on their subsystem and classical communication is allowed only after the measurements, the zero-way quantum deficit  $Q_{AB}^{I_S}(\rho_{AB})$  is recovered instead [87].

Another possibility occurs when the goblins do not know the global state  $\rho_{AB}$ , but rather only their corresponding reduced states  $\rho_A$  and  $\rho_B$ , while one of them can still perform LPMs on A and communicate classically the result to the other [30]. The maximal work that can be extracted by these even less powerful goblins is given by

$$W_{\text{goblins}}^{\star}(\rho_{AB}) := kT \left( \log d_{AB} - \mathcal{S}(\Pi_{A}^{\star}[\rho_{AB}]) \right), \tag{164}$$

where  $\Pi_A^{\star}$  is the LPM onto the eigenbasis of  $\rho_A$ . In this case we have that

$$\Delta W^{\star}(\rho_{AB}) := W_{\text{demon}}(\rho_{AB}) - W^{\star}_{\text{goblins}}(\rho_{AB}) = kT Q_A^{I_S}(\rho_{AB}), \tag{165}$$

where  $Q_A^{I_S^*}(\rho_{AB})$  is the increase in the von Neumann entropy of the composite system AB by performing the projective measurement  $\Pi_A^*$  onto the eigenbasis of the reduced state of subsystem A, i.e.,

$$Q_A^{I_S^{\star}}(\rho_{AB}) := \mathcal{S}\left(\Pi_A^{\star}[\rho_{AB}]\right) - \mathcal{S}\left(\rho_{AB}\right).$$
(166)

The role of QCs, measured specifically by the quantum discord  $Q_A^{\bar{I}_T}$ , has been studied also within the context of quantum thermal machines. In particular, in [282], a quantum machine that needs to drive the temperature of a body outside of the range defined by the surrounding thermal reservoirs has been considered, and it has been shown that not only the machine-body quantum discord is necessary to this aim but also that a high machine efficiency at maximum power is strongly connected to the presence of a peak in such a QCs measure, thus showing the role of the latter as a quantum thermal machine resource. In the same spirit, in [283] it has been shown that QCs drive the Szilard engine containing a diatomic molecule with a semipermiable wall, in the sense that the work that can be extracted from this engine is lower bounded by the quantum discord between the two atoms of the molecule. Furthermore, in [284] it has been shown that the quantum discord shared by the two atoms constituting the quantum reservoir of a photo-Carnot engine can increase the thermodynamic efficiency of the latter as opposed to the case where these two atoms have no QCs, thus demonstrating once again the role of QCs as a thermodynamic resource.

The presence of quantum discord in the operation of quantum absorption refrigerators modelled as three qubits in contact with three baths was also investigated in [285], where it was found that QCs are always present between the qubit which is being cooled (cold qubit A) and the relevant subspace of the other two qubits which is in direct interaction with the cold qubit (machine virtual qubit B), although  $Q_A^{I_T}$  was found to play no quantitative role in the optimisation of the coefficient of performance or the cooling power. On the contrary, in [286] in a simple model of quantum feedback cooling of a two-level system using an interacting ancilla, akin to algorithmic cooling, the cooling performance was found to be closely related to the discord  $Q_A^{I_T}$  (and not to entanglement or total correlations) built up between system Aand ancilla B in the pre-measurement step, before closing the feedback loop; in particular, the performance characteristics of the protocol were shown to be curves of constant quantum discord, with the optimal performance identified by a peak in such a QCs measure.

Another thermodynamic interpretation of QCs can be found when considering quantum predictive processes, wherein a so-called predictive quantum system S is used to predict the future dynamics of a given portion X of its surrounding environment by exploiting the correlations between S and X. In [287] it has indeed been shown that the quantum discord  $Q_X^{I_I}$  between the predictive system S and the target portion of its surrounding environment X exactly quantifies the advantage in the predictive power over the most efficient classical predictive strategy, which in turn coincides with the corresponding reduction in the dissipated

extractable work from the system S. This thus stands as another deep connection between efficient use of quantum information and efficient thermodynamical operation as powered by QCs. It may be interesting to further investigate links between predictive processes and the emergence of objective reality in quantum Darwinism, in particular in view of Eq. (142) [216].

Finally, in [288] it has been shown that any energy transport must necessarily generate QCs. In particular, over a sufficiently short time  $\Delta t$ , the flow of heat  $\Delta E_A$  from a *B* to another system *A*, both initially prepared in thermal states at different temperatures  $T_A$  and  $T_B$ ,  $T_B > T_A$ , is directly proportional to the corresponding increase in the one-way QCs as measured by the so-called diagonal discord  $Q_A^{I_A^*}$ , i.e.,

$$\Delta E_A = (\beta_A - \beta_B)^{-1} \Delta Q_A^{I_A^*}, \qquad (167)$$

where  $\beta_A = 1/kT_A$ ,  $\beta_B = 1/kT_B$ , and finally the diagonal discord is the reduction in the mutual information of the composite system *AB* by performing the LPM  $\Pi_A^*$  onto the eigenbasis of the reduced state of subsystem *A*, i.e.,

$$Q_A^{I_I^{\star}}(\rho_{AB}) := \mathcal{I}(\rho_{AB}) - \mathcal{I}(\Pi_A^{\star}[\rho_{AB}]).$$
(168)

The diagonal discord in Eq. (168) is the mutual information based analogue to the von Neumann entropy based informational quantifier in Eq. (166), and can be seen as the one-sided counterpart to the measurement induced disturbance discussed in Section 3.5.2. Notice that, due to the lack of an optimisation over LPMs, both quantities in Eqs. (166) and (168) are not, in general, valid measures of QCs, yet they both obey Requirement (i), thus being able to faithfully discriminate between classically and quantumly correlated states.

4.3.2. Many-body systems and quantum phase transitions General QCs represent not only essential ingredients for quantum technologies and thermodynamics, but are also a very useful tool for the characterisation of the low temperature macroscopic phases of many-body quantum systems [289, 290]. In particular, the analysis of the QCs of quantum ground states can be exploited to detect the points in the low temperature macroscopic phase space where the system passes from one phase to another, a process referred to as "quantum phase transition".

Quantum phase transitions typically occur at or near absolute zero temperature, where the de Broglie wavelength is greater than the correlation length of the thermal fluctuations, by varying a parameter of the system's Hamiltonian that is known as tuning parameter. This change of phase is only caused by quantum fluctuations stemming from the Heisenberg uncertainty relation, which still survive in the low-temperature regime, as opposed to a classical phase transition that is instead driven by thermal fluctuations. More precisely, when the tuning parameter crosses a particular value, known as critical point, the system's ground state undergoes a dramatic qualitative change. The latter can consist of either the breaking of some symmetry of the system's Hamiltonian and ensuing appearance of a local order or the variation of the topological order of the system's phase as characterised by a nonlocal order parameter based on the pattern of the long-range two-body ground state correlations.

After an initial race to detect the occurrence of quantum phase transitions through the analysis of both the two-body and global entanglement properties of quantum ground states [291], QCs more general than entanglement have been adopted for this purpose for the first time in [292], when considering in particular the one-dimensional spin-1/2 transverse Ising and antiferromagnetic XXZ models. Specifically, there the critical points have been identified by looking at the singularities of the derivatives of the zero-temperature nearest- and next-nearest-neighbour two-body quantum discord in the thermodynamic limit. In [293], such

analysis has been generalised by discussing also the finite size scaling of the two-body nearestneighbour quantum discord and then extended to the Lipkin-Meshkov-Glick model.

In both [292] and [293] only the zero-temperature regime was considered. However, according to the third law of thermodynamics, the absolute zero temperature is unreachable. This makes a low-temperature extension of the aforementioned analyses compelling, thus inevitably motivating the studies that have been carried out [294, 295, 296]. Quite remarkably, in those works it has been shown that, when considering the one-dimensional spin-1/2 XXZ model, the two-body quantum discord spotlights the quantum phase transitions even at finite temperature. On the contrary, the two-body entanglement and other thermodynamic quantities such as entropy, specific heat, magnetic susceptibility, or the two-body (classical) correlation functions, are not able to detect the corresponding quantum phase transitions at finite temperature, thus making bipartite QCs stand out as particularly useful tools in the quest for the experimental detection of quantum phase transitions in the one-dimensional spin-1/2 XXZ model.

However, recently the authors of [297] have pointed out that the two-body quantum discord may provide essentially no information on the quantum nature of two-body correlations in a system at finite temperature. A more appropriate and physically motivated definition of QCs is instead proposed in [297], leading to the introduction of the so-called "quantum correlation function", which stands in full analogy with the ordinary correlations. Interestingly, the quantum variance introduced in [203] and discussed in Sec. 3.8.5 is obtained by integrating the quantum correlation function, in the same way as the ordinary variance of a physical quantity is obtained by integrating the ordinary correlation function.

In the same spirit as the finite-temperature analyses, in [298, 299, 300, 301, 302, 303] the long-range nature of two-body QCs and their scaling with the inter-spin distance have been exploited in order to detect the critical point of the one-dimensional spin-1/2 XY model. Again, the analysis of long-ranged correlations becomes instead infeasible when focusing on two-body entanglement, due to its short-ranged nature.

Other studies then complemented the above ones when either taking symmetry breaking into account [304, 305] or considering more sophisticated quantum phase transitions such as the ones in the one-dimensional spin-1/2 XY model with Dzyaloshinskii-Moriya interaction [306], the Dicke model [307], the Castelnovo-Chamon model [308], cluster-like systems [309], as well as spin-1 XXZ and bilinear-biquadratic chains [310, 168, 311]. Most of the above works adopted the quantum discord or the one-way quantum deficit as measures of QCs, but other valid and more efficiently computable quantifiers such as the local quantum uncertainty were also proven of use in the characterisation of quantum critical points [303, 312, 311].

Finally, in [313] the "global quantum discord" was defined as a multipartite measure of QCs by extending a relative entropy based definition of the quantum discord to many-body systems. The global discord was shown able to characterise the infinite-order quantum phase transition in the Ashkin-Teller spin chain, which could not be detected by using the two-body discord alone. An extended study of one-dimensional spin models at finite temperature in [314] showed that the global discord could detect critical points even in many-body systems out of their ground state; interestingly, it was also found that the global discord scales with universal critical exponents in the proximity of a quantum phase transition in the Ising model. Quantum and global discord were further analysed in spin-1 Heisenberg chains subject to single-ion anisotropy in [315]; these measures were shown to detect the quantum phase transitions confining the symmetry-protected Haldane phase, and to exhibit critical scaling with universal exponents.
## 5. Concluding remarks and outlook

In this review we summarised recent progress in the characterisation of the most general forms of QCs in composite quantum systems, focusing in particular on bipartite states. We devoted particular attention in Section 2 to illustrate the physical significance of the concept of QCs, providing a polyhedral description of how quantumly correlated states differ from classically correlated ones (see Fig. 3). Such a qualitative analysis then formed the basis to introduce in Section 3 a plethora of different measures of QCs (see Tables 1 and 6), each capturing quantitatively one or more defining aspects of quantumly correlated states. The operational significance of several of the measures was then explored in Section 4, where various relevant applications of QCs to quantum information and to physics more broadly were described.

As the topic of QCs is still under development, there remain lots of open questions to investigate in future research. The attentive reader will have found a bunch of them already interspersed in the previous Sections. One of the most pressing matters to address is certainly the development of a consistent resource theory of QCs, more specifically to formulate the appropriate set of free operations under which any *bona fide* measure of QCs should be monotonically non-increasing. We have proposed one such set, namely the local commutativity preserving operations, formalised in Eqs. (43) and (44) of Section 3.1, and used them to postulate a strong monotonicity for measures of QCs as a desideratum, tagged Requirement (v). Fruitful next steps could be either to systematically validate existing measures against this Requirement (thus clearing as many question marks as possible from the last column of Table 6), or to tweak such a Requirement if deemed necessary from a physical point of view. Perhaps insights from the resource theories of entanglement [16, 3] and coherence [69, 43, 44, 72, 84], and their respective free operations, could guide such an endeavour, in view of the close connections that both of these resources have with QCs (refer e.g. to Definitions 3 and 5).

Nevertheless, we have also highlighted some measures of QCs which are already proven to be in full compliance with the Requirements of Section 3.1. Perhaps the most satisfactory ones are the one-sided and two-sided geometric measures  $Q_A^{G_{RE}}$  and  $Q_{AB}^{G_{RE}}$  defined in terms of relative entropy, which are shown to be further equivalent to a variety of other types of measures. These measures enjoy thus multiple fundamental interpretations, namely in terms of: the minimal state distinguishability from any classical state (geometric approach); the minimal state distinguishability following a least perturbing local measurement (measurement induced geometric approach); the minimum loss of global information, or equivalently the minimum added noise, due to a least perturbing local measurement (informational approach in terms of von Neumann entropy); the minimum distillable entanglement created with an ancillary system at the pre-measurement stage during a least perturbing local measurement (entanglement activation approach); and the minimum quantum coherence in any local basis (coherence based approach in terms of relative entropy); plus, they have a natural operational meaning as quantum deficits in the context of thermodynamical work extraction. To the best of our knowledge, the measures of QCs based on relative entropy have not been studied explicitly in either the unitary response or the recoverability approach. It is tempting to conjecture that both of them might reduce as well to the corresponding geometric quantifiers, hence closing the circle of equivalences, and providing yet further physical interpretations for such measures of QCs. We leave this as an open question to investigate for the interested reader.

In this respect, let us also point out that the surprisal of measurement recoverability looks quite lonely in its partition of Table 6, and it will be worth studying more explicit measures of QCs within the recently formulated recoverability approach [50], in particular in view of possible operational connections with practical tasks, as discussed in Section 4.1.1

for local broadcasting. Within such a setting, let us also recall the fundamental interpretation revealed for the quantum discord as the share of correlations that cannot be redistributed to infinitely many subsystems in a one-sided scenario [217, 68]. As remarked after Eq. (142), a deserving next step could be to formalise a similar connection in the two-sided case, and study its implications for the quantum to classical transition.

Further connections highlighted through the exploration of quantitative measures of QCs are also inspiring. For instance, the equivalence between the two-sided negativity of quantumness (an entanglement activation type of measure) and the quantum coherence measured by the  $\ell_1$  norm minimised over all local product bases (capturing the seminal notion of coherence in terms of interference effects), which bridges in particular the interpretations illustrated by Fig. 3(b) and (d), reveals once more that QCs incarnate truly fundamental signatures of quantum mechanics, blending together other more widely celebrated phenomena like entanglement, superposition, non-commutativity, and the perturbing effect of measurements. This is equally, and maybe even more strongly, represented in the chain of equivalences discussed above for relative entropy based measures.

Let us remark that such basic ingredients as the non-commutativity of a bipartite quantumly correlated state with any local observable may be seen as a negative feature, i.e. an unavoidable uncertainty affecting the measurement of any single local observable (as captured by the local quantum uncertainty  $Q_A^{C^{WY}}$ ) [54], but also, crucially, as a positive feature, since all quantumly correlated states are necessarily modified by — hence sensitive to — any local dynamics with a fully non-degenerate spectrum. This is exactly the feat that makes QCs a natural resource for metrology and discrimination tasks in worst case scenarios, as discussed in Section 4.2. Further study of the concrete implications of these observations for quantum technologies would be welcome.

In this review we mainly spoke about bipartite QCs. Several of the measures introduced extend straightforwardly to multipartite systems, in particular the geometric ones (by defining the set of fully classical states as the states diagonal with respect to a product of orthonormal bases for each subsystem [24, 93], and any set of partially classical-quantum states with respect to a selection of subsystems treated as classical, as detailed in [28]), the measurement induced geometric and informational ones (again depending on a selection of subsystems on which local measurements are applied), the entanglement activation ones [35, 36, 28], and the recoverability ones [316]. As previously pointed out, it will be an interesting future direction to develop consistent generalisations of the unitary response and coherence based (in the particular case of asymmetry measures) approaches to define faithful two-sided and more general multipartite measures of QCs, especially in view of the operational merits that these types of measures exhibit in the one-sided case for bipartite systems. Some measures of multipartite QCs have already been defined, such as the global discord [313] (which finds applications in the study of quantum critical points in many-body systems, see Section 4.3.2), the so-called quantum dissension [317], and a few quantifiers of genuine QCs [318, 319]. More general (measure-independent) features of QCs in multipartite systems have also been explored, demonstrating in particular that, unlike entanglement measures [320], no measure of QCs beyond entanglement can satisfy a conventional monogamy inequality on all states of three or more parties [74]. However, alternative monogamy constraints that impose trade-offs between QCs and other resources in tripartite systems, such as entanglement, coherence, and local entropies, can be derived [176, 177, 222, 223, 85, 321, 322]. Even more generally, the results reviewed in Section 4.1.1 reveal that QCs beyond entanglement do obey fundamental limits to their shareability, which is another testament to their true quantum nature, as opposed to the case of freely shareable classical correlations. In this context it is interesting to point out that, while we focused on general QCs within the hierarchy of correlations illustrated in Fig. 1, there also exist finer layers of correlations that can be identified in quantum states, for instance intermediate between QCs and entanglement, as investigated e.g. in [323].

As announced in the Introduction, we have not discussed any study of the dynamical properties of QCs in open quantum systems, even though there are lots of interesting results in such a direction. In particular, due to the geometry of the set of classical states [8], QCs cannot vanish at finite time for any typical dynamics [324, 325], making them by construction more resilient than entanglement. In certain cases, such a resilience can be extreme, as QCs in bipartite and multipartite systems can remain constant (frozen) in time under local decohering maps even though the global state is evolving [326]; this happens for particular classes of states and dynamical conditions. Within the geometric approach to QCs (Section 3.2.1), it has been shown in [55] that, under the allowed conditions, such features are universal and occur for all measures  $Q^{G_{\delta}}$  independently of the specific choice of the distance  $D_{\delta}$  in their definition. More details can be found in [55] and references therein, as well as [12], to which the reader is referred for a collection of pertinent literature (including a number of experimental demonstrations). Note that such freezing phenomena also manifest in the study of quantum coherence [48], under the same dynamical conditions as for QCs, which provides a further objective evidence of the intimate relationship between these two quantum effects.

One historical perspective that certainly needs clearing up is the question of whether QCs beyond entanglement can be responsible for super-classical advantages in quantum computing. Needless to say, pinning down the exact contribution of QCs to information processing presents an extremely relevant goal to the wider development of quantum technologies. In fact, after half a decade of quiescence since their initial definition [6, 7], QCs attracted a flurry of research interest [327] following the suggestion in [328] that quantum discord might be accountable for the power behind the protocol of quantum computing with one clean qubit (DQC1) [329]. This was also supported by experimental implementations of this protocol using separable states of a few qubits [330, 331]. However, numerous questions have been raised about the precise role of QCs in DQC1 [8, 25, 47], with no clear operational link developed so far between the quantum speed-up (for which a figure of merit is not itself available) and any specific quantifier of QCs. While there remains interest in using separable states with nonzero QCs in quantum algorithms [249, 332, 333], and one can even formulate precise arguments for the need of mixed state quantum computations to create QCs in order not to be classically simulatable [334, 335], it is overall a contentious topic, that we have not covered in this review; the interested reader may consult [12] for further details. We chose instead to review a wealth of novel and recent results from operational perspectives other than quantum computation, that themselves highlight the importance of QCs within communication, metrology, thermodynamics, and much more. We believe this is just the beginning to the acknowledgment of QCs as resources, and it will be of pivotal importance to pursue further new operational viewpoints of this multi-faceted phenomenon, in order to better appreciate the practical value of QCs, beyond their intrinsic foundational virtues.

To conclude, our goal with this review was to provide a comprehensive and, in some respects, original introduction to the quantum versus classical frontier, as perceived from the viewpoint of correlations between two or more parts of a composite system. In particular, we have reported detailed answers to the three basic questions posed in the Introduction, i.e.: What are the signature traits of QCs? How can we meaningfully quantify QCs? What are the practical applications of QCs? Along the way we have highlighted many possible avenues to further expand on these answers. We hope to have conveyed sufficient motivation for the reader to be enthused about the study of quantum correlations and we look forward to future progress, along some of the lines discussed here, and beyond.

## Acknowledgments

This work is supported by the European Research Council (ERC) Starting Grant GQCOP "Genuine Quantumness in Cooperative Phenomena" (Grant No. 637352), and by the Foundational Questions Institute (fqxi.org) Physics of the Observer Programme (Grant No. FQXi-RFP-1601). We are grateful to Marco Piani for many fruitful discussions and to an anonymous Referee for very constructive comments. We thank S. Piano, J. Doe, and E. D'Angelo for their patience and support during the writing of this review. We acknowledge useful feedback on an earlier draft of this manuscript by (in alphabetical order): R. Angelo, S. Campbell, N. Canosa, I. Chakrabarty, F. Fanchini, Y. Huang, A. Iqbal Singh, F. Illuminati, S. Pirandola, A. Rastegin, T. Roscilde, A. Smerzi, C. Susa-Quintero, M. Wilde, and M. Yurischev.

## References

- [1] Dowling J P and Milburn G J 2003 Philosophical Transactions of the Royal Society A 361 1655
- [2] Nielsen M A and Chuang I L 2000 Quantum Computation and Quantum Information (Cambridge: Cambridge University Press)
- [3] Horodecki R, Horodecki P, Horodecki M and Horodecki K 2009 Reviews of Modern Physics 81 865
- [4] Cavalcanti D and Skrzypczyk P 2016 arXiv preprint arXiv:1604.00501 to appear in Reports on Progress in Physics
- [5] Brunner N, Cavalcanti D, Pironio S, Scarani V and Wehner S 2014 Reviews of Modern Physics 86 419-478
- [6] Ollivier H and Zurek W H 2001 Physical Review Letters 88 017901
- [7] Henderson L and Vedral V 2001 Journal of Physics A: Mathematical and General 34 6899
- [8] Ferraro A, Aolita L, Cavalcanti D, Cucchietti F and Acín A 2010 Physical Review A 81 052318
- [9] Modi K 2014 Open Systems & Information Dynamics 21 1440006
- [10] Horodecki M and Oppenheim J 2013 International Journal of Modern Physics B 27 1345019
- [11] Streltsov A 2015 Quantum Correlations Beyond Entanglement and their Role in Quantum Information Theory (Cham: SpringerBriefs in Physics) arXiv preprint arXiv:1411.3208
- [12] Modi K, Brodutch A, Cable H, Paterek T and Vedral V 2012 Reviews of Modern Physics 84 1655
- [13] Werner R F 1989 Physical Review A 40 4277
- [14] Piani M and Watrous J 2015 Physical Review Letters 114 060404
- [15] Gisin N 1991 Physics Letters A 154 201–202
- [16] Plenio M B and Virmani S 2007 Quantum Inf. Comput. 7 1
- [17] Einstein A, Podolsky B and Rosen N 1935 Physical Review 47 777
- [18] Schrödinger E 1935 Discussion of probability relations between separated systems Mathematical Proceedings of the Cambridge Philosophical Society vol 31 (Cambridge Univ Press) pp 555–563
- [19] Wiseman H M, Jones S J and Doherty A C 2007 Physical Review Letters 98 140402
- [20] Bell J S 1964 *Physics* **1** 195–200
- [21] Hensen B, Bernien H, Drau A, Reiserer A, Kalb N, Blok M, Ruitenberg J, Vermeulen R, Schouten R, Abelln C, Amaya W, Pruneri V, Mitchell M W, Markham M, Twitchen D, Elkouss D, Wehner S, Taminiau T and Hanson R 2015 Nature 526 682
- [22] Giustina M, Versteegh M A M, Wengerowsky S, Handsteiner J, Hochrainer A, Phelan K, Steinlechner F, Kofler J, Larsson J A, Abellán C, Amaya W, Pruneri V, Mitchell M W, Beyer J, Gerrits T, Lita A E, Shalm L K, Nam S W, Scheidl T, Ursin R, Wittmann B and Zeilinger A 2015 *Physical Review Letters* 115 250401
- [23] Shalm L K, Meyer-Scott E, Christensen B G, Bierhorst P, Wayne M A, Stevens M J, Gerrits T, Glancy S, Hamel D R, Allman M S, Coakley K J, Dyer S D, Hodge C, Lita A E, Verma V B, Lambrocco C, Tortorici E, Migdall A L, Zhang Y, Kumor D R, Farr W H, Marsili F, Shaw M D, Stern J A, Abellán C, Amaya W, Pruneri V, Jennewein T, Mitchell M W, Kwiat P G, Bienfang J C, Mirin R P, Knill E and Nam S W 2015 *Physical Review Letters* 115 250402
- [24] Piani M, Horodecki P and Horodecki R 2008 Physical Review Letters 100 090502
- [25] Dakić B, Vedral V and Brukner Č 2010 Physical Review Letters 105 190502
- [26] Lang M D and Caves C M 2010 Physical Review Letters 105 150501
- [27] Aaronson B, Lo Franco R and Adesso G 2013 Physical Review A 88 012120
- [28] Piani M and Adesso G 2012 Physical Review A 85 040301
- [29] Datta A 2010 arXiv preprint arXiv:1003.5256
- [30] Brodutch A and Terno D R 2010 Physical Review A 81 062103
- [31] Coles P J 2012 Physical Review A 85 042103

- [32] von Neumann J 1932 Mathematical Foundations of Quantum Mechanics (Berlin: Springer-Verlag)
- [33] Spehner D 2014 Journal of Mathematical Physics 55 075211
- [34] Peres A 1993 Quantum Theory: Concepts and Methods (Dordrecht: Kluwer Academic Publishers)
- [35] Streltsov A, Kampermann H and Bruß D 2011 Physical Review Letters 106 160401
- [36] Piani M, Gharibian S, Adesso G, Calsamiglia J, Horodecki P and Winter A 2011 Physical Review Letters 106 220403
- [37] Zurek W H 2003 Reviews of Modern Physics 75 715-775
- [38] Adesso G, DAmbrosio V, Nagali E, Piani M and Sciarrino F 2014 Physical Review Letters 112 140501
- [39] Gharibian S 2012 Physical Review A 86 042106
- [40] Giampaolo S, Streltsov A, Roga W, Bruß D and Illuminati F 2013 Physical Review A 87 012313
- [41] Roga W, Giampaolo S and Illuminati F 2014 Journal of Physics A: Mathematical and Theoretical 47 365301
- [42] Åberg J 2006 arXiv preprint quant-ph/0612146
- [43] Baumgratz T, Cramer M and Plenio M B 2014 Physical Review Letters 113 140401
- [44] Winter A and Yang D 2016 Physical Review Letters 116 120404
- [45] Napoli C, Bromley T R, Cianciaruso M, Piani M, Johnston N and Adesso G 2016 Physical Review Letters 116 150502
- [46] Chitambar E, Streltsov A, Rana S, Bera M N, Adesso G and Lewenstein M 2016 Physical Review Letters 116 070402
- [47] Ma J, Yadin B, Girolami D, Vedral V and Gu M 2016 Physical Review Letters 116 160407
- [48] Bromley T R, Cianciaruso M and Adesso G 2015 Physical Review Letters 114 210401
- [49] Streltsov A, Singh U, Dhar H S, Bera M N and Adesso G 2015 Physical Review Letters 115 020403
- [50] Seshadreesan K P and Wilde M M 2015 Physical Review A 92 042321
- [51] Horodecki M, Shor P W and Ruskai M B 2003 Reviews in Mathematical Physics 15 629-641
- [52] Fukuda M, Nechita I and Wolf M M 2015 Information Theory, IEEE Transactions on 61 1851–1859
- [53] Brodutch A and Modi K 2012 Quantum Information & Computation 12 721–742
- [54] Girolami D, Tufarelli T and Adesso G 2013 Physical Review Letters 110 240402
- [55] Cianciaruso M, Bromley T R, Roga W, Lo Franco R and Adesso G 2015 Scientific Reports 5 10177
- [56] Roga W, Spehner D and Illuminati F 2016 Journal of Physics A: Mathematical and Theoretical 49 235301
- [57] Girolami D, Souza A M, Giovannetti V, Tufarelli T, Filgueiras J G, Sarthour R S, Soares-Pinto D O, Oliveira I S and Adesso G 2014 *Physical Review Letters* 112 210401
- [58] Farace A, De Pasquale A, Rigovacca L and Giovannetti V 2014 New Journal of Physics 16 073010
- [59] Ciccarello F, Tufarelli T and Giovannetti V 2014 New Journal of Physics 16 013038
- [60] Coecke B, Fritz T and Spekkens R W 2016 Information and Computation 250 59 86
- [61] Brandão F G S L, Horodecki M, Ng N, Oppenheim J and Wehner S 2015 Proceedings of the National Academy of Sciences 112 3275–3279
- [62] Vidal G 2000 Journal of Modern Optics 47 355-376
- [63] Hu X, Fan H, Zhou D and Liu W M 2012 Physical Review A 85 032102
- [64] Ciccarello F and Giovannetti V 2012 Physical Review A 85 010102
- [65] Streltsov A, Kampermann H and Bruß D 2011 Physical Review Letters 107 170502
- [66] Guo Y and Hou J 2013 Journal of Physics A: Mathematical and Theoretical 46 155301
- [67] Bromley T R, Silva I A, Oncebay-Segura C O, Soares-Pinto D O, deAzevedo E R, Tufarelli T and Adesso G 2016 arXiv preprint quant-ph:1610.07504
- [68] Brandão F G S L and Plenio M B 2008 Nature Physics 4 873
- [69] Marvian I and Spekkens R W 2013 New Journal of Physics 15 033001
- [70] Marvian I, Spekkens R W and Zanardi P 2016 Physical Review A 93 052331
- [71] Yadin B, Ma J, Girolami D, Gu M and Vedral V 2015 arXiv preprint arXiv:1512.02085
- [72] Chitambar E and Gour G 2016 Physical Review Letters 117 030401
- [73] Marvian I and Spekkens R W 2016 arXiv preprint arXiv:1602.08049
- [74] Streltsov A, Adesso G, Piani M and Bruß D 2012 Physical Review Letters 109 050503
- [75] Plenio M B 2005 Physical Review Letters 95 090503
- [76] Adesso G, Ragy S and Lee A R 2014 Open Systems and Information Dynamics 21 1440001
- [77] Bengtsson I and Zyczkowski K 2007 Geometry of quantum states: an introduction to quantum entanglement (Cambridge: Cambridge University Press)
- [78] Paula F, de Oliveira T R and Sarandy M 2013 Physical Review A 87 064101
- [79] Luo S and Zhang Q 2004 Physical Review A 69 032106
- [80] Nakano T, Piani M and Adesso G 2013 Physical Review A 88 012117
- [81] Monras A, Adesso G, Giampaolo S, Gualdi G, Davies G and Illuminati F 2011 Physical Review A 84 012301
- [82] Girolami D 2014 Physical Review Letters 113 170401
- [83] Marvian I and Spekkens R W 2014 Nature Communications 5 3821
- [84] Streltsov A, Adesso G and Plenio M B 2016 arXiv preprint arXiv:1609.02439
- [85] Xi Z, Li Y and Fan H 2015 Scientific Reports 5 10922

- [86] Killoran N, Steinhoff F E S and Plenio M B 2016 Physical Review Letters 116 080402
- [87] Horodecki M, Horodecki P, Horodecki R, Oppenheim J, Sen A, Sen U, Synak-Radtke B et al. 2005 Physical Review A 71 062307
- [88] Groisman B, Kenigsberg D and Mor T 2007 arXiv preprint quant-ph/0703103
- [89] Vedral V 2002 Reviews of Modern Physics 74 197
- [90] Lindblad G 1975 Communications in Mathematical Physics 40 147-151
- [91] Uhlmann A 1977 Communications in Mathematical Physics 54 21–32
- [92] Vedral V and Plenio M B 1998 Physical Review A 57 1619
- [93] Modi K, Paterek T, Son W, Vedral V and Williamson M 2010 Physical Review Letters 104 080501
- [94] Zurek W H 2003 Physical Review A 67 012320
- [95] Yao Y, Xiao X, Ge L and Sun C P 2015 Physical Review A 92 022112
- [96] Tufarelli T, MacLean T, Girolami D, Vasile R and Adesso G 2013 Journal of Physics A: Mathematical and Theoretical 46 275308
- [97] Luo S 2008 Physical Review A 77 042303
- [98] Chitambar E 2012 Physical Review A 86 032110
- [99] Luo S and Fu S 2010 Physical Review A 82 034302
- [100] Luo S and Fu S 2012 Theoretical and Mathematical Physics 171 870-878
- [101] Vinjanampathy S and Rau A 2012 Journal of Physics A: Mathematical and Theoretical 45 095303
- [102] Tufarelli T, Girolami D, Vasile R, Bose S and Adesso G 2012 Physical Review A 86 052326

[103] Dakić B, Lipp Y O, Ma X, Ringbauer M, Kropatschek S, Barz S, Paterek T, Vedral V, Zeilinger A, Brukner Č et al. 2012 Nature Physics 8 666–670

- [104] Silva I A, Girolami D, Auccaise R, Sarthour R S, Oliveira I S, Bonagamba T J, deAzevedo E R, Soares-Pinto D O and Adesso G 2013 *Physical Review Letters* 110 140501
- [105] Passante G, Moussa O and Laflamme R 2012 Physical Review A 85 032325
- [106] Jin J s, Zhang F y, Yu C s and Song H s 2012 Journal of Physics A: Mathematical and Theoretical 45 115308
- [107] Girolami D and Adesso G 2012 Physical Review Letters 108 150403
- [108] Hassan A S M, Lari B and Joag P S 2012 Physical Review A 85 024302
- [109] Rana S and Parashar P 2012 Physical Review A 85 024102
- [110] Feng-Jian J, Hai-Jiang L, Xin-Hu Y and Ming-Jun S 2013 Chinese Physics B 22 040303
- [111] Piani M 2012 Physical Review A 86 034101
- [112] Hu X, Fan H, Zhou D and Liu W M 2013 Physical Review A 87 032340
- [113] Chang L and Luo S 2013 *Physical Review A* 87 062303
- [114] Rungta P and Caves C M 2003 Physical Review A 67 012307
- [115] Życzkowski K, Horodecki P, Sanpera A and Lewenstein M 1998 Physical Review A 58 883-892
- [116] Vidal G and Werner R F 2002 Physical Review A 65 032314
- [117] Girolami D and Adesso G 2011 Physical Review A 84 052110
- [118] Rana S and Parashar P 2012 *Physical Review A* **86** 030302
- [119] Adesso G and Girolami D 2011 International Journal of Quantum Information 9 1773
- [120] Adesso G and Datta A 2010 Physical Review Letters 105 030501
- [121] Mišta L, McNulty D and Adesso G 2014 Physical Review A 90 022328
- [122] Debarba T, Maciel T O and Vianna R O 2012 Physical Review A 86 024302
- [123] Rana S and Parashar P 2013 *Physical Review A* 87 016301
- [124] Aaronson B, Lo Franco R, Compagno G and Adesso G 2013 New Journal of Physics 15 093022
- [125] Helstrom C W 1976 Quantum detection and estimation theory (New York: Academic press)
- [126] Paula F M, Silva I A, Montealegre J D, Souza A M, deAzevedo E R, Sarthour R S, Saguia A, Oliveira I S,
- Soares-Pinto D O, Adesso G and Sarandy M S 2013 Physical Review Letters 111 250401
- [127] Spehner D and Orszag M 2013 New Journal of Physics 15 103001
- [128] Spehner D and Orszag M 2013 Journal of Physics A: Mathematical and Theoretical 47 035302
- [129] Bromley T R, Cianciaruso M, Lo Franco R and Adesso G 2014 Journal of Physics A: Mathematical and Theoretical 47 405302
- [130] Wei T C and Goldbart P M 2003 Physical Review A 68 042307
- [131] Marian P and Marian T A 2015 Journal of Physics A: Mathematical and Theoretical 48 115301
- [132] Audenaert K M, Calsamiglia J, Munoz-Tapia R, Bagan E, Masanes L, Acín A and Verstraete F 2007 Physical Review Letters 98 160501
- [133] Fuchs C A and Van De Graaf J 1999 Information Theory, IEEE Transactions on 45 1216–1227
- [134] Morozova E A and Chentsov N N 1991 Journal of Soviet Mathematics 56 2648–2669
- [135] Petz D 1996 Linear algebra and its applications 244 81–96
- [136] Wilde M M, Winter A and Yang D 2014 Communications in Mathematical Physics 331 593-622
- [137] Müller-Lennert M, Dupuis F, Szehr O, Fehr S and Tomamichel M 2013 Journal of Mathematical Physics 54 122203
- [138] Frank R L and Lieb E H 2013 Journal of Mathematical Physics 54 122201

- [139] Rajagopal A, Nayak A S, Devi A U et al. 2014 Physical Review A 89 012331
- [140] Luo S 2008 Physical Review A 77 022301
- [141] Piani M, Narasimhachar V and Calsamiglia J 2014 New Journal of Physics 16 113001
- [142] Lang M D, Caves C M and Shaji A 2011 International Journal of Quantum Information 9 1553-1586
- [143] Wu S, Poulsen U V, Mølmer K et al. 2009 Physical Review A 80 032319
- [144] Shannon C E 1948 The Bell System Techncal Journal 27 379-423
- [145] Cover T M and Thomas J A 2012 Elements of information theory (New York: John Wiley & Sons)
- [146] Girolami D, Paternostro M and Adesso G 2011 Journal of Physics A: Mathematical and Theoretical 44 352002
- [147] Terhal B M, Horodecki M, Leung D W and DiVincenzo D P 2002 Journal of Mathematical Physics 43 4286– 4298
- [148] DiVincenzo D P, Horodecki M, Leung D W, Smolin J A and Terhal B M 2004 Physical Review Letters 92 067902
- [149] Hall M J, Andersson E and Brougham T 2006 Physical Review A 74 062308
- [150] Wehrl A 1978 Reviews of Modern Physics 50 221
- [151] Datta A 2008 Studies on the role of entanglement in mixed-state quantum computation Ph.D. thesis University of New Mexico
- [152] Huang Y 2014 New Journal of Physics 16 033027
- [153] Hamieh S, Kobes R and Zaraket H 2004 Physical Review A 70 052325
- [154] D'Ariano G M, Presti P L and Perinotti P 2005 Journal of Physics A: Mathematical and General 38 5979
- [155] Galve F, Giorgi G L and Zambrini R 2011 EPL (Europhysics Letters) 96 40005
- [156] Chen Q, Zhang C, Yu S, Yi X and Oh C 2011 Physical Review A 84 042313
- [157] Shi M, Sun C, Jiang F, Yan X and Du J 2012 Physical Review A 85 064104
- [158] Shi M, Yang W, Jiang F and Du J 2011 Journal of Physics A: Mathematical and Theoretical 44 415304
- [159] Ali M, Rau A and Alber G 2010 Physical Review A 81 042105
- [160] Lu X M, Ma J, Xi Z and Wang X 2011 Physical Review A 83 012327
- [161] Shi M, Jiang F, Sun C and Du J 2011 New Journal of Physics 13 073016
- [162] Sabapathy K K and Simon R 2013 arXiv preprint arXiv:1311.0210
- [163] Huang Y 2013 Physical Review A 88 014302
- [164] Fanchini F, Werlang T, Brasil C, Arruda L and Caldeira A 2010 Physical Review A 81 052107
- [165] Li B, Wang Z X and Fei S M 2011 Physical Review A 83 022321
- [166] Yurischev M 2015 Quantum Information Processing 14 3399-3421
- [167] Jing N and Yu B 2016 Journal of Physics A: Mathematical and Theoretical 49 385302
- [168] Rossignoli R, Matera J M and Canosa N 2012 Physical Review A 86 022104
- [169] Ciliberti L, Rossignoli R and Canosa N 2010 Physical Review A 82 042316
- [170] Ciliberti L, Canosa N and Rossignoli R 2013 Physical Review A 88 012119
- [171] Girolami D and Adesso G 2011 Physical Review A 83 052108
- [172] Wu X and Zhou T 2015 *Quantum Information Processing* 14 1959–1971
- [173] Giorda P and Paris M G A 2010 Physical Review Letters 105 020503
- [174] Pirandola S, Spedalieri G, Braunstein S L, Cerf N J and Lloyd S 2014 Physical Review Letters 113 140405
- [175] Mišta L, Tatham R, Girolami D, Korolkova N and Adesso G 2011 Physical Review A 83 042325
- [176] Koashi M and Winter A 2004 *Physical Review A* 69 022309
- [177] Fanchini F F, Cornelio M F, de Oliveira M C and Caldeira A O 2011 Physical Review A 84 012313
- [178] Devetak I 2005 Physical Review A 71 062303
- [179] Devetak I and Winter A 2004 Information Theory, IEEE Transactions on 50 3183–3196
- [180] Rossignoli R, Canosa N and Ciliberti L 2010 Physical Review A 82 052342
- [181] Rossignoli R, Canosa N and Ciliberti L 2011 Physical Review A 84 052329
- [182] Bosyk G, Bellomo G, Zozor S, Portesi M and Lamberti P 2016 Physica A: Statistical Mechanics and its Applications 462 930–939 ISSN 0378-4371
- [183] Bellomo B, Lo Franco R and Compagno G 2012 Physical Review A 86 012312
- [184] Canosa N, Ciliberti L and Rossignoli R 2015 Entropy 17 1634–1659
- [185] Misra A, Biswas A, Pati A K, Sen A, Sen U et al. 2015 Physical Review E 91 052125
- [186] Berta M, Seshadreesan K P and Wilde M M 2015 Journal of Mathematical Physics 56 022205
- [187] Chi D P, San Kim J and Lee K 2013 Physical Review A 87 062339
- [188] Jurkowski J 2013 International Journal of Quantum Information 11 1350013
- [189] Majtey A, Plastino A and Plastino A 2012 Physica A: Statistical Mechanics and its Applications 391 2491– 2499
- [190] Adesso G, Girolami D and Serafini A 2012 Physical Review Letters 109 190502
- [191] Costa A and Angelo R 2013 Physical Review A 87 032109
- [192] Hiroshima T and Hayashi M 2004 Physical Review A 70 030302
- [193] Orieux A, Ciampini M A, Mataloni P, Bruß D, Rossi M and Macchiavello C 2015 Physical Review Letters 115 160503

- [194] Mazzola L and Paternostro M 2011 Scientific Reports 1 199
- [195] Farace A, Ciccarello F, Fazio R and Giovannetti V 2014 Physical Review A 89 022335
- [196] Giampaolo S M and Illuminati F 2007 Physical Review A 76 042301
- [197] Roga W, Buono D and Illuminati F 2015 New Journal of Physics 17 013031
- [198] Uhlmann A 1976 Reports on Mathematical Physics 9 273–279
- [199] Rigovacca L, Farace A, De Pasquale A and Giovannetti V 2015 Physical Review A 92 042331
- [200] Pirandola S and Lloyd S 2008 Physical Review A 78 012331
- [201] Luo S 2003 Physical Review Letters 91 180403
- [202] Adesso G 2014 Physical Review A 90 022321
- [203] Frérot I and Roscilde T 2016 Physical Review B 94 075121
- [204] Rastegin A E 2016 Physical Review A 93 032136
- [205] Wilde M M 2013 Quantum Information Theory (Cambridge: Cambridge University Press)
- [206] Fawzi O and Renner R 2015 Communications in Mathematical Physics 340 575-611
- [207] Piani M 2016 Physical Review Letters 117(8) 080401
- [208] Seshadreesan K P, Berta M and Wilde M M 2015 Journal of Physics A: Mathematical and Theoretical 48 395303
- [209] Christandl M 2006 The Structure of Bipartite Quantum States Insights from Group Theory and Cryptography Ph.D. thesis University of Cambridge
- [210] Kimble H J 2008 Nature **453** 1023–1030
- [211] Wootters W K and Zurek W H 1982 Nature 299 802-803
- [212] Dieks D 1982 Physics Letters A 92 271-272
- [213] Barnum H, Caves C M, Fuchs C A, Jozsa R and Schumacher B 1996 Physical Review Letters 76 2818
- [214] Luo S 2010 Letters in Mathematical Physics 92 143-153
- [215] Chatterjee S, Sazim S and Chakrabarty I 2016 Physical Review A 93 042309
- [216] Brandão F G S L, Piani M and Horodecki P 2015 Nature Communications 6 7908
- [217] Streltsov A and Zurek W H 2013 Physical Review Letters 111 040401
- [218] Zurek W H 2009 Nature Physics 5 181–188
- [219] Hudson R 2004 International Journal of Theoretical Physics 43 1841-1847
- [220] Gharibian S, Piani M, Adesso G, Calsamiglia J and Horodecki P 2011 International Journal of Quantum Information 9 1701–1713
- [221] Cubitt T S, Verstraete F, Dür W and Cirac J I 2003 Physical Review Letters 91 037902
- [222] Chuan T, Maillard J, Modi K, Paterek T, Paternostro M and Piani M 2012 Physical Review Letters 109 070501
- [223] Streltsov A, Kampermann H and Bruß D 2012 Physical Review Letters 108 250501
- [224] Streltsov A, Kampermann H and Bruß D 2014 Physical Review A 90 032323
- [225] Kay A 2012 Physical Review Letters 109 080503
- [226] Fedrizzi A, Zuppardo M, Gillett G, Broome M, Almeida M, Paternostro M, White A and Paterek T 2013 Physical Review Letters 111 230504
- [227] Peuntinger C, Chille V, Mišta Jr L, Korolkova N, Förtsch M, Korger J, Marquardt C and Leuchs G 2013 Physical Review Letters 111 230506
- [228] Vollmer C E, Schulze D, Eberle T, Händchen V, Fiurášek J and Schnabel R 2013 Physical Review Letters 111 230505
- [229] Slepian D and Wolf J K 1973 Information theory, IEEE Transactions on 19 471-480
- [230] Bennett C H, Brassard G, Crépeau C, Jozsa R, Peres A and Wootters W K 1993 Physical Review Letters 70 1895–1899
- [231] Horodecki M, Oppenheim J and Winter A 2005 Nature 436 673-676
- [232] Horodecki M, Oppenheim J and Winter A 2007 Communications in Mathematical Physics 269 107-136
- [233] Cavalcanti D, Aolita L, Boixo S, Modi K, Piani M and Winter A 2011 Physical Review A 83 032324
- [234] Madhok V and Datta A 2011 Physical Review A 83 032323
- [235] Madhok V and Datta A 2011 Quantum discord in quantum information theory-from strong subadditivity to the mother protocol *Theory of Quantum Computation, Communication, and Cryptography* (Springer) pp 188–197
- [236] Madhok V and Datta A 2013 International Journal of Modern Physics B 27 1345041
- [237] Abeyesinghe A, Devetak I, Hayden P and Winter A 2009 Proceedings of the Royal Society of London A: Mathematical, Physical and Engineering Sciences 465 2537–2563
- [238] Streltsov A, Lee S and Adesso G 2015 Physical Review Letters 115 030505
- [239] Devetak I and Yard J 2008 Physical Review Letters 100 230501
- [240] Wilde M M 2015 Proceedings of the Royal Society of London A: Mathematical, Physical and Engineering Sciences 471 20140941
- [241] Berta M, Brandao F G, Majenz C and Wilde M M 2016 arXiv preprint arXiv:1609.06994
- [242] Pirandola S, Eisert J, Weedbrook C, Furusawa A and Braunstein S L 2015 Nature Photonics 9 641-652
- [243] Bennett C H, DiVincenzo D P, Shor P W, Smolin J A, Terhal B M and Wootters W K 2001 Physical Review

Letters 87 077902

- [244] Bennett C H, Hayden P, Leung D W, Shor P W and Winter A 2005 Information Theory, IEEE Transactions on **51** 56–74
- [245] Lo H K 2000 Physical Review A 62 012313
- [246] Pati A K 2000 Physical Review A 63 014302
- [247] Horodecki P, Tuziemski J, Mazurek P and Horodecki R 2014 Physical Review Letters 112 140507
- [248] Giorgi G L 2013 Physical Review A 88 022315
- [249] Chaves R and de Melo F 2011 Physical Review A 84 022324
- [250] Gisin N, Ribordy G, Tittel W and Zbinden H 2002 Reviews of Modern Physics 74 145
- [251] Pirandola S 2014 Scientific Reports 4 6956
- [252] Pirandola S, Laurenza R, Ottaviani C and Banchi L 2015 arXiv preprint arXiv:1510.08863
- [253] Pati A K, Wilde M M, Devi A U, Rajagopal A et al. 2012 Physical Review A 86 042105
- [254] Weedbrook C, Pirandola S, García-Patrón R, Cerf N J, Ralph T C, Shapiro J H and Lloyd S 2012 Reviews of Modern Physics 84 621
- [255] Durt T, Englert B G, Bengtsson I and Życzkowski K 2010 International journal of quantum information 8 535–640
- [256] Datta A and Gharibian S 2009 Physical Review A 79 042325
- [257] Boixo S, Aolita L, Cavalcanti D, Modi K and Winter A 2011 International Journal of Quantum Information 9 1643–1651
- [258] Giovannetti V, Lloyd S and Maccone L 2011 Nature Photonics 5 222-229
- [259] Pezzè L and Smerzi A 2014 Arxiv preprint arXiv:1411.5164 Published in "Atom Interferometry", Proceedings of the International School of Physics "Enrico Fermi", Course 188, Varenna. Edited by Tino G M and Kasevich M A (IOS Press, Amsterdam, 2014). Page 691
- [260] Tóth G and Apellaniz I 2014 Journal of Physics A: Mathematical and Theoretical 47 424006 ISSN 1751-8121
- [261] Pezzè L, Smerzi A, Oberthaler M K, Schmied R and Treutlein P 2016 arXiv preprint arXiv:1609.01609
- [262] Giovannetti V, Lloyd S and Maccone L 2004 Science 306 1330–1336
- [263] Giovannetti V, Lloyd S and Maccone L 2006 Physical Review Letters 96 010401
- [264] Braunstein S L and Caves C M 1994 Physical Review Letters 72 3439
- [265] Farace A, De Pasquale A, Adesso G and Giovannetti V 2016 New Journal of Physics 18 013049
- [266] Modi K, Cable H, Williamson M and Vedral V 2011 *Physical Review X* 1 021022
- [267] Bae J and Kwek L C 2015 Journal of Physics A: Mathematical and Theoretical 48 083001
- [268] Lloyd S 2008 Science 321 1463-1465
- [269] Tan S H, Erkmen B I, Giovannetti V, Guha S, Lloyd S, Maccone L, Pirandola S and Shapiro J H 2008 Physical Review Letters 101 253601
- [270] Barzanjeh S, Guha S, Weedbrook C, Vitali D, Shapiro J H and Pirandola S 2015 Physical Review Letters 114 080503
- [271] Weedbrook C, Pirandola S, Thompson J, Vedral V and Gu M 2016 New Journal of Physics 18 043027
- [272] Zhang Z, Tengner M, Zhong T, Wong F N and Shapiro J H 2013 Physical Review Letters 111 010501
- [273] Gu M, Chrzanowski H M, Assad S M, Symul T, Modi K, Ralph T C, Vedral V and Lam P K 2012 Nature Physics 8 671–675
- [274] Pirandola S 2011 Physical Review Letters 106 090504
- [275] Pirandola S, Lupo C, Giovannetti V, Mancini S and Braunstein S L 2011 New Journal of Physics 13 113012
- [276] Ragy S and Adesso G 2012 Scientific Reports 2 651
- [277] Ragy S, Berchera I R, Degiovanni I P, Olivares S, Paris M G, Adesso G and Genovese M 2014 JOSA B 31 2045–2050
- [278] Goold J, Huber M, Riera A, del Rio L and Skrzypczyk P 2016 Journal of Physics A: Mathematical and Theoretical 49 143001
- [279] Millen J and Xuereb A 2016 New Journal of Physics 18 011002
- [280] Oppenheim J, Horodecki M, Horodecki P and Horodecki R 2002 Physical Review Letters 89 180402
- [281] Hovhannisyan K V, Perarnau-Llobet M, Huber M and Acín A 2013 Physical Review Letters 111 240401
- [282] Leggio B, Bellomo B and Antezza M 2015 Physical Review A 91 012117
- [283] Park J J, Kim K H, Sagawa T and Kim S W 2013 Physical Review Letters 111 230402
- [284] Dillenschneider R and Lutz E 2009 EPL (Europhysics Letters) 88 50003
- [285] Correa L A, Palao J P, Adesso G and Alonso D 2013 Physical Review E 87 042131
- [286] Liuzzo-Scorpo P, Correa L A, Schmidt R and Adesso G 2016 Entropy 18 48 ISSN 1099-4300
- [287] Grimsmo A L 2013 Physical Review A 87 060302
- [288] Lloyd S, Chiloyan V, Hu Y, Huberman S, Liu Z W and Chen G 2015 arXiv preprint arXiv:1510.05035
- [289] Sarandy M S, De Oliveira T R and Amico L 2013 International Journal of Modern Physics B 27 1345030
- [290] Werlang T, Ribeiro G and Rigolin G 2013 International Journal of Modern Physics B 27 1345032
- [291] Amico L, Fazio R, Osterloh A and Vedral V 2008 Reviews of Modern Physics 80 517-576
- [292] Dillenschneider R 2008 Physical Review B 78 224413

- [293] Sarandy M 2009 Physical Review A 80 022108
- [294] Werlang T and Rigolin G 2010 Physical Review A 81 044101
- [295] Werlang T, Trippe C, Ribeiro G and Rigolin G 2010 Physical Review Letters 105 095702
- [296] Rong X, Wang Z, Jin F, Geng J, Feng P, Xu N, Wang Y, Ju C, Shi M and Du J 2012 Physical Review B 86 104425
- [297] Malpetti D and Roscilde T 2016 Physical Review Letters 117(13) 130401
- [298] Maziero J, Guzman H, Céleri L, Sarandy M and Serra R 2010 Physical Review A 82 012106
- [299] Maziero J, Céleri L, Serra R and Sarandy M 2012 Physics Letters A 376 1540-1544
- [300] Cakmak B, Karpat G and Gedik Z 2012 Physics Letters A 376 2982–2988
- [301] Campbell S, Richens J, Lo Gullo N and Busch T 2013 Physical Review A 88 062305
- [302] Huang Y 2014 Physical Review B 89 054410
- [303] Karpat G, Cakmak B and Fanchini F F 2014 Physical Review B 90 104431
- [304] Tomasello B, Rossini D, Hamma A and Amico L 2011 EPL (Europhysics Letters) 96 27002
- [305] Tomasello B, Rossini D, Hamma A and Amico L 2012 International Journal of Modern Physics B 26 1243002
- [306] Liu B Q, Shao B, Li J G, Zou J and Wu L A 2011 Physical Review A 83 052112
- [307] Wang C, Zhang Y Y and Chen Q H 2012 Physical Review A 85 052112
- [308] Chen Y X and Li S W 2010 Physical Review A 81 032120
- [309] Chen Y X, Li S W and Yin Z 2010 Physical Review A 82 052320
- [310] Allegra M, Giorda P and Montorsi A 2011 Physical Review B 84 245133
- [311] Malvezzi A L, Karpat G, Çakmak B, Fanchini F F, Debarba T and Vianna R O 2016 Physical Review B 93 184428
- [312] Coulamy I, Warnes J, Sarandy M and Saguia A 2016 Physics Letters A 380 1724 1728 ISSN 0375-9601
- [313] Rulli C C and Sarandy M S 2011 Physical Review A 84 042109
- [314] Campbell S, Mazzola L, Chiara G D, Apollaro T J G, Plastina F, Busch T and Paternostro M 2013 New Journal of Physics 15 043033
- [315] Power M J M, Campbell S, Moreno-Cardoner M and De Chiara G 2015 Physical Review B 91 214411
- [316] Wilde M M 2015 Proceedings of the Royal Society of London A: Mathematical, Physical and Engineering Sciences 471 20150338
- [317] Chakrabarty I, Agrawal P and Pati A K 2011 The European Physical Journal D 65 605-612 ISSN 1434-6079
- [318] Zhou D L 2008 Physical Review Letters **101** 180505
- [319] Giorgi G L, Bellomo B, Galve F and Zambrini R 2011 Physical Review Letters 107 190501
- [320] Coffman V, Kundu J and Wootters W K 2000 Physical Review A 61 052306
- [321] Lancien C, Di Martino S, Huber M, Piani M, Adesso G and Winter A 2016 Physical Review Letters 117 060501
- [322] Dhar H S, Pal A K, Rakshit D, Sen De A and Sen U 2016 arXiv preprint arXiv:1610.01069
- [323] Jevtic S, Pusey M, Jennings D and Rudolph T 2014 *Physical Review Letters* **113** 020402
- [324] Maziero J, Céleri L C, Serra R M and Vedral V 2009 Physical Review A 80 044102
- [325] Werlang T, Souza S, Fanchini F F and Villas Boas C J 2009 Physical Review A 80 024103
- [326] Mazzola L, Piilo J and Maniscalco S 2010 Physical Review Letters 104 200401
- [327] Merali Z 2011 Nature 474 24
- [328] Datta A, Shaji A and Caves C M 2008 Physical Review Letters 100 050502
- [329] Knill E and Laflamme R 1998 Physical Review Letters 81 5672
- [330] Lanyon B, Barbieri M, Almeida M and White A 2008 Physical Review Letters 101 200501
- [331] Passante G, Moussa O, Trottier D and Laflamme R 2011 Physical Review A 84 044302
- [332] Cui J and Fan H 2010 Journal of Physics A: Mathematical and Theoretical 43 045305
- [333] Vedral V 2010 Foundations of Physics 40 1141–1154
- [334] Eastin B 2010 arXiv preprint arXiv:1006.4402
- [335] Cable H and Browne D E 2015 New Journal of Physics 17 113049